UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

M			l۶	O	n	^)	
VI	я	r	ĸ		m	e	۱

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended July 3, 2004

OR

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from

to

Commission file number: 0-21116

USANA HEALTH SCIENCES, INC.

(Exact name of registrant as specified in its charter)

Utah

(State or other jurisdiction of incorporation or organization)

87-0500306 (I.R.S. Employer

(I.R.S. Employer Identification No.)

3838 West Parkway Blvd., Salt Lake City, Utah 84120

(Address of principal executive offices, Zip Code)

(801) 954-7100

(Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes ☑ No □

Indicate by check mark whether the registrant is an accelerated filer (as defined in Rule 12b-2 of the Exchange Act). Yes 🗵 No 🗖

The number of shares outstanding of the registrant's common stock as of July 30, 2004 was 19,009,774.

On October 14, 2003, the registrant declared a two-for-one stock split of its common stock that was distributed in the form of a stock dividend on October 30, 2003 to shareholders of record as of October 24, 2003. Outstanding common stock data in this report for periods prior to October 30, 2003 have been adjusted to reflect the stock split.

USANA HEALTH SCIENCES, INC.

FORM 10-Q

For the Quarterly Period Ended July 3, 2004

INDEX

PART I. FINANCIAL INFORMATION

Consolidated Statements of Earnings – Quarter Ended Consolidated Statements of Earnings – Six Months Ended

Consolidated Statement of Stockholders' Equity

Consolidated Statements of Cash Flows

Notes to Consolidated Financial Statements

Management's Discussion and Analysis of Financial Condition and Results of Operations Item 2

Item 3 Quantitative and Qualitative Disclosures About Market Risk

Item 4 **Controls and Procedures**

PART II. OTHER INFORMATION

Changes in Securities and Use of Proceeds. Item 2

Item 4 Submission of Matters to a Vote of Security Holders.

Item 6 Exhibits and Reports on Form 8-K.

Signatures

2

PART I. FINANCIAL INFORMATION

Item 1. Financial Statements

USANA HEALTH SCIENCES, INC. AND SUBSIDIARIES

CONSOLIDATED BALANCE SHEETS

(in thousands, except per share data)

	January 3, 2004			July 3, 2004
			(U	naudited)
ASSETS				
Current assets	Ф	10.065	Ф	12.700
Cash and cash equivalents	\$	18,965 14,069	\$	13,798 13,574
Inventories, net Prepaid expenses and other current assets		3,294		3,658
Deferred income taxes		1,921		1,669
Deterred income taxes	_	1,921		1,009
Total current assets		38,249		32,699
Property and equipment, net		20,195		24,062
Goodwill		4,267		5,690
Other assets		2,416		2,498
	\$	65,127	\$	64,949
	Ψ	03,127	Ψ	04,242
LIABILITIES AND STOCKHOLDERS' EQUITY				
Current liabilities				
Accounts payable	\$	5,215	\$	4,103
Other current liabilities		14,704		15,197
Total current liabilities		19,919		19,300
Other long-term liabilities		837		772
Stockholders' equity				
Common stock, \$0.001 par value; authorized 50,000 shares, issued and outstanding 19,470 as of				
January 3, 2004 and 19,170 as of July 3, 2004		19		19
Additional paid-in capital		14,187		11,547
Retained earnings		28,935		32,169
Accumulated other comprehensive income		1,230		1,142
Total stockholders' equity		44,371		44,877
	\$	65,127	\$	64,949

USANA HEALTH SCIENCES, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF EARNINGS

(in thousands, except per share data)

	 Quarter Ended (unaudited)		
	June 28, 2003		July 3, 2004
Net sales	\$ 47,157	\$	67,246
Cost of sales	10,417		16,195
Gross profit	36,740		51,051
Operating expenses:			
Associate incentives	18,662		25,556
Selling, general and administrative	10,574		13,656
Research and development	 373	_	607
Total operating expenses	29,609	_	39,819
Earnings from operations	7,131		11,232
Other income (expense):			
Interest income	21		36
Interest expense	(1)		
Other, net	(248)	_	(37)
Total other expense	(228)		(1)
Earnings before income taxes	6,903		11,231
Income taxes	2,554		3,818
Net earnings	\$ 4,349	\$	7,413
Earnings per common share			
Basic	\$ 0.23	\$	0.39
Diluted	\$ 0.20	\$	0.36
Weighted average common and dilutive common equivalent shares outstanding			
Basic	19,087		19,199
Diluted	21,450		20,523
The accompanying notes are an integral part of these statements.			
4			

USANA HEALTH SCIENCES, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF EARNINGS

(in thousands, except per share data)

	Six Montl (unau	d
_	June 28, 2003	July 3, 2004
\$	88,021	\$ 129,021
<u> </u>	19,637	31,253

Gross profit	68,384	97,768
Operating expenses:		
Associate incentives	34,759	49,168
Selling, general and administrative	20,146	26,918
Research and development	 707	 1,185
Total operating expenses	 55,612	77,271
Earnings from operations	12,772	20,497
Other income (expense):		
Interest income	41	86
Interest expense	(48)	
Other, net	 (187)	 62
Total other income (expense)	 (194)	148
Earnings before income taxes	12,578	20,645
Income taxes	 4,654	7,019
Net earnings	\$ 7,924	\$ 13,626
Earnings per common share		
Basic	\$ 0.42	\$ 0.71
Diluted	\$ 0.37	\$ 0.66
Weighted average common and dilutive common equivalent shares outstanding		
Basic	18,853	19,288
Diluted	21,247	20,688

The accompanying notes are an integral part of these statements.

5

USANA HEALTH SCIENCES, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENT OF STOCKHOLDERS' EQUITY

Six Months Ended June 28, 2003 and July 3, 2004

(in thousands)

(unaudited)

	Commo	n Sto	ock		Additional Paid-in	Retained	(Accumulated Other Comprehensive	
	Shares		Value	_	Capital	 Earnings		Income (Loss)	Total
For the Six Months Ended June 28, 2003									
Balance at December 28, 2002	18,273	\$	18	\$	3,666	\$ 14,520	\$	(111) \$	18,093
Comprehensive income									
Net earnings	_		_		_	7,924		_	7,924
Foreign currency translation adjustment	_		_		_	_		892	892
Comprehensive income									8,816
Common stock retired	(87)		_		(254)	(894)		_	(1,148)
Common stock issued under stock option plan, including tax benefit of \$3,007	1,038	_	1	_	5,338				5,339
Balance at June 28, 2003	19,224	\$	19	\$	8,750	\$ 21,550	\$	781 \$	31,100
For the Six Months Ended July 3, 2004									

Balance at January 3, 2004	19,470	5 19	\$ 14,187	\$ 28,935	\$ 1,230	\$ 44,371
Comprehensive income						
Net earnings	_	_	_	13,626	_	13,626
Foreign currency translation adjustment	_	_	_	_	(88)	(88)
Comprehensive income						13,538
•						
Common stock retired	(524)	_	(4,505)	(10,392)	_	(14,897)
Common stock issued under stock option						
plan, including tax benefit of \$1,372	224	_	1,865	_	_	1,865
Balance at July 3, 2004	19,170	19	\$ 11,547	\$ 32,169	\$ 1,142	\$ 44,877

The accompanying notes are an integral part of these statements.

6

USANA HEALTH SCIENCES, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF CASH FLOWS

(in thousands)

(unaudited)

		Six Months Ended		
	J	une 28, 2003		July 3, 2004
Increase (decrease) in cash and cash equivalents				
Cash flows from operating activities				
Net earnings	\$	7,924	\$	13,626
Adjustments to reconcile net earnings to net cash provided by operating activities				
Depreciation and amortization		1,839		2,255
Gain on sale of property and equipment		(24)		(2)
Deferred income taxes		(257)		174
Allowance for inventory valuation		1,187		553
Changes in operating assets and liabilities:				
Inventories		(1,097)		(34)
Prepaid expenses and other assets		80		(246)
Accounts payable		1,730		(1,144)
Other current liabilities		5,667		1,688
Total adjustments		9,125		3,244
Net cash provided by operating activities		17,049		16,870
Cash flows from investing activities				
Acquisition, net of cash acquired		_		(2,140)
Purchases of property and equipment		(2,115)		(5,349)
Proceeds from the sale of property and equipment		40		21
Net cash used in investing activities		(2,075)		(7,468)

The accompanying notes are an integral part of these statements.

7

	Six Months	Ended
	June 28, 2003	July 3, 2004
Cash flows from financing activities		
Principal payments of long-term debt	(6,000)	_
Net proceeds from the sale of common stock	2,332	493

Redemption of common stock		(1,148)	(14,897)
Decrease in line of credit		(2,913)	_
Payments on capital lease obligations		(91)	
	·		
Net cash used in financing activities		(7,820)	(14,404)
Effect of exchange rate changes on cash and cash equivalents		124	(165)
Net increase (decrease) in cash and cash equivalents		7,278	(5,167)
Cash and cash equivalents, beginning of period		6,686	18,965
Cash and cash equivalents, end of period	\$	13,964	\$ 13,798
Supplemental disclosures of cash flow information			
Cash paid during the period for:			
Interest	\$	65	\$ —
Income taxes		1,167	5,727

Non-cash activities

In February 2004, the Company acquired FMG Productions, LLC for \$2,060 in cash. In conjunction with the acquisition, liabilities totaling \$80 were assumed for professional fees directly associated with the acquisition.

The accompanying notes are an integral part of these statements.

8

USANA HEALTH SCIENCES, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (in thousands, except per share data) (unaudited)

Basis of Presentation

The unaudited interim consolidated financial information of USANA Health Sciences, Inc. and Subsidiaries (the "Company" or "USANA") has been prepared in accordance with Article 10 of Regulation S-X promulgated by the Securities and Exchange Commission. Certain information and footnote disclosures normally included in financial statements prepared in accordance with accounting principles generally accepted in the United States of America have been condensed or omitted pursuant to such rules and regulations. In the opinion of management, the accompanying interim consolidated financial information contains all adjustments, consisting of normal recurring adjustments, necessary to present fairly the Company's financial position as of July 3, 2004, and results of operations for the quarters and six months ended June 28, 2003 and July 3, 2004. These financial statements should be read in conjunction with the audited consolidated financial statements and notes thereto included in the Company's Annual Report on Form 10-K for the year ended January 3, 2004. The results of operations for the quarter and six months ended July 3, 2004 may not be indicative of the results that may be expected for the fiscal year ending January 1, 2005.

NOTE A – STOCK-BASED COMPENSATION

The Company has applied the disclosure provisions of Statement of Financial Accounting Standards No. 148, "Accounting for Stock-Based Compensation — Transition and Disclosure — An Amendment of FASB Statement No. 123," for the quarters and six months ended June 28, 2003 and July 3, 2004. Issued in December 2002, SFAS No. 148 amends SFAS No. 123, "Accounting for Stock-Based Compensation" to provide alternative methods of transition for a voluntary change to the fair value based method of accounting for stock-based compensation. In addition, this Statement amends the disclosure requirements of SFAS No. 123 to require prominent disclosures in both annual and interim financial statements about the method of accounting for stock-based employee compensation and the effect of the method used on reported results. As permitted by SFAS No. 148, the Company continues to account for stock options under APB Opinion No. 25, under which no compensation has been recognized. The following table illustrates the effects on net earnings and earnings per share as if the Company had applied the fair value recognition provisions of SFAS No. 123, as amended by SFAS No. 148, to stock-based compensation:

		Quarter Ended				Six Mont	ded	
		ne 28, 2003		July 3, 2004	J	une 28, 2003		July 3, 2004
Net earnings	As reported	\$ 4,349	\$	7,413	\$	7,924	\$	13,626
	Pro forma	\$ 4,229	\$	6,973	\$	7,696	\$	12,962
Earnings per share - basic	As reported	\$ 0.23	\$	0.39	\$	0.42	\$	0.71

	Pro forma	\$	0.22	\$ 0.36	\$ 0.41	\$ 0.67
Earnings per share - diluted	As reported	\$	0.20	\$ 0.36	\$ 0.37	\$ 0.66
	Pro forma	\$	0.20	\$ 0.34	\$ 0.36	\$ 0.63
		9				

Weighted average assumptions used to determine the Black-Scholes fair value for options granted during the periods indicated:

	Quarter	Quarter Ended		Six Months E	nded
	June 28, 2003	July 3, 2004	J	une 28, 2003	July 3, 2004
Expected volatility	*	*		77%	76%
Risk free interest rate	*	*		3.86%	4.02%
Expected life	*	*	10 yrs.		10 yrs.
Expected dividend yield	*	*	0%		0%
Weighted average fair value of options granted**	*	*	\$	7.90 \$	29.72

^{*} No grants were issued during the quarters ended June 28, 2003 and July 3, 2004.

Option pricing models require the input of highly subjective assumptions including the expected stock price volatility. Additionally, the Company's employee stock options have characteristics significantly different from those of traded options, including long vesting schedules and changes in the subjective input assumptions that can materially affect the fair value estimate. Management believes the best assumptions available were used to value the options and that the resulting option values were reasonable as of the dates the options were granted.

NOTE B – ACQUISITIONS

In February 2004, the Company completed the acquisition of the net assets of FMG Productions, LLC (FMG), a Utah limited liability company that produces video and audio promotional and training materials for large companies and sales organizations, including the Company. The aggregate investment was \$2,140 including \$2,060 in cash and \$80 for professional fees directly associated with the acquisition. The purchase was made through a newly formed wholly owned subsidiary of the Company, which will operate the business formerly conducted by FMG. The former employees of FMG, including its founders and primary creative directors, will continue to operate the business now owned by USANA. The Company expects to realize future benefits from this acquisition primarily through the motivation and training of its independent Associates.

The acquisition was accounted for in accordance with SFAS No. 141, "Business Combinations" and, as such, the results of operations for FMG have been included in the consolidated financial statements since the effective date of the acquisition. The assets acquired and liabilities assumed were recorded at estimated fair values as of the date of the acquisition as determined by the Company's management and are summarized on the following page:

10

	At February 1, 2004
Assets acquired	
Accounts receivable	\$ 133
Property and equipment	790
Goodwill	1,423
Total assets acquired	2,346
Liabilities assumed	
Accounts payable	23
Deferred revenue	94
Other liabilities	89
Total liabilities assumed	206
Net assets acquired	\$ 2,140

Goodwill has been recognized for the amount of the excess of the purchase price paid over the fair market value of the net assets

^{**} All options during the periods indicated have been granted at the market price on the date of grant, which is established by averaging the closing price of the Company's common stock over the ten trading days preceding the date of grant.

acquired. In accordance with SFAS No. 142, "Goodwill and Other Intangible Assets," goodwill will not be amortized; however, it will be tested at least annually for impairment.

NOTE C – INVENTORIES

Inventories consist of the following:

	 anuary 3, 2004	 July 3, 2004
Raw materials	\$ 5,498	\$ 5,799
Work in progress	2,497	2,350
Finished goods	7,563	6,455
	15,558	14,604
Less allowance for inventory valuation	 1,489	1,030
	\$ 14,069	\$ 13,574

NOTE D – PREPAID EXPENSES AND OTHER CURRENT ASSETS

Prepaid expenses and other current assets consist of the following:

		January 3, 2004	July 3, 2004
Prepaid expenses			
Miscellaneous receivables, net	:	\$ 1,376	\$ 1,127
Other current assets		1,449	1,959
		469	572
	:	\$ 3,294	\$ 3,658
1	11		

NOTE E – PROPERTY AND EQUIPMENT

Cost of property and equipment and their estimated useful lives is as follows:

	Years	J:	anuary 3, 2004	 July 3, 2004
Buildings	40	\$	8,120	\$ 9,283
Laboratory and production equipment	5-7		6,879	7,897
Sound and video library	5		_	600
Computer equipment and software	3-5		18,702	21,782
Furniture and fixtures	3-5		2,227	2,290
Automobiles	3-5		180	200
Leasehold improvements	3-5		1,626	2,282
Land improvements	15		931	931
			38,665	45,265
Less accumulated depreciation and amortization			21,740	23,718
			16,925	21,547
Land			1,773	1,899
Deposits and projects in process			1,497	616
		\$	20,195	\$ 24,062

NOTE F - GOODWILL

Goodwill represents the excess of the purchase price paid of acquired entities over the fair market value of the net assets acquired. As of July 3, 2004, goodwill totaled \$5,690, comprised of \$4,267 associated with the July 1, 2003 acquisition of Wasatch Product Development, Inc. (WPD) and \$1,423 in connection with the February 1, 2004 acquisition of FMG. No events have occurred subsequent to either acquisition that have resulted in an impairment of the original goodwill amounts initially recorded from the

transactions. In accordance with SFAS No. 142, "Goodwill and Other Intangible Assets," goodwill must be tested at least annually and if the carrying amount of goodwill exceeds its fair value, an impairment loss must be recognized in an amount equal to that excess.

During June 2004, an independent third party conducted the annual impairment test of goodwill related to the acquisition of WPD. The fair market value of the net assets of WPD was estimated using widely accepted valuation methods, including both a market approach and an income approach. In determining the fair market value as part of the impairment test, certain assumptions were used to project future results that management believes are reasonable, given current facts and circumstances; however, there can be no assurance that, under the assumptions used, these projections will materialize. Based upon the results of the independent appraisal, the fair market value of the net assets of WPD has been determined to be in excess of the carrying amount of the net assets, and, therefore, no impairment loss for goodwill has been recognized.

The changes in the carrying amount of goodwill by acquired subsidiary for the six months ended July 3, 2004, are as follows:

		WPD	FMG		Consolidated Total	
Balance at January 3, 2004	!	\$ 4,267	\$	_	\$	4,267
Goodwill acquired		_		1,423		1,423
Impairment adjustments	_	_		_		
Balance at July 3, 2004		\$ 4,267	\$	1,423	\$	5,690
	12					

NOTE G - OTHER CURRENT LIABILITIES

Other current liabilities consist of the following:

	January 3, 2004		July 3, 2004	
	ф	2 (02	Φ	2 200
Associate incentives	\$	2,692	\$	2,390
Accrued compensation		4,186		2,988
Income taxes		1,255		1,540
Sales taxes		1,667		1,769
Associate promotions		29		1,515
Deferred revenue		1,404		1,788
Provision for returns and allowances		998		1,142
Accrued loss on foreign currency forwards		337		_
All other		2,136		2,065
	\$	14,704	\$	15,197

NOTE H - COMMON STOCK AND EARNINGS PER SHARE

Basic earnings per share are based on the weighted average number of shares outstanding for each period. Weighted average shares redeemed have been included in the calculation of weighted average shares outstanding for basic earnings per share. Diluted earnings per common share are based on shares outstanding (computed under basic EPS) and potentially dilutive shares. Shares included in dilutive earnings per share calculations include stock options granted that are in the money but have not yet been exercised.

	For the Quarter Ended			Ended
	J	une 28, 2003		July 3, 2004
Earnings available to common shareholders	\$	4,349	\$	7,413
Basic EPS				
Shares				
Common shares outstanding entire period		18,273		19,470
Weighted average common shares:				
Issued during period		864		180
Canceled during period		(50)		(451)
Weighted average common shares outstanding during period		19,087		19,199
Earnings per common share - basic	\$	0.23	\$	0.39
<u>Diluted EPS</u>				
Shares				

Divigite defererages that as point at and ing during period - basic	12,983	1 9,32 9
Weighted average shares outstanding during period - diluted	21,450	20,523
Earnings per common share - diluted	\$ 0.20	\$ 0.36

13

Options to purchase 390 shares of stock were not included in the computation of EPS for the quarter ended July 3, 2004 due to their exercise price being greater than the average market price of the shares. For the quarter ended June 28, 2003, all options were included in the computation of EPS.

	For th	For the Six Months Ended			
	June 2 2003	,	July 3, 2004		
Earnings available to common shareholders	\$	7,924 \$	13,626		
Basic EPS					
Shares					
Common shares outstanding entire period	1	8,273	19,470		
Weighted average common shares:	1	0,273	17,470		
Issued during period		605	104		
Canceled during period		(25)	(286		
Weighted average common shares outstanding during period	1	8,853	19,288		
Earnings per common share - basic	\$	0.42 \$	0.71		
Diluted EPS					
Shares					
Weighted average shares outstanding during period - basic		8,853	19,288		
Dilutive effect of stock options		2,394	1,400		
Weighted average shares outstanding during period - diluted	2	1,247	20,688		
Earnings per common share - diluted	\$	0.37 \$	0.66		

Options to purchase 290 shares of stock were not included in the computation of EPS for the six months ended July 3, 2004 due to their exercise price being greater than the average market price of the shares. For the six months ended June 28, 2003, all options to purchase shares of stock were included in the computation of EPS.

During the six months ended July 3, 2004, the Company expended \$14,897 to purchase 524 shares under the Company's share repurchase plan. The purchase of shares under this plan reduces the number of shares issued and outstanding.

NOTE I – SEGMENT INFORMATION

The Company's operations are distinguished by markets served and method of distribution employed and are classified into two reportable business segments: Direct Selling and Contract Manufacturing. These operating segments are evaluated regularly by management in determining the allocation of resources and in assessing the performance of the Company. Management evaluates performance based on net sales and the amount of operating income or loss.

Segment profit or loss is based on profit or loss from operations before income taxes. Interest income and expense, as well as income taxes, are not included in the Company's determination of segment profit or loss in assessing the performance of a segment.

Prior to the quarter ended September 27, 2003, the Company was only engaged in a single line of business, which was developing, manufacturing, and distributing nutritional and personal care products through a network marketing system. As such,

14

only one business segment was reported, which was distinguished by geography. During July 2003, Contract Manufacturing was added as a new business segment. This change does not affect the presentation of historical segment information.

Direct Selling

The Direct Selling segment comprises the Company's principal line of business: developing, manufacturing, and distributing nutritional and personal care products. Products are distributed through a network marketing system using independent distributors referred to as "Associates." Products are also sold directly to "Preferred Customers" who purchase products for personal use and are not permitted to resell or distribute the products. Sales to Associates and Preferred Customers are reported for seven operating geographic regions including North America, Australia-New Zealand, Hong Kong, Japan, Taiwan, South Korea, and Singapore.

Operating activities for the Contract Manufacturing segment include the manufacture of premium personal care products, produced under the brand name of its customers, including manufacturing and packaging for the Company's Sensé product line of skin and personal care products. Operations are located in Draper, Utah and sales are made to a limited number of customers.

Sales made by the Contract Manufacturing segment to one customer accounted for 78%, or approximately \$2,420, of segment revenues for the quarter ended July 3, 2004. No other individual customer accounted for 10% or more of segment net revenues.

Financial information summarized by operating segment and geographic region for the quarters ended June 28, 2003 and July 3, 2004 is listed below:

		evenues from External Customers	Intersegment Revenues		Earnings before Income Taxes	
Quarter ended June 28, 2003:						
North America	\$	33,155	\$	8,543	\$ 6,620	
Australia - New Zealand		7,084		503	823	
Hong Kong		2,105		_	346	
Japan (2)		1,445		_	(748)	
Taiwan		3,368		_	932	
Reportable Segments Total		47,157		9,046	 7,973	
Unallocated and Other (4)		<u> </u>		(9,046)	(1,070)	
Consolidated Total	\$	47,157	\$	<u> </u>	\$ 6,903	
	15					

	F	Revenues from External Customers	Intersegment Revenues			Earnings before Income Taxes	
Quarter ended July 3, 2004:							
Direct Selling							
North America (1)	\$	42,415	\$	11,598	\$	11,389	
Australia - New Zealand		8,471		912		514	
Hong Kong		2,750		_		277	
Japan (2)		2,176		_		(602)	
Taiwan		3,898		_		(76)	
South Korea		1,804		_		(66)	
Singapore		2,612	_		_	234	
Segment Total		64,126		12,510		11,670	
Contract Manufacturing (3)		3,120		192		190	
Reportable Segments Total		67,246		12,702		11,860	
Unallocated and Other (4)		<u> </u>		(12,702)	_	(629)	
Consolidated Total	\$	67,246	\$	_	\$	11,231	

Financial information summarized by operating segment and geographic region for the six months ended June 28, 2003 and July 3, 2004 is listed below:

Six months ended June 28, 2003:	Revenues from External Customers		Intersegment Revenues		Earnings before Income Taxes		Long-lived Assets		T	otal Assets
North America Australia - New Zealand	\$	62,854 12,407	\$	13,275 913	\$	12,154 1,510	\$	22,591 286	\$	42,291
Hong Kong		3,995		— —		572		201		2,698 1,150
Japan (2) Taiwan		2,683 6,082		— —		(1,519) 1,486		1,189 341		2,072 2,749
Reportable Segments Total		88,021		14,188		14,203		24,608		50,960

Unallocated and Other (4)	 	_	(14,188)	 (1,625)	(3,809)	 (3,118)
Consolidated Total	\$ 88,021	\$		\$ 12,578	\$ 20,799	\$ 47,842
			16			

	l	enues from External ustomers	I	ntersegment Revenues	be	Earnings efore Income Taxes	1	Long-lived Assets	7	Γotal Assets
Six months ended July 3, 2004:									1	
Direct Selling										
North America (1)	\$	82,701	\$	20,921	\$	22,018	\$	37,733	\$	55,049
Australia - New Zealand		16,747		1,584		(294)		223		5,038
Hong Kong		5,207		_		485		210		2,977
Japan (2)		4,392		_		(1,065)		1,009		2,396
Taiwan		7,627		_		(562)		368		2,046
South Korea		3,074		_		(710)		797		2,206
Singapore		4,619	_	_		460		295		2,631
Segment Total		124,367		22,505		20,332		40,635		72,343
Contract Manufacturing (3)		4,654		680		(124)		5,923		9,510
Reportable Segments Total		129,021		23,185		20,208		46,558		81,853
Unallocated and Other (4)		_		(23,185)		437		(14,308)		(16,904)
Consolidated Total	\$	129,021	\$	_	\$	20,645	\$	32,250	\$	64,949

⁽¹⁾ Includes results from the FMG subsidiary acquired in February 2004 and operations in Mexico initiated in March 2004.

17

Item 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following discussion and analysis of USANA's financial condition and results of operations should be read in conjunction with the Unaudited Consolidated Financial Statements and Notes thereto contained in this quarterly report.

General

USANA Health Sciences, Inc. develops and manufactures high-quality nutritional and personal care products. We market all of our products on the basis of high levels of bioavailability, safety, and quality. We distribute our products through a network marketing system using independent distributors that we refer to as "Associates." As of July 3, 2004, we had 104,000 active Associates in the United States, Canada, Australia, New Zealand, Hong Kong, Japan, Taiwan, South Korea, Singapore, Mexico, and the United Kingdom. We also sell products directly to "Preferred Customers" who purchase product for personal use and are not permitted to resell or distribute the products. As of July 3, 2004, we had 59,000 active Preferred Customers worldwide. Sales to Preferred Customers accounted for approximately 15% of net sales for the Direct Selling segment during the second quarter of 2004. For purposes of this report, we only count as active customers those Associates and Preferred Customers who have purchased product from USANA at any time during the most recent three-month period.

The fiscal year end of USANA is the Saturday closest to December 31 of each year. Fiscal year 2004 will end on January 1, 2005 and is a 52-week year. Fiscal year 2003 ended on January 3, 2004 and was a 53-week year.

As discussed more fully in Note I – Segment Information, beginning on page 14 to the consolidated financial statements, we have two reportable segments: Direct Selling and Contract Manufacturing. The Direct Selling segment constitutes our principal line of business: developing, manufacturing, and distributing nutritional and personal care products through a network marketing system. The Contract Manufacturing segment includes the manufacture of premium personal care products, produced under the brand name of its

⁽²⁾ Includes results from local operations in Japan. Direct U.S. export sales to Japan are included in the North America geographic region.

⁽³⁾ Reportable business activities for the Contract Manufacturing segment commenced July 1, 2003.

^{(4) &}quot;Unallocated and Other" includes certain corporate items and eliminations that are not allocated to the operating segments.

customers, including manufacturing and packaging for the Company's Sensé product line of skin and personal care products.

Our primary product lines within the Direct Selling segment consist of USANA $^{\hat{a}}$ Nutritionals and Sensé – beautiful science $^{\hat{a}}$ (Sensé). The USANA $^{\hat{a}}$ Nutritionals product line is further categorized into three separate classifications: Essentials, Optimizers, and Macro Optimizers. Prior to the third quarter of 2003, the USANA $^{\hat{a}}$ Nutritionals product line was comprised of the Essentials and Optimizers product categories. At our Annual International Convention in September 2003, we announced that the $L \cdot E \cdot A \cdot N$ Lifelong \hat{O} brand name would be discontinued and that the weight management products associated with this brand were to be sold under the new Macro Optimizers classification within the USANA $^{\hat{a}}$ Nutritionals product line. This transition was made to simplify the overall product story, which our Associates share with prospects. The change includes a shift away from a low-fat/weight-loss positioning to a focus on low-glycemic carbohydrates, soy protein and dietary fiber, as well as the specific health benefits associated with these ingredients.

USANA^â Nutritionals.

The Essentials include core vitamin and mineral supplements that provide a foundation of advanced nutrition for every age group. To help meet the "essential" nutrient needs of children and teens during the years of development, when good nutrition is most important, USANA offers: UsanimalsÔ, a great-tasting formulation of vitamins, minerals, and antioxidants, in an easy-to-take chewable tablet for children 13 months to 12 years old and Body RoxÔ, a nutritional supplement containing 31 essential vitamins, minerals, antioxidants, and cofactors for adolescents 12 to 18 years old. USANA^â Essentials for adults is a combination of two products: Mega Antioxidant, a balanced, high-potency blend of 30 vitamins, antioxidants, and other important nutrients to support cellular metabolism and to counteract free-radical damage and Chelated Mineral, a complete spectrum of essential minerals, in balanced, highly bioavailable forms. The USANA^â Essentials are also provided in a convenient pillow pack format, HealthPak 100Ô.

Optimizers are more targeted supplements designed to meet individual health and nutritional needs. Products in this category include Proflavanol^â, Poly C^â, Procosa^â II, CoQuinone^â 30, BiOmega-3Ô, E-PrimeÔ, Active CalciumÔ, PhytoEstrinÔ, Palmetto PlusÔ, Ginkgo-PSÔ, Garlic ECÔ, VisionexÔ, and OptOmega^â.

18

The Macro Optimizers include healthy convenience foods and other related products, including powdered drink mixes and nutrition bars that were previously sold under the $L \cdot E \cdot A \cdot N$ Lifelong \hat{O} brand name. Nutrimeal \hat{O} and Fibergy $^{\hat{a}}$ drink mixes, SoyaMax \hat{O} , and Nutrition Bar and Fibergy Bar are included in this product category.

Sensé - beautiful scienceâ

The Sensé product line includes premium, science-based personal care products that support healthy skin and hair by providing advanced topical nourishment, moisturization and protection. Products in this line include Perfecting Essence, Gentle Daily Cleanser, Hydrating Toner, Daytime Protective Emulsion SPF 15, Eye Nourisher, Night Renewal (Replenishing Crème), Serum Intensive (Skin Revival Complex), Rice Bran Polisher, Nutritious Crème Masque, Revitalizing Shampoo, Nourishing Conditioner, Firming Body Nourisher, and Energizing Shower Gel.

All Other

In addition to these principal product lines, we have developed and sell to Associates materials and online tools designed to assist them in building their business and selling products. These resource materials or sales aids include product brochures and business forms designed by us and printed by outside publishers. We periodically contract with authors and publishers to produce or provide books, tapes, and other items dealing with health topics and personal motivation, which are made available to Associates. We also write and develop our own materials for audio and videotapes, which are produced by FMG and third parties. New Associates are required to purchase a starter kit containing USANA training materials that assist the Associates in starting and growing their business. Associates do not earn commissions on the sale of sales aids or starter kits.

The following table summarizes the approximate percentage of total product revenue for the Direct Selling segment contributed by major product line for the six months ended as of the dates indicated:

	Sales By Product Line * Six Months Ended					
Product Line	June 28, 2003	July 3, 2004				
USANA ^â Nutritionals						
Essentials **	39%	38%				
Optimizers	34%	34%				
Macro Optimizers	8%	10%				
Sensé – beautiful science ^â	14%	13%				
All Other	5%	5%				

^{*} Product sales previously categorized as Combination Packs have been allocated to their respective product lines based on the weighted average price of the product components that comprise each pack.

** The Essentials category under the USANA^â Nutritionals product line includes USANA^â Essentials, HealthPak 100Ô, Body RoxÔ, and UsanimalsÔ

Key Products

The following highlights sales data for our top-selling products as a percentage of Direct Selling segment product sales for the six months ended July 3, 2004.

USANA ^â Essentials	25%
HealthPak 100Ô	11%
Proflavanol ^â	10%

19

Results of Operations

Quarters Ended June 28, 2003 and July 3, 2004

Net Sales. Net sales increased 42.6% to \$67.2 million for the quarter ended July 3, 2004, an increase of \$20.0 million from \$47.2 million for the comparable quarter in 2003. The increase was comprised of \$16.9 million associated with our Direct Selling segment and \$3.1 million associated with our Contract Manufacturing segment acquired in July 2003.

The following table summarizes the growth in net sales by segment and geographic region for the fiscal quarters ended June 28, 2003 and July 3, 2004.

		Sal	es By Segment and (in thousands Quarter Ende)	C	hange from	Percent	
Segment / Region	. =	June 28, 2003		July 3, 2004	1 1	Prior Year	Change	
Direct Selling								
United States	\$	21,884	46.4%\$	27,821	41.4%\$	5,937	27.1%	
Canada		11,271	23.9%	12,378	18.4%	1,107	9.8%	
Australia-New Zealand		7,084	15.0%	8,471	12.6%	1,387	19.6%	
Hong Kong		2,105	4.5%	2,750	4.1%	645	30.6%	
Japan		1,445	3.1%	2,176	3.2%	731	50.6%	
Taiwan		3,368	7.1%	3,898	5.8%	530	15.7%	
South Korea		_	0.0%	1,804	2.7%	1,804	N/A	
Singapore		_	0.0%	2,612	3.9%	2,612	N/A	
Mexico		_	0.0%	2,216	3.3%	2,216	N/A	
Segment Total		47,157	100.0%	64,126	95.4%	16,969	36.0%	
			0.00/	2.120	4.60/	2 120	>T/A	
Contract Manufacturing			0.0%	3,120	4.6%	3,120	N/A	
Consolidated	\$	47,157	100.0 % \$	67,246	100.0 % \$	20,089	42.6%	

The increase in net sales contributed by the Direct Selling segment can be primarily attributed to the following factors:

- A 16.9% increase in the active Associate base and an 18.0% increase in the active Preferred Customer base for the second quarter of 2004 in markets that have been open longer than one year,
- Sales from South Korea, Singapore, and Mexico, which were not open during the second quarter of 2003, and
- Stronger foreign currencies, relative to the U. S. dollar, which positively affected the translation of sales in foreign currencies by \$1.5 million.

Contract Manufacturing net sales for the quarter ended July 3, 2004 were positively affected by a higher than normal backlog of sales orders accumulated in the first quarter, during facility construction upgrades, which were subsequently filled in the second quarter. We believe that Contract Manufacturing net sales for the third quarter 2004 will be approximately \$2.0 million.

Based on information currently available to the Company, we expect consolidated net sales to approach \$69 million for the third quarter of 2004. We expect consolidated net sales to approach \$270 million for fiscal year 2004.

Active Associates By Region

. .	As of		As of		Change from	Percent
Region	June 28, 200	3	July 3, 200)4	Prior Year	Change
United States	32,000	41.5%	40,000	38.5%	8,000	25.0%
Canada	18,000	23.4%	20,000	19.2%	2,000	11.1%
Australia-New Zealand	12,000	15.6%	13,000	12.5%	1,000	8.3%
Hong Kong	4,000	5.2%	5,000	4.8%	1,000	25.0%
Japan	3,000	3.9%	4,000	3.8%	1,000	33.3%
Taiwan	8,000	10.4%	8,000	7.7%	_	0.0%
South Korea	_	0.0%	3,000	2.9%	3,000	N/A
Singapore	_	0.0%	6,000	5.8%	6,000	N/A
Mexico	_	0.0%	5,000	4.8%	5,000	N/A
Total	77,000	100.0%	104,000	100.0%	27,000	35.1%

We believe that various factors contributed to the year-over-year second quarter increase in the active Associate base, including new market openings, ongoing communication with Associate leaders in the field, improved infrastructure to enhance the Associate service level, and company-sponsored events and promotions held to motivate Associates.

Active Preferred Customers By Region

Region	As June 28			of 5, 2004	Change from Prior Year	Percent Change
United States	30,000	60.0%	36,000	61.0%	6,000	20.0%
Canada	15,000	30.0%	16,000	27.1%	1,000	6.7%
Australia-New Zealand	4,000	8.0%	5,000	8.5%	1,000	25.0%
Hong Kong	1,000	2.0%	1,000	1.7%	_	0.0%
Japan	**	0.0%	**	0.0%	_	N/A
Taiwan	**	0.0%	1,000	1.7%	1,000	N/A
South Korea	_	0.0%	**	0.0%	_	N/A
Singapore	_	0.0%	**	0.0%	_	N/A
Mexico	_	0.0%	**	0.0%	_	N/A
Total	50,000	100.0%	59,000	100.0%	9,000	18.0%

^{**}Active Preferred Customer count is less than 500.

Total Active Customers By Region

Region	As of June 28, 20	03	As of July 3, 200	04	Change from Prior Year	Percent Change
United States	62,000	48.8%	76,000	46.6%	14,000	22.6%
Canada	33,000	26.0%	36,000	22.1%	3,000	9.1%
Australia-New Zealand	16,000	12.6%	18,000	11.0%	2,000	12.5%
Hong Kong	5,000	3.9%	6,000	3.7%	1,000	20.0%
Japan	3,000	2.4%	4,000	2.5%	1,000	33.3%
Taiwan	8,000	6.3%	9,000	5.5%	1,000	12.5%
South Korea	_	0.0%	3,000	1.8%	3,000	N/A
Singapore	_	0.0%	6,000	3.7%	6,000	N/A
Mexico	_	0.0%	5,000	3.1%	5,000	N/A
Total	127,000	100.0%	163,000	100.0%	36,000	28.3%
		21				

Gross Profit. Consolidated gross profit decreased to 75.9% of net sales for the quarter ended July 3, 2004 from 77.9% for the comparable quarter in 2003. This decrease in consolidated gross profit can be primarily attributed to increased sales from our Contract Manufacturing segment, which has significantly lower gross profit margins than that of our Direct Selling segment.

Gross profit for the Contract Manufacturing segment was 16.3% of net segment sales for the quarter ended July 3, 2004. We anticipate modest sequential improvements to the gross profit margin in our Contract Manufacturing segment during the remainder of fiscal 2004.

Gross profit in the Direct Selling segment, for the quarter ended July 3, 2004, improved to 78.8% of net segment sales from 77.9% for the comparable quarter in 2003. This increase primarily resulted from improved cost efficiencies generated from procurement and production activities and leverage benefits of fixed costs on a rising sales base.

We continued to encounter higher purchase prices for one of our raw materials, CoQ10, during the quarter ended July 3, 2004, due to a persistent shortage in supply. We expect sustained pressure on the gross profit margin for our Direct Selling segment from higher purchase prices for CoQ10, until suppliers re-tool their manufacturing facilities to increase production capacity in order to meet rising demand.

Associate Incentives. Expenses related to Associate incentives are incurred only by the Direct Selling segment and represent the most significant cost as a percentage of net sales for this segment. Associate incentives increased to 39.9% of net segment sales during the second quarter of 2004, compared to 39.6% for the second quarter in 2003. The modest increase in Associate incentives relative to net sales was the result of various incentive programs that took place during the current year quarter.

We believe that Associate incentives as a percentage of net segment sales for the third quarter of 2004 will be slightly lower due to an anticipated increase in sales of non-commissionable items such as sales aids and logo merchandise at our Annual International Convention, which will be held in September 2004.

Selling, General and Administrative Expenses. Selling, general and administrative expense decreased to 20.3% of net sales for the quarter ended July 3, 2004 from 22.4% for the comparable quarter in 2003. The decrease, as a percentage of net sales, can be attributed to leverage generated on an increasing sales base and lower relative selling, general and administrative expense in our Contract Manufacturing segment.

In absolute terms, selling, general and administrative expenses increased by \$3.1 million for the quarter ended July 3, 2004 when compared to second quarter of 2003. The increase in selling, general and administrative expenses can be primarily attributed to an increase in spending in many of our markets to support a growing Associate base and related sales, and spending in our newer markets of South Korea, Singapore, and Mexico.

We believe that selling, general and administrative expenses, as a percentage of net sales during the third quarter 2004, will be modestly higher than second quarter 2004 due to increased expenses related to our upcoming Annual International Convention in September of 2004 and lower third quarter sales expected from our Contract Manufacturing segment.

Other Income (Expense). Other income (expense) improved by \$227,000 in the second quarter of 2004 when compared to the second quarter of 2003. This increase in net other income can be primarily attributed to lower foreign currency losses incurred during the current year quarter.

Net Earnings. Net earnings increased 70.5% to \$7.4 million for the quarter ended July 3, 2004, an increase of \$3.1 million from \$4.3 million for the comparable quarter in 2003. The increase in net earnings can be attributed primarily to higher net sales and lower relative selling, general, and administrative expenses.

Diluted earnings per share improved to \$0.36 for the second quarter of 2004, an increase of \$0.16, or 80.0%, from the \$0.20 reported for the comparable quarter in 2003.

We expect earnings per share for the third quarter 2004 to be in the range of \$0.36 to \$0.38. For the fiscal year ended 2004, we expect earnings per share between \$1.42 and \$1.44.

22

Six Months Ended June 28, 2003 and July 3, 2004

Net Sales. Consolidated net sales increased 46.6% to \$129.0 million for the six months ended July 3, 2004, an increase of \$41.0 million from \$88.0 million for the comparable six-month period in 2003. The increase primarily consisted of \$36.3 million attributable to the Direct Selling segment. The Contract Manufacturing segment acquired at the beginning of the third quarter 2003 contributed \$4.7 million to the increase in consolidated net sales.

The following table summarizes the growth in net sales by segment and geographic region for the six months ended June 28, 2003 and July 3, 2004.

Sales By Segment and Region (in thousands)

			(in thousand Six Months Ei	,		Change from	Percent
Segment / Region		June 28		July 3,	2004	Prior Year	Change
Direct Selling							
United States	\$	41,873	47.6%\$	54,922	42.6%	\$ 13,049	31.2%
Canada		20,981	23.8%	24,815	19.2%	3,834	18.3%
Australia-New Zealand		12,407	14.1%	16,747	13.0%	4,340	35.0%
Hong Kong		3,995	4.5%	5,207	4.0%	1,212	30.3%
Japan		2,683	3.0%	4,392	3.4%	1,709	63.7%
Taiwan		6,082	7.0%	7,627	5.9%	1,545	25.4%
South Korea		_	0.0%	3,074	2.4%	3,074	N/A
Singapore		_	0.0%	4,619	3.6%	4,619	N/A
Mexico			0.0%	2,964	2.3%	2,964	N/A
Segment Total		88,021	100.0%	124,367	96.4%	36,346	41.3%

Contract Manufacturing		0.0%	4,654	3.6%	4,654	N/A
Consolidated	\$ 88,021	100.0%\$	129,021	100.0%\$	41,000	46.6%

The increase in net sales contributed by the Direct Selling segment can be primarily attributed to the following factors:

- An increase in the active customer base in markets that have been open longer than one year,
- · Sales from South Korea, Singapore, and Mexico, which were not open during the second quarter of 2003, and
- Stronger foreign currencies, relative to the U. S. dollar, which positively affected the translation of sales in foreign currencies by \$5.2 million.

Gross Profit. Consolidated gross profit decreased to 75.8% of net sales for the six months ended July 3, 2004 from 77.7% for the comparable period in 2003. Direct Selling's gross profit improved modestly to 78.2% of net sales for the segment from 77.7% for the six months ended June 28, 2003, primarily resulting from improved cost efficiencies generated from procurement and production activities and leverage benefits of fixed costs on a rising sales base.

Gross profit for the Contract Manufacturing segment was 10.0% for the six months ended July 3, 2004. This relatively low gross profit margin was primarily a result of construction upgrades that took place during the first quarter of 2004 to get the manufacturing facility operating at the pharmaceutical level for good manufacturing practices.

Associate Incentives. Expenses related to Associate incentives are incurred only by the Direct Selling segment and represent the most significant cost as a percent of net sales for the segment. Associate incentives remained constant at 39.5% of net segment sales for the six months ended July 3, 2004 and the comparable period in 2003.

23

Selling, General and Administrative Expenses. Selling, general and administrative expense decreased to 20.9% of net sales for the six months ended July 3, 2004 from 22.9% for the comparable period in 2003. The decrease, as a percentage of net sales, can be primarily attributed to leverage generated on an increasing sales base.

In absolute terms, selling, general and administrative expenses increased by \$6.8 million for the six months ended July 3, 2004, when compared to the first six months of 2003, the majority of which can be attributed to the Direct Selling segment. The increase in selling, general and administrative expenses can be primarily attributed to an increase in spending in many of our markets to support a growing Associate base and related sales, as well as spending in our newer markets of South Korea and Singapore.

Other Income (Expense). Other income (expense) changed from net other expense of \$194,000 for the first six months of 2003 to net other income of \$148,000 for the comparable period in 2004. This improvement in net other income of \$342,000 can, in great part, be attributed to a change from foreign currency losses for the first six months of 2003 compared to modest foreign currency gains for the comparable period of 2004. Additionally, lower interest expense and higher interest income also contributed to the improvement in net other income.

Income Taxes. Income taxes totaled 34.0% of earnings before income taxes for the first six months of 2004, compared to 33.5% for the full fiscal year 2003. The increase in the effective tax rate for the first six months of 2004, compared to the effective tax rate for the full fiscal year 2003, was primarily due to an increase in the effective federal income tax rate from 34% to 35% because of higher expected earnings and the utilization of a foreign tax credit carryforward in the tax year 2003 that will not occur again in the tax year 2004. The foreign tax credit carryforward utilized in 2003 lowered the effective tax rate as a result of decreasing a valuation allowance that had previously provided for the potential non-recovery of the foreign tax credit carryforwards prior to expiration.

Net Earnings. Net earnings increased 72.0% to \$13.6 million for the six months ended July 3, 2004, an increase of \$5.7 million from \$7.9 million for the comparable period in 2003. The increase in net earnings can be attributed primarily to increased net sales and lower relative selling, general and administrative expenses.

Diluted earnings per share improved to \$0.66 for the first six months of 2004, an increase of \$0.29, or 78.4%, from the \$0.37 reported for the comparable period in 2003.

Liquidity and Capital Resources

We have historically financed growth with cash flows from operations. In the first six months of 2004, net cash flows from operating activities totaled \$16.9 million, compared to \$17.0 million for the same period in 2003. Cash and cash equivalents decreased to \$13.8 million at July 3, 2004 from \$19.0 million at January 3, 2004. Additionally, net working capital declined to \$13.4 million at July 3, 2004, compared to net working capital of \$18.3 million at January 3, 2004. The decrease in cash and cash equivalents and net working capital during the first six months of 2004 can be primarily attributed to the purchase of shares under the Company's share repurchase plan totaling \$14.9 million.

On June 16, 2004, we entered into a new agreement with a financial institution to provide a credit facility consisting of a \$10 million two-year revolving line of credit. The credit facility has been secured through a pledge of the capital stock of first-tier material subsidiaries. At July 3, 2004, we had no amounts outstanding on the line of credit. The credit facility contains restrictive covenants

requiring that we maintain certain financial ratios. As of July 3, 2004, we were in compliance with these covenants.

We believe that current cash balances, cash provided by operations, and amounts available under the line of credit will be sufficient to cover our capital needs in the ordinary course of business for the foreseeable future. If we experience an adverse operating environment or unusual capital expenditure requirements, additional financing may be required. However, no assurance can be given that additional financing, if required, would be available on favorable terms. We might also require or seek additional financing for the purpose of expanding new markets, growing our existing markets, and for other reasons. Such financing may include the sale of additional equity securities. Any financing which involves the sale of equity securities or instruments convertible into equity securities could result in immediate and possibly significant dilution to existing shareholders.

24

Forward-Looking Statements

The statements contained in this report that are not purely historical are considered to be "forward-looking statements" within the meaning of the Private Securities Litigation Reform Act of 1995 and Section 21E of the Securities Exchange Act. These statements represent our expectations, hopes, beliefs, anticipations, commitments, intentions and strategies regarding the future. They may be identified by the use of words or phrases such as "believes," "expects," "anticipates," "should," "plans," "estimates," and "potential," among others. Forward-looking statements include, but are not limited to, statements contained in Management's Discussion and Analysis of Financial Condition and Results of Operations regarding our financial performance, revenue and expense levels in the future and the sufficiency of our existing assets to fund future operations and capital spending needs. Readers are cautioned that actual results could differ materially from the anticipated results or other expectations expressed in these forward-looking statements for the reasons detailed in our most recent Annual Report on Form 10-K at pages 32 through 39. The fact that some of the risk factors may be the same or similar to our past reports filed with the Securities and Exchange Commission means only that the risks are present in multiple periods. We believe that many of the risks detailed here and in the Company's other SEC filings are part of doing business in the industry in which we operate and compete and will likely be present in all periods reported. The fact that certain risks are endemic to the industry does not lessen their significance. The forward-looking statements contained in this report are made as of the date of this report and we assume no obligation to update them or to update the reasons why actual results could differ from those projected in such forward-looking statements. Among others, risks and uncertainties that may affect our business, financial condition, performance, development and results of operations includ

- Our ability to attract and maintain a sufficient number of Associates,
- High turnover of Associates,
- Our dependence upon a network marketing system to distribute our products,
- Activities of our independent Associates,
- Risks related to our expansion into international markets,
- Rigorous government scrutiny of network marketing practices,
- Potential effects of adverse publicity regarding nutritional supplements or the network marketing industry,
- Reliance on key management personnel, including our Founder, Chairman of the Board of Directors, and Chief Executive Officer Myron W. Wentz, Ph.D.,
- Extensive government regulation of the Company's products and manufacturing,
- Potential inability to sustain or manage growth, including the failure to continue to develop new products,
- Our reliance on information technology,
- The adverse effect of the loss of a high-level sponsoring Associate together with a group of leading Associates in that person's downline,
- The loss of product market share or Associates to competitors,
- Potential adverse effects of taxation and transfer pricing regulations,
- The fluctuation in the value of foreign currencies against the US dollar,
- Our reliance on outside suppliers for raw materials,
- Product liability claims and other manufacturing activity risks,

- Intellectual property risks particularly applicable to our business, and
- Liability claims associated with our "Athlete Guarantee" program.

Item 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

We conduct our business in several countries and intend to continue to expand our foreign operations. Net sales, earnings from operations and net earnings are affected by fluctuations in currency exchange rates, interest rates and other uncertainties inherent in doing business and selling product in more than one currency. In addition, our operations are exposed to risks associated with changes in social, political and economic conditions inherent in foreign operations, including changes in the laws and policies that govern foreign investment in countries where we have operations as well as, to a lesser extent, changes in United States laws and regulations relating to foreign trade and investment.

Foreign Currency Risks. Consolidated net sales outside the United States represented 52.4% and 53.8% of net sales for the six months ended June 28, 2003 and July 3, 2004, respectively. Inventory purchases are transacted primarily in U.S. dollars from vendors located in the United States. The local currency of each international subsidiary is considered the functional currency, with all revenue and expenses translated at weighted average exchange rates for reported periods. Consequently, our reported sales and earnings are affected positively by a weakening of the U.S. dollar and negatively by a strengthening of the U.S. dollar. Changes in currency exchange rates affect the relative prices at which we sell our products. Given the uncertainty of exchange rate fluctuations, we cannot estimate the effect of these fluctuations on our future business, product pricing, results of operations, or financial condition.

We seek to reduce exposure to fluctuations in foreign exchange rates by creating offsetting positions through the use of foreign currency exchange contracts. We do not use derivative financial instruments for trading or speculative purposes. We enter into forward and option contracts to hedge expected net cash flows from our foreign affiliates, which is primarily represented by intercompany cash transfers. During the six months ended July 3, 2004, forward and option contracts were in place to offset exposure to the Canadian Dollar, Australian Dollar, New Zealand Dollar, and New Taiwan Dollar.

As a last recourse for hedging currency risk, we may elect to adjust prices in non-U.S. markets to reflect changes in foreign currency exchange rates. However, there can be no assurance that these practices will be successful in eliminating all or substantially all of the risks encountered in connection with our foreign currency transactions.

Following are the average exchange rates of foreign currency units to one U.S. dollar for each of our foreign markets for the periods ended as of the dates indicated:

	Quarte	Ended	Six Months Ended		
	June 28, 2003	July 3, 2004	June 28, 2003	July 3, 2004	
Canadian Dollar	1.40	1.36	1.45	1.34	
Australian Dollar	1.56	1.40	1.62	1.35	
New Zealand Dollar	1.76	1.59	1.79	1.54	
Hong Kong Dollar	7.80	7.80	7.80	7.80	
Japanese Yen	118.48	109.74	118.72	108.43	
New Taiwan Dollar	34.67	33.30	34.65	33.31	
Korean Won	*	1,162.47	*	1,166.37	
Singapore Dollar	*	1.70	*	1.70	
Mexican Peso **	*	11.41	*	11.32	

- * Market was not in operation during period indicated.
- ** The six-month 2004 Mexican Peso exchange rate represents the average for the first four months of Mexico operations that commenced in March 2004.

26

Interest Rate Risks. As of July 3, 2004, we had no outstanding debt and therefore, we currently have no direct exposure to interest rate risk. It may become necessary to borrow in the future in order to meet our financing needs, as circumstances require. In the event that it becomes necessary to finance with debt, there can be no assurance that we will be able to borrow at favorable rates.

Item 4. CONTROLS AND PROCEDURES

We maintain disclosure controls and procedures that are designed to ensure that information required to be disclosed in our Exchange Act reports is recorded, processed, summarized, and reported within the time periods specified in the SEC's rules and forms, and that such information is accumulated and communicated to management, including the Chief Executive Officer and Chief Financial Officer, as appropriate, to allow timely decisions regarding required disclosure. In designing and evaluating the disclosure controls and

procedures, management recognized that any controls and procedures, no matter how well designed and operated, can provide only reasonable assurance of achieving the desired control objectives, and management necessarily was required to apply its judgment in evaluating the cost-benefit relationship of possible controls and procedures.

- (a) Evaluation of Disclosure Controls and Procedures. As of the end of the period covered by this Quarterly Report on Form 10-Q, our Chief Executive Officer and Chief Financial Officer conducted an evaluation of the effectiveness of the design and operation of our disclosure controls and procedures (as such term is defined in Rule 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934, as amended). Based on the foregoing, the Chief Executive Officer and Chief Financial Officer concluded that the disclosure controls and procedures were effective.
- (b) Changes in Internal Controls. There were no significant changes in our internal controls or in other factors that could significantly affect those internal controls subsequent to the date of the evaluation, nor were there any significant deficiencies or material weaknesses in our internal controls. As a result, no corrective actions were required or undertaken.

PART II. OTHER INFORMATION

Item 2. CHANGES IN SECURITIES AND USE OF PROCEEDS

Purchases made during the quarter ended July 3, 2004 and for each fiscal month are summarized in the following table:

Issuer Purchases of Equity Securities (amounts in thousands, except per share data)

Period	Total Number of Shares Purchased	erage Price d per Share	Total Number of Shares Purchased as Part of Publicly Announced Plans or Programs	Val Pur	proximate Dollar lue of Shares that May Yet Be chased Under the ns or Programs *
April 4, 2004 through May 8, 2004 (Fiscal April)	195	\$ 28.48	195	\$	1,156
May 9, 2004 through June 5, 2004 (Fiscal May)	44	\$ 24.96	44	\$	70
June 6, 2004 through July 3, 2004 (Fiscal June)		\$ _		\$	70
Total	239	\$ 27.84	239		

^{*} At their July 2004 meeting, the Board of Directors approved an increase in the dollar amount that may be purchased under the Company's share repurchase plan from \$70.0 thousand up to \$10.0 million.

27

Item 4. SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS

At our Annual Meeting of Shareholders on April 21, 2004, the following actions were submitted and approved by vote of the shareholders:

- (1) Election of five directors, and
- (2) Ratification of the Board's selection of Grant Thornton LLP as our independent certified public accountants.

A total of 17,423,547 shares (approximately 90%) of the issued and outstanding shares of USANA were represented by proxy or in person at the meeting. These shares were voted on the matters described above as follows:

For the directors as follows:

Name	Number of Shares For	Number of Shares Abstaining/Withheld
Myron W. Wentz, PhD	17,246,021	177,526
Ronald S. Poelman	17,318,351	105,196
Robert Anciaux	17,311,598	111,949
Denis E. Waitley, PhD	17,231,731	191,816
Jerry G. McClain	17,318,109	105,438

2. For the ratification of the Board's selection of Grant Thornton LLP as the independent certified public accountants of USANA as follows:

Number of Shares	Number of Shares	Number of Shares
For	Against	Abstaining/Withheld
17,329,169	89,592	4,786

Item 6. EXHIBITS AND REPORTS ON FORM 8-K

(a) Exhibits.

Exhibit Number	Description
3.1	Articles of Incorporation [Incorporated by reference to Registration Statement on Form 10, File No. 0-21116, effective April 16, 1993]
3.2	Bylaws [Incorporated by reference to Registration Statement on Form 10, File No. 0-21116, effective April 16, 1993]
3.3	Amendment to Articles of Incorporation to change name and increase par value [Incorporated by reference to Report on Form 10-Q for the period ended July 1, 2000]
4.1	Specimen Stock Certificate for Common Stock, no par value [Incorporated by reference to Registration Statement on Form 10, File No. 0-21116, effective April 16, 1993]
10.1	Amended and Restated Long-Term Stock Investment and Incentive Plan [Incorporated by reference to Report on Form 10-Q for the period ended June 27, 1998]*
10.2	2002 USANA Health Sciences, Inc. Stock Option Plan [Incorporated by reference to Registration Statement on Form S-8, filed July 18, 2002]*
10.3	Credit Agreement by and between Bank of America, N.A. and USANA Health Sciences, Inc.
11.1	Computation of Net Income per Share (included in Notes to Consolidated Financial Statements)
	28
31.1	Certification of Chief Executive Officer pursuant to section 302 of the Sarbanes-Oxley Act of 2002
31.2	Certification of Chief Financial Officer pursuant to section 302 of the Sarbanes-Oxley Act of 2002
32.1	Certification of Chief Executive Officer pursuant to section 906 of the Sarbanes-Oxley act of 2002, 18 U.S.C. Section 1350
32.2	Certification of Chief Financial Officer pursuant to section 906 of the Sarbanes-Oxley act of 2002, 18 U.S.C. Section 1350

^{*} Denotes a management contract or compensatory plan or arrangement.

(b) Reports on Form 8-K.

On April 20, 2004, a Form 8-K was filed to disclose financial results for the first quarter 2004.

29

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

USANA HEALTH SCIENCES, INC.

Date: August 5, 2004 /s/ Gilbert A. Fuller

Gilbert A. Fuller Chief Financial Officer

(Principal Financial and Accounting Officer)

CREDIT AGREEMENT

Dated as of June 16, 2004

between

USANA HEALTH SCIENCES, INC.,

as Borrower,

BANK OF AMERICA, N.A.,

as Bank

TABLE OF CONTENTS



SECTION

ARTICLE I DEFINITIONS AND ACCOUNTING TERMS

	_ ~
1.01	Defined Terms

- 1.02 Other Interpretive Provisions
- 1.03 **Accounting Terms**
- 1.04 Rounding
- 1.05 References to Agreements and Laws
- Letter of Credit Amounts 1.06

ARTICLE II THE COMMITMENT AND CREDIT EXTENSIONS.

- 2.01
- 2.02 Borrowings, Conversions and Continuations of Loans.
- 2.03 Letters of Credit.
- 2.04 Prepayments.
- 2.05 2.06 Repayment of Loans
- Interest.
- 2.07 <u>Fees</u>
- 2.08 Computation of Interest and Fees
- 2.09 **Evidence of Debt**
- 2.10 Payments Generally.

ARTICLE III TAXES, YIELD PROTECTION AND ILLEGALITY

- 3.01 Taxes.
- 3.02
- 3.03 Inability to Determine Eurodollar Rates
- Increased Cost and Reduced Return; Capital Adequacy. 3.04
- **Funding Losses** 3.05
- Requests for Compensation 3.06
- 3.07 Survival

ARTICLE IV CONDITIONS PRECEDENT TO CREDIT EXTENSIONS

- Conditions of Initial Credit Extension 4.01
- Conditions to all Credit Extensions 4.02

ARTICLE V REPRESENTATIONS AND WARRANTIES

<u>5.01</u>	Existence, Qualification and Power; Compliance with Laws
<u>5.02</u>	Authorization; No Contravention
<u>5.03</u>	Governmental Authorization; Other Consents
<u>5.04</u>	Binding Effect
<u>5.05</u>	Financial Statements; No Material Adverse Effect,
<u>5.06</u>	Litigation
<u>5.07</u>	No Default
5.03 5.04 5.05 5.06 5.07 5.08	Ownership of Property; Liens
<u>5.09</u>	Environmental Compliance
<u>5.10</u>	<u>Insurance</u>
<u>5.11</u>	<u>Taxes</u>
<u>5.12</u>	ERISA Compliance.
5.11 5.12 5.13 5.14 5.15 5.16	Subsidiaries
<u>5.14</u>	Margin Regulations; Investment Company Act; Public Utility Holding Company Act.
<u>5.15</u>	<u>Disclosure</u>
<u>5.16</u>	Compliance with Laws
<u>5.17</u>	Intellectual Property; Licenses, Etc.
	i
<u>5.18</u>	Rights in Collateral; Priority of Liens
<u>ARTICLE</u>	VI AFFIRMATIVE COVENANTS

<u>6.01</u>	<u>Financial Statements</u>
<u>6.02</u>	Certificates; Other Information
<u>6.03</u>	<u>Notices</u>
<u>6.04</u>	Payment of Obligations
<u>6.05</u>	Preservation of Existence, Etc.
<u>6.06</u>	Maintenance of Properties
<u>6.07</u>	Maintenance of Insurance
6.08	Compliance with Laws
<u>6.09</u>	Books and Records
6.10	Inspection Rights
6.11	Use of Proceeds
6.12	Financial Covenants.
6.13	Additional Guarantors: Pledge of Capital Stock

6.13 Additional Guarantors; Pledge of Capital Stock.

6.14 Security Interests

ARTICLE VII NEGATIVE COVENANTS

7.01	<u>Liens</u>
<u>7.02</u>	<u>Investments</u>
<u>7.03</u>	<u>Indebtedness</u>
<u>7.04</u>	Fundamental Changes
<u>7.05</u>	<u>Dispositions</u>
<u>7.06</u>	Restricted Payments
<u>7.07</u>	Change in Nature of Business
<u>7.08</u>	Transactions with Affiliates
<u>7.09</u>	Burdensome Agreements
7.10	Use of Proceeds

ARTICLE VIII EVENTS OF DEFAULT AND REMEDIES

<u>8.01</u>	Events of Default
<u>8.02</u>	Remedies Upon Event of Default
<u>8.03</u>	Application of Funds
ARTICLE	IX MISCELLANEOUS

<u>9.01</u>	Amendments; Etc.
<u>9.02</u>	Notices and Other Communications; Facsimile Copie
9.03	No Waiver; Cumulative Remedies
<u>9.04</u>	Attorney Costs, Expenses and Taxes
9.05	Indemnification by Borrower
9.06	Payments Set Aside
9.07	Successors and Assigns
9.08	Confidentiality
0.00	C + CC

9.09Set-off9.10Interest Rate Limitation

9.11 Gountstrearts
9.13 Survival of Representations and Warranties
9.14 Severability
9.15 Governing Law; Submission to Jurisdiction.
9.16 Waiver of Right to Trial by Jury
9.17 Mandatory Arbitration.
9.18 USA Patriot Act Notice
9.19 Time of the Essence

ii

SCHEDULES

9.20

5.06	Litigation

- 5.09 Environmental Matters
- 5.13 Subsidiaries and Other Equity Investments
- 5.18 <u>Intellectual Property Matters</u>

Oral Agreements

- 7.01 Existing Liens
- 7.03 Existing Indebtedness

EXHIBITS

Form of

A Loan Notice
B Note
C Guaranty
D Pledge Agreement
E Compliance Certificate

iii

CREDIT AGREEMENT

This CREDIT AGREEMENT ("Agreement") is entered into as of June 16, 2004, between USANA HEALTH SCIENCES, INC., a Utah corporation ("Borrower") and BANK OF AMERICA, N.A., a national banking association ("Bank").

Borrower has requested that Bank provide a revolving credit facility, and Bank is willing to do so on the terms and conditions set forth herein.

In consideration of the mutual covenants and agreements herein contained, the parties hereto covenant and agree as follows:

ARTICLE I DEFINITIONS AND ACCOUNTING TERMS

1.01 Defined Terms. As used in this Agreement, the following terms shall have the meanings set forth below:

"Affiliate" means, with respect to any Person, another Person that directly, or indirectly through one or more intermediaries, Controls, or is Controlled by or is under common Control with, the Person specified. "Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ability to exercise voting power, by contract or otherwise. "Controlling" and "Controlled" have meanings correlative thereto.

"Agreement" means this Credit Agreement.

"Applicable Rate" means, from time to time, the following percentages per annum, based upon the Consolidated Leverage Ratio set forth in Section 6.12(c) (the "Financial Covenant") as set forth in the most recent Compliance Certificate received by Bank pursuant to Section 6.02(b):

Applicable Rate

				Eurodollar Rate +	
Pricing Level		Consolidated Leverage Ratio	Commitment fee	Letters of Credit	Base Rate +
	1	.75:1	0.25 %	1.25 %	0.00 %
	2	$> .75:1 \text{ but} \leq 1.0:1$	0.50 %	1.75 %	0.50 %

Any increase or decrease in the Applicable Rate resulting from a change in the Financial Covenant shall become effective as of the first Business Day of the month immediately following the date a Compliance Certificate is delivered pursuant to Section 6.02(b); provided, however, that if no Compliance Certificate is delivered when due in accordance with such Section, then Pricing Level 2 shall apply as of the first Business Day of the month following the date such Compliance Certificate was required to have been delivered. The Applicable Rate in effect from the Closing Date through the first Business Day of the month immediately following the date a Compliance Certificate is delivered for Borrower's fiscal quarter ended July 2, 2004 shall be determined based upon Pricing Level 1.

"Attorney Costs" means and includes all fees, expenses and disbursements of any law firm or other external counsel and, without duplication, the allocated cost of internal legal services and all expenses and disbursements of internal counsel.

"Attributable Indebtedness" means, on any date, (a) in respect of any capital lease of any Person, the capitalized amount thereof that would appear on a balance sheet of such Person prepared as of such date in accordance with GAAP, and (b) in respect of any Synthetic Lease Obligation, the capitalized

amount of the remaining lease payments under the relevant lease that would appear on a balance sheet of such Person prepared as of such date in accordance with GAAP if such lease were accounted for as a capital lease.

"<u>Audited Financial Statements</u>" means the audited consolidated balance sheet of Borrower and its Subsidiaries for the fiscal year ended January 3, 2004, and the related consolidated statements of income or operations, shareholders' equity and cash flows for such fiscal year of Borrower and its Subsidiaries, including the notes thereto.

"Availability Period" means the period from and including the Closing Date to the earliest of (a) the Maturity Date and (b) the date of termination of the commitment of Bank to make Loans and of the obligation of Bank to make L/C Credit Extensions pursuant to Section 8.02.

"Bank" has the meaning specified in the introductory paragraph hereto.

"Base Rate" means for any day a fluctuating rate per annum equal to the higher of (a) the Federal Funds Rate <u>plus</u> 1/2 of 1% and (b) the rate of interest in effect for such day as publicly announced from time to time by Bank as its "prime rate." The "prime rate" is a rate set by Bank based upon various factors including Bank's costs and desired return, general economic conditions and other factors, and is used as a reference point for pricing some loans, which may be priced at, above, or below such announced rate. Any change in such rate announced by Bank shall take effect at the opening of business on the day specified in the public announcement of such change.

"Base Rate Loan" means a Loan that bears interest based on the Base Rate.

"Borrower" has the meaning specified in the introductory paragraph hereto.

"Borrower Account" means account number 68504810 maintained by Borrower with Bank or such other ordinary checking account maintained by Borrower with Bank designated by Borrower in a written notice to Bank from time to time.

"Business Day" means any day other than a Saturday, Sunday or other day on which commercial banks are authorized to close under the Laws of, or are in fact closed in, the state of Washington and, if such day relates to any Eurodollar Rate Loan, means any such day on which dealings in Dollar deposits are conducted by and between banks in the London interbank eurodollar market.

"<u>Capital Expenditure</u>" means, with respect to any Person, any liability incurred or expenditure made (net of any casualty insurance proceeds or condemnation awards used to replace fixed assets following a casualty event or condemnation with respect thereto) by such Person in respect of the purchase or other acquisition of any fixed or capital asset.

"<u>Capital Stock</u>" means (a) in the case of a corporation, capital stock, (b) in the case of an association or business entity, any and all shares, interests, participations, rights or other equivalents (however designated) of capital stock, (c) in the case of a partnership, partnership interests (whether general or limited), (d) in the case of a limited liability company, membership interests and (e) any other interest or participation that confers on a Person the right to receive a share of the profits and losses of, or distributions of assets of, the issuing Person.

"Cash Collateralize" has the meaning specified in Section 2.03(d).

"Change of Control" means an event or series of events by which (a) Myron W. Wentz, his spouse, members of his immediate family, and/or any of the lineal descendants of any thereof and/or (b) any trust of similar entity all of the beneficiaries of which, or a corporation, partnership or limited liability company all of the stockholders and other equity holders, limited and general partners or members of which, are (i) solely the Persons in the foregoing clause (a) and/or (ii) any entity described in this clause (b) all the beneficiaries of which, or all the stockholders and other equity holders, limited and

"Change of Management" means an event or series of events (including without limitation, death, disability, resignation or termination) by which Myron W. Wentz, David A. Wentz or an Approved Successor with respect to each, ceases to direct or cause the direction of the management and policies of each Loan Party. "Approved Successor" means a natural person elected by the board of directors of the applicable Loan Party and reasonably acceptable to Bank not more than one hundred twenty (120) days after the death, disability, resignation or termination of Myron W. Wentz, David A. Wentz or an Approved Successor, as the case may be.

"Closing Date" means the first date all the conditions precedent in Section 4.01 are satisfied or waived by Bank.

"Code" means the Internal Revenue Code of 1986.

"Collateral" shall mean any and all assets and rights and interests in or to property of Borrower and each of the other Loan Parties, whether real or personal, tangible or intangible, in which a Lien is granted or purported to be granted pursuant to the Collateral Documents.

"Collateral Documents" means, collectively, the Pledge Agreement and all other agreements, instruments and documents now or hereafter executed and delivered in connection with this Agreement pursuant to which Liens are granted or purported to be granted to Bank in Collateral securing all or part of the Obligations each in form and substance satisfactory to Bank.

"Commitment" means the obligation of Bank to make Loans and L/C Credit Extensions hereunder in an aggregate principal amount at any one time not to exceed \$10,000,000, as such amount may be adjusted from time to time in accordance with this Agreement.

"Compliance Certificate" means a certificate substantially in the form of Exhibit E.

"Consolidated EBITDA" means, for any period, for Borrower and its Subsidiaries on a consolidated basis, an amount equal to the net income of Borrower and its Subsidiaries (excluding extraordinary gains but including extraordinary losses) for such period ("Net Income") plus (a) the following to the extent deducted in calculating such Net Income: (i) the sum of (A) all interest, premium payments, debt discount, fees (including commitment fees and the amortization of upfront fees), charges and related expenses of Borrower and its Subsidiaries in connection with borrowed money (excluding capitalized interest) or in connection with the deferred purchase price of assets for such period, in each case to the extent treated as interest in accordance with GAAP and (B) the portion of rent expense of Borrower and its Subsidiaries for such period under capital leases that is treated as interest in accordance with GAAP, (ii) the provision for federal, state, local and foreign income taxes payable by Borrower and its Subsidiaries for such period, (iii) the amount of depreciation, depletion and amortization expense deducted in determining such Net Income and (iv) other expenses of Borrower and its Subsidiaries reducing such Net Income which do not represent a cash item in such period or any future period and minus (b) all non-cash items increasing Net Income for such period.

"Consolidated Leverage Ratio" means, as of any date of determination, the ratio of (a) Consolidated Total Liabilities as of such date to (b) Consolidated Tangible Net Worth as of such date.

"Consolidated Tangible Net Worth" means, as of any date of determination, for Borrower and its Subsidiaries on a consolidated basis, the value of total assets (including leaseholds and leasehold improvements and reserves against assets but excluding goodwill, patents, trademarks, trade names, organization expense, unamortized debt discount and expense, capitalized or deferred research and development costs, deferred marketing expenses, and other like intangibles, and monies due from Affiliates, officers, directors, employees, shareholders, members or managers) of Borrower and its Subsidiaries determined in accordance with GAAP minus Consolidated Total Liabilities.

3

"Consolidated Total Liabilities" means, as of any date of determination, for Borrower and its Subsidiaries on a consolidated basis, the total liabilities of Borrower and its Subsidiaries determined in accordance with GAAP.

"Contractual Obligation" means, as to any Person, any provision of any security issued by such Person or of any agreement, instrument or other undertaking to which such Person is a party or by which it or any of its property is bound.

"Control" has the meaning specified in the definition of "Affiliate."

"Credit Extension" means each of the following: (a) a borrowing of a Loan and (b) an L/C Credit Extension.

"<u>Debtor Relief Laws</u>" means the Bankruptcy Code of the United States, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief Laws of the United States or other applicable jurisdictions from time to time in effect and affecting the rights of creditors generally.

"<u>Default</u>" means any event or condition that constitutes an Event of Default or that, with the giving of any notice, the passage of time, or both, would be an Event of Default.

"<u>Default Rate</u>" means an interest rate equal to (a) the Base Rate <u>plus</u> (b) the Applicable Rate, if any, applicable to Base Rate Loans <u>plus</u> (c) 4% per annum; <u>provided</u>, <u>however</u>, that with respect to a Eurodollar Rate Loan, the Default Rate shall be an interest rate equal to the interest rate (including any Applicable Rate) otherwise applicable to such Loan <u>plus</u> 4% per annum, in each case to the fullest extent permitted by applicable Laws.

"Disposition" or "Dispose" means the sale, transfer, license, lease or other disposition (including any sale and leaseback

transaction) of any property by any Person, including any sale, assignment, transfer or other disposal, with or without recourse, of any notes or accounts receivable or any rights and claims associated therewith.

"<u>Disregarded Foreign Subsidiary</u>" shall mean any Foreign Subsidiary the separate existence of which is disregarded for United States Federal tax purposes under Treas. Reg. Section 301.7701 3.

"Dollar" and "\$" mean lawful money of the United States.

"<u>Domestic Subsidiary</u>" means any Subsidiary that is organized and existing under the laws of the United States or any state or commonwealth thereof or under the laws of the District of Columbia.

"Environmental Laws" means any and all Federal, state, local, and foreign statutes, laws, regulations, ordinances, rules, judgments, orders, decrees, permits, concessions, grants, franchises, licenses, agreements or governmental restrictions relating to pollution and the protection of the environment or the release of any materials into the environment, including those related to hazardous substances or wastes, air emissions and discharges to waste or public systems.

"Environmental Liability" means any liability, contingent or otherwise (including any liability for damages, costs of environmental remediation, fines, penalties or indemnities), of Borrower, any other Loan Party or any of their respective Subsidiaries directly or indirectly resulting from or based upon (a) violation of any Environmental Law, (b) the generation, use, handling, transportation, storage, treatment or disposal of any Hazardous Materials, (c) exposure to any Hazardous Materials, (d) the release or threatened release of any Hazardous Materials into the environment or (e) any contract, agreement or other consensual arrangement pursuant to which liability is assumed or imposed with respect to any of the foregoing.

"ERISA" means the Employee Retirement Income Security Act of 1974.

4

"ERISA Affiliate" means any trade or business (whether or not incorporated) under common control with Borrower within the meaning of Section 414(b) or (c) of the Code (and Sections 414(m) and (o) of the Code for purposes of provisions relating to Section 412 of the Code).

"ERISA Event" means (a) a Reportable Event with respect to a Pension Plan; (b) a withdrawal by Borrower or any ERISA Affiliate from a Pension Plan subject to Section 4063 of ERISA during a plan year in which it was a substantial employer (as defined in Section 4001(a)(2) of ERISA) or a cessation of operations that is treated as such a withdrawal under Section 4062(e) of ERISA; (c) a complete or partial withdrawal by Borrower or any ERISA Affiliate from a Multiemployer Plan or notification that a Multiemployer Plan is in reorganization; (d) the filing of a notice of intent to terminate, the treatment of a Plan amendment as a termination under Sections 4041 or 4041A of ERISA, or the commencement of proceedings by the PBGC to terminate a Pension Plan or Multiemployer Plan; (e) an event or condition which constitutes grounds under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Pension Plan or Multiemployer Plan; or (f) the imposition of any liability under Title IV of ERISA, other than for PBGC premiums due but not delinquent under Section 4007 of ERISA, upon Borrower or any ERISA Affiliate.

"Eurodollar Base Rate" has the meaning specified in the definition of Eurodollar Rate.

"Eurodollar Rate" means for any Interest Period with respect to any Eurodollar Rate Loan, a rate per annum determined by Bank pursuant to the following formula:

Eurodollar Rate Eurodollar Base Rate

1.00 - Eurodollar Reserve Percentage

Where,

"Eurodollar Base Rate" means, for such Interest Period:

- (a) the rate per annum equal to the rate determined by Bank to be the offered rate that appears on the page of the Telerate screen (or any successor thereto) that displays an average British Bankers Association Interest Settlement Rate for deposits in Dollars (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period, determined as of approximately 11:00 a.m. (London time) two Business Days prior to the first day of such Interest Period, or
- (b) if the rate referenced in the preceding clause (a) does not appear on such page or service or such page or service shall not be available, the rate per annum equal to the rate determined by Bank to be the offered rate on such other page or other service that displays an average British Bankers Association Interest Settlement Rate for deposits in Dollars (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period, determined as of approximately 11:00 a.m. (London time) two Business Days prior to the first day of such Interest Period, or
- (c) if the rates referenced in the preceding clauses (a) and (b) are not available, the rate per annum determined by Bank as the rate of interest at which deposits in Dollars for delivery on the first day of such Interest Period in same day funds in the approximate amount of the Eurodollar Rate Loan being made, continued or converted and with a term equivalent to such Interest Period would be offered by Bank's London Branch to major banks in the London interbank eurodollar market at their request at approximately 4:00 p.m. (London time) two Business Days prior to the first day of such Interest Period.

"Eurodollar Reserve Percentage" means, for any day during any Interest Period, the reserve percentage (expressed as a decimal, carried out to five decimal places) in effect on such day applicable to Bank, under regulations issued from time to time by the Board of Governors of the Federal Reserve System of the United States for determining the maximum reserve requirement (including any emergency,

5

supplemental or other marginal reserve requirement) with respect to Eurocurrency funding (currently referred to as "Eurocurrency liabilities"). The Eurodollar Rate for each outstanding Eurodollar Rate Loan shall be adjusted automatically as of the effective date of any change in the Eurodollar Reserve Percentage.

"Eurodollar Rate Loan" means a Loan that bears interest based on the Eurodollar Rate.

"Event of Default" has the meaning specified in Section 8.01.

"Federal Funds Rate" means, for any day, the rate per annum equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Business Day next succeeding such day; provided that (a) if such day is not a Business Day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on such next succeeding Business Day, the Federal Funds Rate for such day shall be the average rate (rounded upward, if necessary, to a whole multiple of 1/100 of 1%) charged to Bank on such day on such transactions as determined by Bank.

"Financial Transaction Contract" means any agreement (including all schedules thereto, confirmations of transactions thereunder, and documents, definitions, and agreements incorporated therein by reference or relating thereto) between Borrower or any Subsidiary and Bank or an Affiliate of Bank, whether or not in writing, pursuant to which Bank or an Affiliate of Bank has agreed to (i) permit daylight overdrafts to occur on accounts maintained by Borrower or any Subsidiary with Bank or an Affiliate of Bank, (ii) provide remote disbursement services for Borrower or any Subsidiary, (iii) process automated clearing house (ACH) transactions for the account of Borrower or any Subsidiary of Borrower or (iv) extend credit to Borrower or any Subsidiary, in the form of credit card accounts and merchant card accounts, and, unless the context otherwise clearly requires, any master agreement relating to or governing any or all of the foregoing.

"Foreign Subsidiary" shall mean any Subsidiary that is not a Domestic Subsidiary.

"GAAP" means generally accepted accounting principles in the United States set forth in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or such other principles as may be approved by a significant segment of the accounting profession in the United States, that are applicable to the circumstances as of the date of determination, consistently applied.

"Governmental Authority" means any nation or government, any state or other political subdivision thereof, any agency, authority, instrumentality, regulatory body, court, administrative tribunal, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government.

"Guarantee" means, as to any Person, any (a) any obligation, contingent or otherwise, of such Person guaranteeing or having the economic effect of guaranteeing any Indebtedness or other obligation payable or performable by another Person (the "primary obligor") in any manner, whether directly or indirectly, and including any obligation of such Person, direct or indirect, (i) to purchase or pay (or advance or supply funds for the purchase or payment of) such Indebtedness or other obligation, (ii) to purchase or lease property, securities or services for the purpose of assuring the obligee in respect of such Indebtedness or other obligation of the payment or performance of such Indebtedness or other obligation, (iii) to maintain working capital, equity capital or any other financial statement condition or liquidity or level of income or cash flow of the primary obligor so as to enable the primary obligor to pay such Indebtedness or other obligation, or (iv) entered into for the purpose of assuring in any other manner the obligee in respect of such Indebtedness or other obligation of the payment or performance thereof or to protect such obligee against loss in respect thereof (in whole or in part), or (b) any Lien on any assets

6

of such Person securing any Indebtedness or other obligation of any other Person, whether or not such Indebtedness or other obligation is assumed by such Person. The amount of any Guarantee shall be deemed to be an amount equal to the stated or determinable amount of the related primary obligation, or portion thereof, in respect of which such Guarantee is made or, if not stated or determinable, the maximum reasonably anticipated liability in respect thereof as determined by the guaranteeing Person in good faith. The term "Guarantee" as a verb has a corresponding meaning.

"Guarantors" means, collectively, USANA Acquisition Corporation, a Utah corporation, Wasatch Product Development, Inc., a Utah corporation, and each other Person becomes a party to the Guaranty pursuant to Section 6.13(a).

"Guaranty" means the Continuing Guaranty made by the Guarantors to and in favor of Bank, substantially in the form of Exhibit \underline{C} , as from time to time amended, restated or otherwise modified.

"<u>Hazardous Materials</u>" means all explosive or radioactive substances or wastes and all hazardous or toxic substances, wastes or other pollutants, including petroleum or petroleum distillates, asbestos or asbestos-containing materials, polychlorinated biphenyls, radon gas, infectious or medical wastes and all other substances or wastes of any nature regulated pursuant to any Environmental Law.

"Honor Date" has the meaning specified in Section 2.03(c)(i).

"Indebtedness" means, as to any Person at a particular time, without duplication, all of the following, whether or not included as indebtedness or liabilities in accordance with GAAP:

- (a) all obligations of such Person for borrowed money and all obligations of such Person evidenced by bonds, debentures, notes, loan agreements or other similar instruments;
- (b) all direct or contingent obligations of such Person arising under letters of credit (including standby and commercial), bankers' acceptances, bank guaranties, surety bonds and similar instruments;
 - (c) net obligations of such Person under any Swap Contract;
- (d) all obligations of such Person to pay the deferred purchase price of property or services (other than trade accounts payable in the ordinary course of business);
- (e) indebtedness (excluding prepaid interest thereon) secured by a Lien on property owned or being purchased by such Person (including indebtedness arising under conditional sales or other title retention agreements), whether or not such indebtedness shall have been assumed by such Person or is limited in recourse;
 - (f) capital leases and Synthetic Lease Obligations; and
 - (g) all Guarantees of such Person in respect of any of the foregoing.

For all purposes hereof, the Indebtedness of any Person shall include the Indebtedness of any partnership or joint venture (other than a joint venture that is itself a corporation or limited liability company) in which such Person is a general partner or a joint venturer, unless such Indebtedness is expressly made non-recourse to such Person. The amount of any net obligation under any Swap Contract on any date shall be deemed to be the Swap Termination Value thereof as of such date. The amount of any capital lease or Synthetic Lease Obligation as of any date shall be deemed to be the amount of Attributable Indebtedness in respect thereof as of such date.

"Indemnified Liabilities" has the meaning specified in Section 9.05.

"Indemnitees" has the meaning specified in Section 9.05.

7

"Information" has the meaning specified in Section 9.08.

"Interest Payment Date" means, (a) as to any Loan other than a Base Rate Loan, the last day of each Interest Period applicable to such Loan and the Maturity Date; and (b) as to any Base Rate Loan, the first Business Day after the end of each March, June, September and December and the Maturity Date.

"Interest Period" means as to each Eurodollar Rate Loan, the period commencing on the date such Eurodollar Rate Loan is disbursed or converted to or continued as a Eurodollar Rate Loan and ending on the date one, two or three months thereafter, as selected by Borrower in its Loan Notice; provided that:

- (a) any Interest Period that would otherwise end on a day that is not a Business Day shall be extended to the next succeeding Business Day unless, such Business Day falls in another calendar month, in which case such Interest Period shall end on the next preceding Business Day;
- (b) any Interest Period that begins on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last Business Day of the calendar month at the end of such Interest Period; and
 - (c) no Interest Period shall extend beyond the Maturity Date.

"Investment" means, as to any Person, any direct or indirect acquisition or investment by such Person, whether by means of (a) the purchase or other acquisition of capital stock or other securities of another Person, (b) a loan, advance or capital contribution to, Guarantee or assumption of debt of, or purchase or other acquisition of any other debt or equity participation or interest in, another Person, including any partnership or joint venture interest in such other Person, or (c) the purchase or other acquisition (in one transaction or a series of transactions) of assets of another Person that constitute a business unit. For purposes of covenant compliance, the amount of any Investment shall be the amount actually invested, without adjustment for subsequent increases or decreases in the value of such Investment.

"IRS" means the United States Internal Revenue Service.

"<u>Laws</u>" means, collectively, all international, foreign, Federal, state and local statutes, treaties, rules, guidelines, regulations, ordinances, codes and administrative or judicial precedents or authorities, including the interpretation or administration thereof by any Governmental Authority charged with the enforcement, interpretation or administration thereof, and all applicable administrative orders, directed duties, requests, licenses, authorizations and permits of, and agreements with, any Governmental Authority, in each case whether or not having the force of law.

"<u>L/C Credit Extension</u>" means, with respect to any Letter of Credit, the issuance thereof or extension of the expiry date thereof, or the renewal or increase of the amount thereof.

"<u>L/C Obligations</u>" means, as at any date of determination, the aggregate undrawn amount of all outstanding Letters of Credit <u>plus</u> the aggregate of all unreimbursed drawings under all Letters of Credit.

"<u>Lending Office</u>" means the office of Bank located at the address set forth in <u>Section 9.02</u>, or such other office or offices as Bank may from time to time notify Borrower.

"<u>Letter of Credit</u>" means any letter of credit issued hereunder. A Letter of Credit may be a commercial letter of credit or a standby letter of credit.

8

"<u>Letter of Credit Application</u>" means an application and agreement for the issuance or amendment of a Letter of Credit in the form from time to time in use by Bank.

"<u>Letter of Credit Expiration Date</u>" means the day that is seven days prior to the Maturity Date then in effect (or, if such day is not a Business Day, the next preceding Business Day).

"<u>Letter of Credit Sublimit</u>" means an amount equal to \$500,000. The Letter of Credit Sublimit is part of, and not in addition to, the Commitment.

"<u>Lien</u>" means any mortgage, pledge, hypothecation, assignment, deposit arrangement, encumbrance, lien (statutory or other), charge, or preference, priority or other security interest or preferential arrangement of any kind or nature whatsoever (including any conditional sale or other title retention agreement, and any financing lease having substantially the same economic effect as any of the foregoing.

"Loan" has the meaning specified in Section 2.01.

"Loan Documents" means this Agreement, the Note, each Collateral Document and the Guaranty.

"Loan Notice" means a notice of (a) a borrowing of a Loan, (b) a conversion of a Loan from one Type to the other, or (c) a continuation of a Eurodollar Rate Loan as the same Type, pursuant to Section 2.02(a), which, if in writing, shall be substantially in the form of Exhibit A.

"<u>Loan Parties</u>" means, collectively, Borrower and each Person (other than Bank) executing a Loan Document, including each Guarantor and each Person executing a Collateral Document.

"Material Adverse Effect" means (a) a material adverse change in, or a material adverse effect upon, the operations, business, properties, liabilities (actual or contingent), condition (financial or otherwise) or prospects of Borrower or Borrower and its Subsidiaries taken as a whole or the Collateral; (b) a material impairment of the ability of any Loan Party to perform its obligations under any Loan Document to which it is a party; or (c) a material adverse effect upon the legality, validity, binding effect or enforceability against any Loan Party of any Loan Document to which it is a party.

"Material Subsidiary" means any Subsidiary that during the then current fiscal year of Borrower (on a pro forma basis) or either of the two most recently ended fiscal years of Borrower, accounts or accounted for 10% or more of the consolidated revenue of Borrower.

"Maturity Date" means May 30, 2006.

"Minimum Tangible Net Worth" means, as of any date of determination, the sum of (a) Thirty-two Million Dollars (\$32,000,000) plus (b) the sum of the following for Borrower and its Subsidiaries on a consolidated basis: (i) the sum of (A) 20% of net income after income taxes (without subtracting losses) earned in each quarterly accounting period commencing January 3, 2004 through January 7, 2005 and (B) 30% of net income after taxes (without subtracting losses) earned in each quarterly accounting period commencing after January 7, 2005; (ii) the net proceeds from any equity securities issued after January 3, 2004; and (iii) any increase in stockholders' equity resulting from the conversion of debt securities to equity securities after January 3, 2004.

"Multiemployer Plan" means any employee benefit plan of the type described in Section 4001(a)(3) of ERISA, to which Borrower or any ERISA Affiliate makes or is obligated to make contributions, or during the preceding five plan years, has made or been obligated to make contributions.

"Obligations" means all advances to, and debts, liabilities, obligations, covenants and duties of, any Loan Party arising under any Loan Document or otherwise with respect to any Loan or Letter of Credit, whether direct or indirect (including those acquired by assumption), absolute or contingent, due or to become due, now existing or hereafter arising and including interest and fees that accrue after the commencement by or against any Loan Party or any Affiliate thereof of any proceeding under any Debtor Relief Laws naming such Person as the debtor in such proceeding, regardless of whether such interest and fees are allowed claims in such proceeding.

"Organization Documents" means, (a) with respect to any corporation, the certificate or articles of incorporation and the bylaws (or equivalent or comparable constitutive documents with respect to any non-U.S. jurisdiction); (b) with respect to any limited liability company, the certificate or articles of formation or organization and operating agreement; and (c) with respect to any partnership, joint venture, trust or other form of business entity, the partnership, joint venture or other applicable agreement of formation or organization and any agreement, instrument, filing or notice with respect thereto filed in connection with its formation or organization with the applicable Governmental Authority in the jurisdiction of its formation or organization and, if applicable, any certificate or articles of formation or organization of such entity.

"<u>Outstanding Amount</u>" means (i) with respect to Loans on any date, the aggregate outstanding principal amount thereof after giving effect to any borrowings and prepayments or repayments of Loans occurring on such date; and (ii) with respect to any L/C Obligations on any date, the amount of such L/C Obligations on such date after giving effect to any L/C Credit Extension occurring on such date and any other changes in the aggregate amount of the L/C Obligations as of such date, including as a result of any reimbursements of outstanding unpaid drawings under any Letters of Credit or any reductions in the maximum amount available for drawing under Letters of Credit taking effect on such date.

"PBGC" means the Pension Benefit Guaranty Corporation.

"Pension Plan" means any "employee pension benefit plan" (as such term is defined in Section 3(2) of ERISA), other than a Multiemployer Plan, that is subject to Title IV of ERISA and is sponsored or maintained by Borrower or any ERISA Affiliate or to which Borrower or any ERISA Affiliate contributes or has an obligation to contribute, or in the case of a multiple employer or other plan described in Section 4064(a) of ERISA, has made contributions at any time during the immediately preceding five plan years.

"<u>Permitted Acquisition</u>" means any acquisition, whether by purchase, merger or otherwise, of all or substantially all of the assets of, or more than 50% of the voting Capital Stock of, or a business line or a division of, any Person; <u>provided</u> that:

- (a) all Persons, assets, business lines or divisions acquired shall be in the type of business permitted to be engaged in by Borrower and its Subsidiaries pursuant to Section 7.07 or such other lines of business as may be consented to by Bank;
 - (b) no Default or Event of Default shall then exist or would exist after giving effect to such acquisition;
- (c) as of the closing of any acquisition, such acquisition shall have been approved by the board of directors or equivalent governing body of the Person to be acquired or from which such assets, business line or division is to be acquired;
- (d) not less than 15 Business Days prior to the consummation of any acquisition for consideration (including assumed liabilities, earnout payments and any other deferred payment) in excess of \$5,000,000, Borrower shall have delivered to Bank a written description of the Person, assets, business line or division to be acquired and its operations;

10

- (e) Borrower shall demonstrate to the reasonable satisfaction of Bank that, after giving effect to such acquisition, Borrower will be in pro forma compliance with all of the terms and provisions of the financial covenants set forth in Section 6.12; and
- (f) if such acquisition is structured as a merger, Borrower (or if such merger is with any Subsidiary, then such Subsidiary) shall be the surviving Person after giving effect to such merger.

"Person" means any natural person, corporation, limited liability company, trust, joint venture, association, company, partnership, Governmental Authority or other entity.

"Plan" means any "employee benefit plan" (as such term is defined in Section 3(3) of ERISA) established by Borrower or, with respect to any such plan that is subject to Section 412 of the Code or Title IV of ERISA, any ERISA Affiliate.

"<u>Pledge Agreement</u>" means the Pledge Agreement made by Borrower to and in favor of Bank, substantially in the form of <u>Exhibit D</u>, as from time to time amended, restated or otherwise modified.

"Reportable Event" means any of the events set forth in Section 4043(c) of ERISA, other than events for which the 30 day notice period has been waived.

"Request for Credit Extension" means (a) with respect to a borrowing, conversion or continuation of a Loan, a Loan Notice, and (b) with respect to an L/C Credit Extension, a Letter of Credit Application.

"Responsible Officer" means the chief executive officer, president, chief financial officer, or treasurer of a Loan Party. Any document delivered hereunder that is signed by a Responsible Officer of a Loan Party shall be conclusively presumed to have been authorized by all necessary corporate, partnership and/or other action on the part of such Loan Party and such Responsible Officer shall be conclusively presumed to have acted on behalf of such Loan Party.

"Restricted Payment" means any dividend or other distribution (whether in cash, securities or other property) with respect to any capital stock or other equity interest of Borrower or any Subsidiary, or any payment (whether in cash, securities or other property), including any sinking fund or similar deposit on account of the purchase, redemption, retirement, acquisition, cancellation or termination of any such capital stock or other equity interest

"Subordinated Liabilities" means liabilities subordinated to the Obligations in a manner acceptable to Bank its sole discretion.

"Subsidiary" of a Person means a corporation, partnership, joint venture, limited liability company or other business entity of which a majority of the Capital Stock having ordinary voting power for the election of directors or other governing body (other than Capital Stock having such power only by reason of the happening of a contingency) are at the time beneficially owned, or the management of which is otherwise controlled, directly, or indirectly through one or more intermediaries, or both, by such Person. Unless otherwise specified, all references herein to a "Subsidiary" or to "Subsidiaries" shall refer to a Subsidiary or Subsidiaries of Borrower.

"Swap Contract" means (a) any and all rate swap transactions, basis swaps, credit derivative transactions, forward rate transactions, commodity swaps, commodity options, forward commodity contracts, equity or equity index swaps or options, bond or bond price or bond index swaps or options or forward bond or forward bond price or forward bond index transactions, interest rate options, forward foreign exchange transactions, cap transactions, floor transactions, collar transactions, currency swap transactions, cross-currency rate swap transactions, currency options, spot contracts, or any other similar

11

transactions or any combination of any of the foregoing (including any options to enter into any of the foregoing), whether or not any such transaction is governed by or subject to any master agreement, and (b) any and all transactions of any kind, and the related confirmations, which are subject to the terms and conditions of, or governed by, any form of master agreement published by the International Swaps and Derivatives Association, Inc., any International Foreign Exchange Master Agreement, or any other master agreement (any such master agreement, together with any related schedules, a "Master Agreement"), including any such obligations or liabilities under any Master Agreement.

"Swap Termination Value" means, in respect of any one or more Swap Contracts, after taking into account the effect of any legally enforceable netting agreement relating to such Swap Contracts, (a) for any date on or after the date such Swap Contracts have been closed out and termination value(s) determined in accordance therewith, such termination value(s), and (b) for any date prior to the date referenced in clause (a), the amount(s) determined as the mark-to-market value(s) for such Swap Contracts, as determined based upon one or more mid-market or other readily available quotations provided by any recognized dealer in such Swap Contracts (which may include Bank or any Affiliate Bank).

"Synthetic Lease Obligation" means the monetary obligation of a Person under (a) a so-called synthetic, off-balance sheet or tax retention lease, or (b) an agreement for the use or possession of property creating obligations that do not appear on the balance sheet of such Person but which, upon the insolvency or bankruptcy of such Person, would be characterized as the indebtedness of such Person (without regard to accounting treatment).

"Threshold Amount" means \$500,000.

"Total Outstandings" means the aggregate Outstanding Amount of all Loans and all L/C Obligations.

"Type" means, with respect to a Loan, its character as a Base Rate Loan or a Eurodollar Rate Loan.

"<u>Unfunded Pension Liability</u>" means the excess of a Pension Plan's benefit liabilities under Section 4001(a)(16) of ERISA, over the current value of that Pension Plan's assets, determined in accordance with the assumptions used for funding the Pension Plan pursuant to Section 412 of the Code for the applicable plan year.

"United States" and "U.S." mean the United States of America.

- **1.02 Other Interpretive Provisions.** With reference to this Agreement and each other Loan Document, unless otherwise specified herein or in such other Loan Document:
 - (a) The meanings of defined terms are equally applicable to the singular and plural forms of the defined terms.
 - (b) (i) The words "herein," "hereof," "hereof" and "hereunder" and words of similar import when used in any Loan

Document shall refer to such Loan Document as a whole and not to any particular provision thereof; (ii) Article, Section, Exhibit and Schedule references are to the Loan Document in which such reference appears; (iii) the term "including" is by way of example and not limitation; and (iv) the term "documents" includes any and all instruments, documents, agreements, certificates, notices, reports, financial statements and other writings, however evidenced, whether in physical or electronic form.

(c) In the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including;" the words "to" and "until" each mean "to but excluding;" and the word "through" means "to and including."

12

- (d) Section headings herein and in the other Loan Documents are included for convenience of reference only and shall not affect the interpretation of this Agreement or any other Loan Document.
- 1.03 Accounting Terms. (a) All accounting terms not specifically or completely defined herein shall be construed in conformity with, and all financial data (including financial ratios and other financial calculations) required to be submitted pursuant to this Agreement shall be prepared in conformity with, GAAP applied on a consistent basis, as in effect from time to time, applied in a manner consistent with that used in preparing the Audited Financial Statements, except as otherwise specifically prescribed herein.
- (b) If at any time any change in GAAP would affect the computation of any financial ratio or requirement set forth in any Loan Document, and either Borrower or Bank shall so request, Bank and Borrower shall negotiate in good faith to amend such ratio or requirement to preserve the original intent thereof in light of such change in GAAP (subject to the approval of Bank); provided that, until so amended, (i) such ratio or requirement shall continue to be computed in accordance with GAAP prior to such change therein and (ii) Borrower shall provide to Bank financial statements and other documents required under this Agreement or as reasonably requested hereunder setting forth a reconciliation between calculations of such ratio or requirement made before and after giving effect to such change in GAAP.
- **1.04 Rounding.** Any financial ratios required to be maintained by Borrower pursuant to this Agreement shall be calculated by dividing the appropriate component by the other component, carrying the result to one place more than the number of places by which such ratio is expressed herein and rounding the result up or down to the nearest number (with a rounding-up if there is no nearest number).
- 1.05 References to Agreements and Laws. Unless otherwise expressly provided herein, (a) references to Organization Documents, agreements (including the Loan Documents) and other contractual instruments shall be deemed to include all subsequent amendments, restatements, extensions, supplements and other modifications thereto, but only to the extent that such amendments, restatements, extensions, supplements and other modifications are not prohibited by any Loan Document; and (b) references to any Law shall include all statutory and regulatory provisions consolidating, amending, replacing, supplementing or interpreting such Law.
- 1.06 Letter of Credit Amounts. Unless otherwise specified, all references herein to the amount of a Letter of Credit at any time shall be deemed to mean the maximum face amount of such Letter of Credit after giving effect to all increases thereof contemplated by such Letter of Credit or the Letter of Credit Application therefor, whether or not such maximum face amount is in effect at such time.

ARTICLE II THE COMMITMENT AND CREDIT EXTENSIONS.

2.01 Loans. Subject to the terms and conditions set forth herein, Bank agrees to make loans (each such loan, a "<u>Loan</u>") to Borrower from time to time, on any Business Day during the Availability Period, in an aggregate amount not to exceed at any time outstanding the amount of the Commitment; <u>provided</u>, <u>however</u>, that after giving effect to any borrowing, the Total Outstandings shall not exceed the Commitment. Within the limits of the Commitment, and subject to the other terms and conditions hereof, Borrower may borrow under this <u>Section 2.01</u>, prepay under <u>Section 2.04</u>, and reborrow under this <u>Section 2.01</u>. A Loan may be a Base Rate Loan or a Eurodollar Rate Loan, as further provided herein.

2.02 Borrowings, Conversions and Continuations of Loans.

(a) Each borrowing, each conversion of a Loan from one Type to the other, and each continuation of a Eurodollar Rate Loan shall be made upon Borrower's irrevocable notice to Bank, which may be given by telephone. Each such notice must be received by Bank not later than 1:00 p.m., Seattle time, (i) three Business Days prior to the requested date of any borrowing of, conversion to or continuation of a Eurodollar Rate Loan or of any conversion of a Eurodollar Rate Loan, and (ii) on the requested date of any borrowing of a Base Rate Loan. Notwithstanding anything to

13

the contrary contained herein, but subject to the provisions of Section 9.02(d), any such telephonic notice may be given by an individual who has been authorized in writing to do so by a Responsible Officer of Borrower. Each such telephonic notice must be confirmed promptly by delivery to Bank of a written Loan Notice, appropriately completed and signed by a Responsible Officer of Borrower. Each borrowing of, conversion to or continuation of a Eurodollar Rate Loan shall be in a principal amount of \$100,000 or a whole multiple of \$50,000 in excess thereof. Except as provided in Section 2.03(c), each borrowing of or conversion to a Base Rate Loan shall be in a principal amount of \$50,000 or a whole multiple of \$10,000 in excess thereof. Each Loan Notice (whether telephonic or written) shall specify (i) whether Borrower is requesting a borrowing, a conversion of a Loan from one Type to the other, or a continuation of a Eurodollar Rate Loan, (ii) the requested date of the borrowing, conversion or continuation, as the case may be (which shall be a Business

Day), (iii) the principal amount of the Loan to be borrowed, converted or continued, (iv) the Type of Loan to be borrowed or to which an existing Loan is to be converted, and (v) if applicable, the duration of the Interest Period with respect thereto. If Borrower fails to specify a Type of Loan in a Loan Notice or if Borrower fails to give a timely notice requesting a conversion or continuation, then the applicable Loan shall be made as, or converted to, a Base Rate Loan. Any such automatic conversion to a Base Rate Loan shall be effective as of the last day of the Interest Period then in effect with respect to the applicable Eurodollar Rate Loan. If Borrower requests a borrowing of, conversion to, or continuation of a Eurodollar Rate Loan in any such Loan Notice, but fails to specify an Interest Period, it will be deemed to have specified an Interest Period of one month.

- (b) Upon satisfaction of the applicable conditions set forth in Section 4.02 (and, if a borrowing is the initial Credit Extension, Section 4.01), Bank shall make the proceeds of each Loan available to Borrower either by (i) crediting the Borrower Account with the amount of such proceeds or (ii) wire transfer of such proceeds, in each case in accordance with instructions provided to (and reasonably acceptable to) Bank by Borrower; provided, however, that if on the date of the Loan Notice with respect to such borrowing is given, there are drawings under Letters of Credit that have not been reimbursed by Borrower, then the proceeds of such borrowing shall be applied, first, to the payment in full of any such unreimbursed drawings, and second, to Borrower as provided above.
- (c) Except as otherwise provided herein, a Eurodollar Rate Loan may be continued or converted only on the last day of an Interest Period for such Eurodollar Rate Loan. During the existence of a Default, no Loan may be requested as, converted to or continued as Eurodollar Rate Loans without the consent of Bank, and Bank may demand that any or all of the then outstanding Eurodollar Rate Loans be converted immediately to Base Rate Loans and Borrower agrees to pay all amounts due under Section 3.05 in accordance with the terms thereof due to any such conversion.
- (d) Bank shall promptly notify Borrower of the interest rate applicable to any Interest Period for a Eurodollar Rate Loan upon determination of such interest rate. The determination of the Eurodollar Rate by Bank shall be conclusive in the absence of manifest error. At any time that a Base Rate Loan is outstanding, Bank shall notify Borrower of any change in Bank's prime rate used in determining the Base Rate promptly following the public announcement of such change.
- (e) After giving effect to all borrowings, all conversions of Loans from one Type to the other, and all continuations of Loans as the same Type, there shall not be more than ten Interest Periods in effect.

2.03 Letters of Credit.

(a) <u>The Letter of Credit Commitment.</u>

(i) Subject to the terms and conditions set forth herein, Bank agrees (A) from time to time on any Business Day during the period from the Closing Date until the Letter of Credit Expiration Date, to issue Letters of Credit for the account of Borrower or certain Subsidiaries, and to amend or renew Letters of Credit previously issued by it, in accordance with subsection (b) below, and (B) to honor drafts under the Letters of Credit; <u>provided</u> that Bank shall not be obligated to make any L/C Credit Extension with respect to any Letter of Credit, if as of the date

14

of such L/C Credit Extension, (y) the Total Outstandings would exceed the Commitment, or (z) the Outstanding Amount of the L/C Obligations would exceed the Letter of Credit Sublimit. Within the foregoing limits, and subject to the terms and conditions hereof, Borrower's ability to obtain Letters of Credit shall be fully revolving, and accordingly Borrower may, during the foregoing period, obtain Letters of Credit to replace Letters of Credit that have expired or that have been drawn upon and reimbursed.

- (ii) Bank shall be under no obligation to issue any Letter of Credit if:
- (A) any order, judgment or decree of any Governmental Authority or arbitrator shall by its terms purport to enjoin or restrain Bank from issuing such Letter of Credit, or any Law applicable to Bank or any request or directive (whether or not having the force of law) from any Governmental Authority with jurisdiction over Bank shall prohibit, or request that Bank refrain from, the issuance of letters of credit generally or such Letter of Credit in particular or shall impose upon Bank with respect to such Letter of Credit any restriction, reserve or capital requirement (for which Bank is not otherwise compensated hereunder) not in effect on the Closing Date, or shall impose upon Bank any unreimbursed loss, cost or expense which was not applicable on the Closing Date and which Bank in good faith deems material to it;
- (B) the expiry date of such requested Letter of Credit would occur more than twelve months after the date of issuance or last renewal;
- (C) the expiry date of such requested Letter of Credit would occur after the Letter of Credit Expiration Date; or
 - (D) the issuance of such Letter of Credit would violate one or more policies of Bank.
- (iii) Bank shall be under no obligation to amend any Letter of Credit if (A) Bank would have no obligation at such time to issue such Letter of Credit in its amended form under the terms hereof, or (B) the beneficiary of such Letter of Credit does not accept the proposed amendment to such Letter of Credit.
- (b) <u>Procedures for Issuance and Amendment of Letters of Credit.</u>

- (i) Each Letter of Credit shall be issued or amended, as the case may be, upon the request of Borrower delivered to Bank in the form of a Letter of Credit Application, appropriately completed and signed by a Responsible Officer of Borrower. Such Letter of Credit Application must be received by Bank not later than 1:00 p.m., Seattle time, at least two Business Days (or such later date and time as Bank may agree in a particular instance in its sole discretion) prior to the proposed issuance date or date of amendment, as the case may be.
- (ii) Upon Bank's determination that the requested issuance or amendment is permitted in accordance with the terms hereof, then, subject to the terms and conditions hereof, Bank shall, on the requested date, issue a Letter of Credit for the account of Borrower or enter into the applicable amendment, as the case may be, in each case in accordance with Bank's usual and customary business practices.
- (iii) Promptly after its delivery of any Letter of Credit or any amendment to a Letter of Credit to an advising bank with respect thereto or to the beneficiary thereof, Bank will also deliver to Borrower a true and complete copy of such Letter of Credit or amendment.

15

(c) <u>Drawings and Reimbursements</u>.

- (i) Upon receipt from the beneficiary of any Letter of Credit of any notice of a drawing under such Letter of Credit, Bank shall notify Borrower thereof. Not later than 1:00 p.m., Seattle time, on the date of any payment by Bank under a Letter of Credit (each such date, an "Honor Date"), Borrower shall reimburse Bank in an amount equal to the amount of such drawing. If Borrower fails to so reimburse Bank, Borrower shall be deemed to have requested a borrowing of a Base Rate Loan to be disbursed on the Honor Date in an amount equal to the amount of such unreimbursed drawing, without regard to the minimum and multiples specified in Section 2.02 for the principal amount of Base Rate Loans, but subject to the amount of the unutilized portion of the Commitment and the conditions set forth in Section 4.02 (other than the delivery of a Loan Notice).
- (ii) If Borrower fails to reimburse Bank for any drawing under any Letter of Credit (whether by means of a borrowing or otherwise), such unreimbursed amount shall be due and payable on demand (together with interest) and shall bear interest at the Default Rate.
- (d) <u>Cash Collateral</u>. Upon the request of Bank, (i) if Bank has honored any full or partial drawing request under any Letter of Credit and such drawing has not been reimbursed on the applicable Honor Date, or (ii) if, as of the Letter of Credit Expiration Date, any Letter of Credit may for any reason remain outstanding and partially or wholly undrawn, Borrower shall immediately Cash Collateralize the then Outstanding Amount of all L/C Obligations (in an amount equal to such Outstanding Amount determined as of the applicable Honor Date or the Letter of Credit Expiration Date, as the case may be). For purposes hereof, "Cash Collateralize" means to pledge and deposit with or deliver to Bank, as collateral for the L/C Obligations, cash or deposit account balances pursuant to documentation in form and substance satisfactory to Bank. Derivatives of such term have corresponding meanings. Borrower hereby grants to Bank a security interest in all such cash, deposit accounts and all balances therein and all proceeds of the foregoing. Cash collateral shall be maintained in blocked deposit accounts at Bank.
- (e) Letter of Credit Fees. Borrower shall pay to Bank (i) a Letter of Credit fee for each commercial Letter of Credit equal to one-half of 1% per annum times the daily maximum amount available to be drawn under such Letter of Credit (whether or not such maximum amount is then in effect under such Letter of Credit), and (ii) a Letter of Credit fee for each standby Letter of Credit equal to the Applicable Rate times the daily maximum amount available to be drawn under such Letter of Credit (whether or not such maximum amount is then in effect under such Letter of Credit). Such letter of credit fees shall be computed on a quarterly basis in arrears. Such letter of credit fees shall be due and payable on the first Business Day after the end of each March, June, September and December, commencing with the first such date to occur after the issuance of such Letter of Credit, on the Letter of Credit Expiration Date and thereafter on demand. If there is any change in the Applicable Rate during any quarter, the daily maximum amount of each standby Letter of Credit shall be computed and multiplied by the Applicable Rate separately for each period during such quarter that such Applicable Rate was in effect.
- (f) <u>Documentary and Processing Charges Payable to Bank.</u> Borrower shall pay to Bank the customary issuance, presentation, amendment and other processing fees, and other standard costs and charges, of Bank relating to letters of credit as from time to time in effect. Such customary fees and standard costs and charges are due and payable on demand and are nonrefundable.
- (g) <u>Conflict with Letter of Credit Application</u>. In the event of any conflict between the terms hereof and the terms of any Letter of Credit Application, the terms hereof shall control.

2.04 Prepayments.

(a) Borrower may, upon notice to Bank, at any time or from time to time voluntarily prepay any Loan in whole or in part without premium or penalty; provided that (i) such notice must be received by Bank not later than 1:00 p.m., Seattle time, (A) three Business Days prior to any date of prepayment of a Eurodollar Rate Loan, and (B) on the date of prepayment of a Base Rate Loan; (ii) any prepayment of a Eurodollar Rate Loan shall be in a principal amount of \$100,000 or a whole multiple of \$50,000 in excess

thereof; and (iii) any prepayment of a Base Rate Loan shall be in a principal amount of \$50,000 or a whole multiple of \$10,000 in excess thereof or, in each case, if less, the entire principal amount thereof then outstanding. Each such notice shall specify (i) the date and amount of such prepayment and (ii) the Type(s) of Loans to be prepaid. If such notice is given by Borrower, Borrower shall make such prepayment and the payment amount specified in such notice shall be due and payable on the date specified therein. Any prepayment of a Eurodollar Rate Loan shall be accompanied by all accrued interest thereon, together with any additional amounts required pursuant to Section 3.05.

- (b) If for any reason the Total Outstandings at any time exceed the Commitment then in effect, Borrower shall immediately prepay Loans and/or Cash Collateralize the L/C Obligations in an aggregate amount equal to such excess; provided, however, that Borrower shall not be required to Cash Collateralize the L/C Obligations pursuant to this Section 2.04(b) unless after the prepayment in full of the Loans the Total Outstandings exceed the Commitment then in effect.
- **2.05 Repayment of Loans.** Borrower shall repay to Bank on the Maturity Date the aggregate principal amount of Loans outstanding on such date.

2.06 Interest.

- (a) Subject to the provisions of subsection (b) below, (i) each Eurodollar Rate Loan shall bear interest on the outstanding principal amount thereof for each Interest Period at a rate per annum equal to the Eurodollar Rate for such Interest Period plus the Applicable Rate; and (ii) each Base Rate Loan shall bear interest on the outstanding principal amount thereof from the applicable borrowing date at a rate per annum equal to the Base Rate plus the Applicable Rate.
- (b) If any amount payable by Borrower under any Loan Document is not paid when due (without regard to any applicable grace periods), whether at stated maturity by acceleration or otherwise, such amount shall thereafter bear interest at a fluctuating interest rate per annum at all times equal to the Default Rate to the fullest extent permitted by applicable Laws. Furthermore, while any Event of Default exists (or after acceleration), Borrower shall pay interest on the principal amount of all outstanding Obligations at a fluctuating interest rate per annum at all times equal to the Default Rate to the fullest extent permitted by applicable Laws. Accrued and unpaid interest on past due amounts (including interest on past due interest) shall be due and payable upon demand.
- (c) Interest on each Loan shall be due and payable in arrears on each Interest Payment Date applicable thereto and at such other times as may be specified herein. Interest hereunder shall be due and payable in accordance with the terms hereof before and after judgment, and before and after the commencement of any proceeding under any Debtor Relief Law.
- 2.07 Fees. In addition to certain fees described in subsections (e) and (f) of Section 2.03, Borrower shall pay to Bank a commitment fee equal to the Applicable Rate times the actual daily amount by which the Commitment exceeds the Total Outstandings. The commitment fee shall accrue at all times during the Availability Period, including at any time during which one or more conditions in Article IV is not met, and shall be due and payable quarterly in arrears on the first Business Day after the end of each March, June, September and December, commencing with the first such date to occur after the Closing Date, and on the Maturity Date. The commitment fee shall be calculated quarterly in arrears, and if there is any change in the Applicable Rate during any quarter, the actual daily amount shall be computed and multiplied by the Applicable Rate separately for each period during such quarter that such Applicable Rate was in effect.
- **2.08** Computation of Interest and Fees. All computations of interest for Base Rate Loans when the Base Rate is determined by Bank's "prime rate" shall be made on the basis of a year of 365 or 366 days, as the case may be, and actual days elapsed. All other computations of fees and interest shall be made on the basis of a 360-day year and actual days elapsed (which results in more fees or interest, as applicable, being paid than if computed on the basis of a 365-day year). Interest shall accrue on each Loan for the day on which the Loan is made, and shall not accrue on a Loan, or any portion thereof, for

17

the day on which the Loan or such portion is paid, <u>provided</u> that any Loan that is repaid on the same day on which it is made shall, subject to <u>Section 2.10(a)</u>, bear interest for one day.

2.09 Evidence of Debt. The Credit Extensions made by Bank shall be evidenced by one or more accounts or records maintained by Bank in the ordinary course of business. The accounts or records maintained by Bank shall be conclusive absent manifest error of the amount of the Credit Extensions made by Bank to Borrower and the interest and payments thereon. Any failure to so record or any error in doing so shall not, however, limit or otherwise affect the obligation of Borrower hereunder to pay any amount owing with respect to the Obligations. Borrower shall execute and deliver to Bank a Note, which shall evidence the Loans, in addition to such accounts or records. Bank may attach schedules to the Note and endorse thereon the date, Type, amount and maturity of each Loan and payments with respect thereto.

2.10 Payments Generally.

- (a) (i) All payments to be made by Borrower shall be made without condition or deduction for any counterclaim, defense, recoupment or setoff. Except as otherwise expressly provided herein, all payments by Borrower hereunder shall be made to Bank in Dollars and in immediately available funds not later than 3:00 p.m., Seattle time, on the date specified herein. All payments received by Bank after 3:00 p.m., Seattle time, shall be deemed received on the next succeeding Business Day and any applicable interest or fee shall continue to accrue.
 - (ii) On each date when the payment of any principal, interest or fees are due hereunder or under the Note, Borrower

agrees to maintain on deposit in the Borrower Account an amount sufficient to pay such principal, interest or fees in full on such date. Borrower hereby authorizes Bank (A) to deduct automatically all principal, interest or fees when due hereunder or under the Note from the Borrower Account, and (B) if and to the extent any payment of principal, interest or fees under this Agreement or the Note is not made when due to deduct any such amount from any or all of the accounts of Borrower maintained at Bank. Bank agrees to provide written notice to Borrower of any automatic deduction made pursuant to this Section 2.10(a)(ii) showing in reasonable detail the amounts of such deduction.

- (b) If any payment to be made by Borrower shall come due on a day other than a Business Day, payment shall be made on the next following Business Day, and such extension of time shall be reflected in computing interest or fees, as the case may be.
- (c) Nothing herein shall be deemed to obligate Bank to obtain the funds for any Loan in any particular place or manner or to constitute a representation by Bank that it has obtained or will obtain the funds for any Loan in any particular place or manner.

ARTICLE III TAXES, YIELD PROTECTION AND ILLEGALITY

3.01 Taxes.

(a) Any and all payments by Borrower to or for the account of Bank under any Loan Document shall be made free and clear of and without deduction for any and all present or future taxes, duties, levies, imposts, deductions, assessments, fees, withholdings or similar charges, and all liabilities with respect thereto, excluding taxes imposed on or measured by its overall net income, and franchise taxes imposed on it (in lieu of net income taxes), by the jurisdiction (or any political subdivision thereof) under the Laws of which Bank is organized or maintains a lending office (all such non-excluded taxes, duties, levies, imposts, deductions, assessments, fees, withholdings or similar charges, and liabilities being hereinafter referred to as "Taxes"). If Borrower shall be required by any Laws to deduct any Taxes from or in respect of any sum payable under any Loan Document to Bank, (i) the sum payable shall be increased as necessary so that after making all required deductions (including deductions applicable to additional sums payable under this Section), Bank receives an amount equal to the sum it would have received had no such deductions been made, (ii) Borrower shall make such deductions, (iii)

Borrower shall pay the full amount deducted to the relevant taxation authority or other authority in accordance with

18

applicable Laws, and (iv) within 30 days after the date of such payment, Borrower shall furnish to Bank the original or a certified copy of a receipt evidencing payment thereof.

- (b) In addition, Borrower agrees to pay any and all present or future stamp, court or documentary taxes and any other excise or property taxes or charges or similar levies which arise from any payment made under any Loan Document or from the execution, delivery, performance, enforcement or registration of, or otherwise with respect to, any Loan Document (hereinafter referred to as "Other Taxes").
- (c) If Borrower shall be required to deduct or pay any Taxes or Other Taxes from or in respect of any sum payable under any Loan Document to Bank, Borrower shall also pay to Bank, at the time interest is paid, such additional amount that Bank specifies is necessary to preserve the after-tax yield (after factoring in all taxes, including taxes imposed on or measured by net income) that Bank would have received if such Taxes or Other Taxes had not been imposed.
- (d) Borrower agrees to indemnify Bank for (i) the full amount of Taxes and Other Taxes (including any Taxes or Other Taxes imposed or asserted by any jurisdiction on amounts payable under this Section) paid by Bank, (ii) amounts payable under Section 3.01(c) and (iii) any liability (including additions to tax, penalties, interest and expenses) arising therefrom or with respect thereto, in each case whether or not such Taxes or Other Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. Payment under this subsection (d) shall be made within 30 days after the date Bank makes a demand therefor.
- 3.02 Illegality. If Bank determines that any Law has made it unlawful, or that any Governmental Authority has asserted that it is unlawful, for Bank or its Lending Office to make, maintain or fund Eurodollar Rate Loans, or to determine or charge interest rates based upon the Eurodollar Rate, then, on notice thereof by Bank to Borrower, any obligation of Bank to make or continue Eurodollar Rate Loans or to convert Base Rate Loans to Eurodollar Rate Loans shall be suspended until Bank notifies Borrower that the circumstances giving rise to such determination no longer exist. Upon receipt of such notice, Borrower shall, upon demand from Bank, prepay or, if applicable, convert all Eurodollar Rate Loans to Base Rate Loans, either on the last day of the Interest Period therefor, if Bank may lawfully continue to maintain such Eurodollar Rate Loans to such day, or immediately, if Bank may not lawfully continue to maintain such Eurodollar Rate Loans. Upon any such prepayment or conversion, Borrower shall also pay accrued interest on the amount so prepaid or converted.
- 3.03 Inability to Determine Eurodollar Rates. If Bank determines that for any reason adequate and reasonable means do not exist for determining the Eurodollar Base Rate for any requested Interest Period with respect to a proposed Eurodollar Rate Loan, or that the Eurodollar Base Rate for any requested Interest Period with respect to a proposed Eurodollar Rate Loan does not adequately and fairly reflect the cost to Bank of funding such Loan, Bank will promptly so notify Borrower. Thereafter, the obligation of Bank to make or maintain Eurodollar Rate Loans shall be suspended until Bank revokes such notice. Upon receipt of such notice, Borrower may revoke any pending request for a borrowing of, conversion to or continuation of a Eurodollar Rate Loan or, failing that, will be deemed to have converted such request into a request for a borrowing of a Base Rate Loan in the amount specified therein.

3.04 Increased Cost and Reduced Return; Capital Adequacy.

(a) If Bank determines that as a result of the introduction of or any change in or in the interpretation of any Law, or Bank's

from time to time upon demand of Bank, Borrower shall pay to Bank such additional amounts as will compensate Bank for such increased cost or reduction.

- (b) If Bank determines that the introduction of any Law regarding capital adequacy or any change therein or in the interpretation thereof, or compliance by Bank (or its Lending Office) therewith, has the effect of reducing the rate of return on the capital of Bank or any corporation controlling Bank as a consequence of Bank's obligations hereunder (taking into consideration its policies with respect to capital adequacy and Bank's desired return on capital), then from time to time upon demand of Bank, Borrower shall pay to Bank such additional amounts as will compensate Bank for such reduction.
- **3.05 Funding Losses.** Upon demand of Bank from time to time, Borrower shall promptly compensate Bank for and hold Bank harmless from any loss, cost or expense incurred by it as a result of:
- (a) any continuation, conversion, payment or prepayment of any Loan other than a Base Rate Loan on a day other than the last day of the Interest Period for such Loan (whether voluntary, mandatory, automatic, by reason of acceleration, or otherwise); or
- (b) any failure by Borrower (for a reason other than the failure of Bank to make a Loan) to prepay, borrow, continue or convert any Loan other than a Base Rate Loan on the date or in the amount notified by Borrower,

including any loss of anticipated profits and any loss or expense arising from the liquidation or reemployment of funds obtained by it to maintain such Loan or from fees payable to terminate the deposits from which such funds were obtained. Borrower shall also pay any customary administrative fees charged by Bank in connection with the foregoing.

For purposes of calculating amounts payable by Borrower to Bank under this <u>Section 3.05</u>, Bank shall be deemed to have funded each Eurodollar Rate Loan at the Eurodollar Base Rate used in determining the Eurodollar Rate for such Loan by a matching deposit or other borrowing in the London interbank eurodollar market for a comparable amount and for a comparable period, whether or not such Eurodollar Rate Loan was in fact so funded.

- **3.06** Requests for Compensation. A certificate of Bank claiming compensation under this Article III and setting forth the additional amount or amounts to be paid to it hereunder shall be conclusive in the absence of manifest error. In determining such amount, Bank may use any reasonable averaging and attribution methods.
- **3.07 Survival.** All of Borrower's obligations under this <u>Article III</u> shall survive termination of the Commitment and repayment of all other Obligations hereunder.

ARTICLE IV CONDITIONS PRECEDENT TO CREDIT EXTENSIONS

- **4.01 Conditions of Initial Credit Extension.** The obligation of Bank to make its initial Credit Extension hereunder is subject to satisfaction of the following conditions precedent:
- (a) Bank's receipt of the following, each of which shall be originals or facsimiles (followed promptly by originals) unless otherwise specified, each properly executed by a Responsible Officer of the signing Loan Party, each dated the Closing Date (or, in the case of certificates of governmental officials, a recent date before the Closing Date) and each in form and substance satisfactory to Bank and its legal counsel:
 - (i) executed counterparts of this Agreement, sufficient in number for distribution to Bank and Borrower;
 - (ii) a Note executed by Borrower in favor of Bank;

- (iii) executed counterparts of the Guaranty, sufficient in number for distribution to Bank and Borrower;
- (iv) such certificates of resolutions or other action, incumbency certificates and/or other certificates of Responsible Officers of each Loan Party as Bank may require evidencing the identity, authority and capacity of each Responsible Officer thereof authorized to act as a Responsible Officer in connection with this Agreement and the other Loan Documents to which such Loan Party is a party;
- (v) such documents and certifications as Bank may reasonably require to evidence that each Loan Party is duly organized or formed and that Borrower and each Guarantor is, validly existing, in good standing and qualified to engage in

business in each jurisdiction where its ownership, lease or operation of properties or the conduct of its business requires such qualification, except to the extent that failure to do so could not reasonably be expected to have a Material Adverse Effect;

- (vi) a certificate of a Responsible Officer of each Loan Party either (A) attaching copies of all consents, licenses and approvals required in connection with the execution, delivery and performance by such Loan Party and the validity against such Loan Party of the Loan Documents to which it is a party, and such consents, licenses and approvals shall be in full force and effect, or (B) stating that no such consents, licenses or approvals are so required;
- (vii) a certificate signed by a Responsible Officer of Borrower certifying (A) that the conditions specified in Sections 4.02(a) and (b) have been satisfied, and (B) that there has been no event or circumstance since the date of the Audited Financial Statements that has had or could be reasonably be expected to have a Material Adverse Effect;
- (viii) evidence that all insurance required to be maintained pursuant to the Loan Documents has been obtained and is in effect; and
- (ix) such other assurances, certificates, documents, consents, evidence of perfection of all Liens securing the Obligations or opinions as Bank reasonably may require.
- (b) Bank's receipt of the Collateral Documents, each of which shall be originals or facsimiles (followed promptly by originals) unless otherwise specified, each properly executed by a Responsible Officer of the signing Loan Party, each dated the Closing Date (or, in the case of certificates of governmental officials, a recent date before the Closing Date) and each in form and substance satisfactory to Bank and its legal counsel, together with:
 - (i) acknowledgment copies of all Uniform Commercial Code financing statements filed to perfect the security interests of Bank, or other evidence satisfactory to Bank that there has been filed all financing statements, registrations and other recordings necessary and advisable to perfect the Liens of Bank in accordance with applicable law;
 - (ii) all certificates and instruments representing the Collateral (including the Capital Stock of each first tier Material Subsidiary of the types and in the percentages described in <u>Section 6.13(b)</u> based on the organizational chart referred to in subsection (c) below), stock transfer powers executed in blank with signatures guaranteed as Bank may specify and, if relevant, a completed Federal Reserve Form U-1 for Bank;
 - (iii) evidence that all other actions necessary or, in the opinion of Bank, desirable to perfect and protect the first priority security interest created by the Collateral Documents and to enhance the ability of Bank to preserve and protect its interests in the Collateral have been taken.

- (c) Bank's receipt of the following information:
- (i) a financial projection for Borrower and its Subsidiaries, prepared on a quarterly basis, for the fiscal year ending January 1, 2005 setting forth the projected revenues, expenses, assets, liabilities and equity and the underlying assumptions therefore, all in reasonable detail and certified by a Responsible Officer of Borrower as having been prepared and furnished to Bank in good faith and based on estimates and assumptions that were believed by the management of Borrower to be reasonable in light of the then current and foreseeable business conditions of Borrower and its Subsidiaries; and
- (ii) an organizational chart for Borrower and its Subsidiaries as of January 3, 2004, setting forth the identity, ownership, location, revenues, assets and equity of each Person legally or beneficially owned, directly, or indirectly through one or more intermediaries, by Borrower, certified by a Responsible Officer of Borrower as being true and correct in all material respects.
- (d) Any fees required to be paid on or before the Closing Date shall have been paid.
- (e) Borrower shall have paid all Attorney Costs of Bank to the extent invoiced prior to or on the Closing Date, <u>plus</u> such additional amounts of Attorney Costs as shall constitute its reasonable estimate of Attorney Costs incurred or to be incurred by it through the closing proceedings (<u>provided</u> that such estimate shall not thereafter preclude a final settling of accounts between Borrower and Bank).
 - (f) The Closing Date shall have occurred on or before July 15, 2004.
- **4.02** Conditions to all Credit Extensions. The obligation of Bank to make any Credit Extension is subject to the following conditions precedent:
- (a) The representations and warranties of Borrower and each other Loan Party contained in <u>Article V</u> or any other Loan Document, or which are contained in any document furnished at any time under or in connection herewith or therewith, shall be true and correct in all material respects on and as of the date of such Credit Extension, except to the extent that such representations and warranties specifically refer to an earlier date, in which case they shall be true and correct as of such earlier date, and except that for purposes of this <u>Section 4.02</u>, the representations and warranties contained in subsections (a) and (b) of <u>Section 5.05</u> shall be deemed to refer to the most recent statements furnished pursuant to clauses (a) and (b), respectively, of <u>Section 6.01</u>.

- (b) No Default shall exist, or would result from such proposed Credit Extension.
- (c) Bank shall have received a Request for Credit Extension in accordance with the requirements hereof.
- (d) Bank shall have received, in form and substance satisfactory to it, such other assurances, certificates, documents or consents related to the foregoing as Bank reasonably may require.

Each Request for Credit Extension (other than a Loan Notice requesting only a conversion of a Loan to the other Type or a continuation of a Eurodollar Rate Loan) submitted by Borrower shall be deemed to be a representation and warranty that the conditions specified in Sections 4.02(a) and (b) have been satisfied on and as of the date of the applicable Credit Extension.

ARTICLE V REPRESENTATIONS AND WARRANTIES

Borrower represents and warrants to Bank that:

5.01 Existence, Qualification and Power; Compliance with Laws. Each Loan Party (a) is a corporation or limited liability company duly organized or formed, validly existing and in good standing under the Laws of the jurisdiction of its incorporation or organization, (b) has all requisite power and

22

authority and all requisite governmental licenses, authorizations, consents and approvals to (i) own its assets and carry on its business and (ii) execute, deliver and perform its obligations under the Loan Documents to which it is a party, (c) is duly qualified and is licensed and in good standing under the Laws of each jurisdiction where its ownership, lease or operation of properties or the conduct of its business requires such qualification or license, and (d) is in compliance with all Laws; except in each case referred to in clauses (b)(i), (c) or (d), to the extent that failure to do so could not reasonably be expected to have a Material Adverse Effect.

- **5.02 Authorization; No Contravention.** The execution, delivery and performance by each Loan Party of each Loan Document to which such Person is party, have been duly authorized by all necessary corporate or other organizational action, and do not and will not (a) contravene the terms of any of such Person's Organization Documents; (b) conflict with or result in any breach or contravention of, or the creation of any Lien under, (i) any Contractual Obligation to which such Person is a party or (ii) any order, injunction, writ or decree of any Governmental Authority or any arbitral award to which such Person or its property is subject; or (c) violate any Law.
- **5.03 Governmental Authorization; Other Consents.** No approval, consent, exemption, authorization, or other action by, or notice to, or filing with, any Governmental Authority or any other Person is necessary or required in connection with the execution, delivery or performance by, or enforcement against, any Loan Party of this Agreement or any other Loan Document.
- **5.04 Binding Effect.** This Agreement has been, and each other Loan Document, when delivered hereunder, will have been, duly executed and delivered by each Loan Party that is party thereto. This Agreement constitutes, and each other Loan Document when so delivered will constitute, a legal, valid and binding obligation of such Loan Party, enforceable against each Loan Party that is party thereto in accordance with its terms, except as may be limited by equitable principles relating to enforceability.

5.05 Financial Statements; No Material Adverse Effect.

- (a) The Audited Financial Statements (i) were prepared in accordance with GAAP consistently applied throughout the period covered thereby, except as otherwise expressly noted therein; (ii) fairly present the financial condition of Borrower and its Subsidiaries as of the date thereof and their results of operations for the period covered thereby in accordance with GAAP consistently applied throughout the period covered thereby, except as otherwise expressly noted therein; and (iii) show all material indebtedness and other liabilities, direct or contingent, of Borrower and its Subsidiaries as of the date thereof, including liabilities for taxes, material commitments and Indebtedness.
- (b) The unaudited consolidated balance sheet of Borrower and its Subsidiaries dated April 3, 2004, and the related consolidated statements of income or operations, shareholders' equity and cash flows for the fiscal quarter ended on that date and for the portion of Borrower's fiscal year then ended (i) were prepared in accordance with GAAP consistently applied throughout the period covered thereby, except as otherwise expressly noted therein, and; (ii) fairly present the financial condition of Borrower and its Subsidiaries as of the date thereof and their results of operations for the period covered thereby in accordance with GAAP consistently applied throughout the period covered thereby, subject in the case of clauses (i) and (ii), to the absence of footnotes and to normal year-end audit adjustments. Schedule 5.05 sets forth all material indebtedness and other liabilities, direct or contingent, of Borrower and its consolidated Subsidiaries as of the date of such financial statements, including liabilities for taxes, material commitments and Indebtedness.
- (c) Since the date of the Audited Financial Statements, there has been no event or circumstance, either individually or in the aggregate, that has had or could reasonably be expected to have a Material Adverse Effect.
- **5.06 Litigation.** Except as specifically disclosed in <u>Schedule 5.06</u>, there are no actions, suits, proceedings, claims or disputes pending or, to the knowledge of Borrower after due and diligent

investigation, threatened or contemplated, at law, in equity, in arbitration or before any Governmental Authority, by or against Borrower or any of its Subsidiaries or against any of their properties or revenues that (a) purport to affect or pertain to this Agreement or any other Loan Document, or any of the transactions contemplated hereby, or (b) either individually or in the aggregate, if determined adversely, could reasonably be expected to have a Material Adverse Effect.

- **5.07 No Default.** Neither Borrower nor any Subsidiary is in default under or with respect to any Contractual Obligation that could either individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. No Default has occurred and is continuing or would result from the consummation of the transactions contemplated by this Agreement or any other Loan Document.
- **5.08 Ownership of Property; Liens.** Each of Borrower and each Subsidiary has good record and marketable title in fee simple to, or valid leasehold interests in, all real property necessary or used in the ordinary conduct of its business, except for such defects in title as could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. The property of Borrower and its Subsidiaries is subject to no Liens, other than Liens permitted by Section 7.01.
- **5.09 Environmental Compliance.** Borrower and its Subsidiaries conduct in the ordinary course of business a review of the effect of existing Environmental Laws and claims alleging potential liability or responsibility for violation of any Environmental Law on their respective businesses, operations and properties, and as a result thereof Borrower has reasonably concluded that, except as specifically disclosed in <u>Schedule 5.09</u>, such Environmental Laws and claims could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- **5.10 Insurance.** The properties of Borrower and its Subsidiaries are insured with financially sound and reputable insurance companies not Affiliates of Borrower, in such amounts, after giving effect to any self-insurance compatible with the following standards, with such deductibles and covering such risks as are customarily carried by companies engaged in similar businesses and owning similar properties in localities where Borrower or the applicable Subsidiary operates.
- **5.11 Taxes.** Borrower and its Subsidiaries have filed all Federal, state and other material tax returns and reports required to be filed, and have paid all Federal, state and other material taxes, assessments, fees and other governmental charges levied or imposed upon them or their properties, income or assets otherwise due and payable, except those which are being contested in good faith by appropriate proceedings diligently conducted and for which adequate reserves have been provided in accordance with GAAP. To the best knowledge of Borrower, there is no proposed tax assessment against Borrower or any Subsidiary that would, if made, have a Material Adverse Effect.

5.12 ERISA Compliance.

- (a) Each Plan is in compliance in all material respects with the applicable provisions of ERISA, the Code and other Federal or state Laws. Each Plan that is intended to qualify under Section 401(a) of the Code has received a favorable determination letter from the IRS or an application for such a letter is currently being processed by the IRS with respect thereto and, to the best knowledge of Borrower, nothing has occurred which would prevent, or cause the loss of, such qualification. Borrower and each ERISA Affiliate have made all required contributions to each Plan subject to Section 412 of the Code, and no application for a funding waiver or an extension of any amortization period pursuant to Section 412 of the Code has been made with respect to any Plan.
- (b) There are no pending or, to the best knowledge of Borrower, threatened claims, actions or lawsuits, or action by any Governmental Authority, with respect to any Plan that could reasonably be expected to have a Material Adverse Effect. There has been no prohibited transaction or violation of the fiduciary responsibility rules with respect to any Plan that has resulted or could reasonably be expected to result in a Material Adverse Effect.

24

- (c) (i) No ERISA Event has occurred or is reasonably expected to occur; (ii) no Pension Plan has any Unfunded Pension Liability; (iii) neither Borrower nor any ERISA Affiliate has incurred, or reasonably expects to incur, any liability under Title IV of ERISA with respect to any Pension Plan (other than premiums due and not delinquent under Section 4007 of ERISA); (iv) neither Borrower nor any ERISA Affiliate has incurred, or reasonably expects to incur, any liability (and no event has occurred which, with the giving of notice under Section 4219 of ERISA, would result in such liability) under Sections 4201 or 4243 of ERISA with respect to a Multiemployer Plan; and (v) neither Borrower nor any ERISA Affiliate has engaged in a transaction that could be subject to Sections 4069 or 4212(c) of ERISA.
- **5.13 Subsidiaries.** As of the Closing Date, Borrower has no Subsidiaries other than those specifically disclosed in <u>Part (a)</u> of <u>Schedule 5.13</u> and has no equity investments in any other corporation or entity other than those specifically disclosed in <u>Part (b)</u> of <u>Schedule 5.13</u>.

5.14 Margin Regulations; Investment Company Act; Public Utility Holding Company Act.

- (a) Borrower is not engaged and will not engage, principally or as one of its important activities, in the business of purchasing or carrying margin stock (within the meaning of Regulation U issued by the Board of Governors of the Federal Reserve System of the United States), or extending credit for the purpose of purchasing or carrying margin stock.
- (b) None of Borrower, any Person Controlling Borrower, or any Subsidiary (i) is a "holding company," or a "subsidiary company" of a "holding company," or an "affiliate" of a "holding company" or of a "subsidiary company" of a "holding company," within

the meaning of the Public Utility Holding Company Act of 1935, or (ii) is or is required to be registered as an "investment company" under the Investment Company Act of 1940.

- **5.15 Disclosure.** Borrower has disclosed to Bank all agreements, instruments and corporate or other restrictions to which it or any of its Subsidiaries is subject, and all other matters known to it, that, individually or in the aggregate, could reasonably be expected to result in a Material Adverse Effect. No report, financial statement, certificate or other information furnished (whether in writing or orally) by or on behalf of any Loan Party to Bank in connection with the transactions contemplated hereby and the negotiation of this Agreement or delivered hereunder (as modified or supplemented by other information so furnished) contains any material misstatement of fact or omits to state any material fact necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading; provided that, with respect to projected financial information, Borrower represents only that such information was prepared in good faith based upon assumptions believed to be reasonable at the time.
- **5.16** Compliance with Laws. Borrower, each Subsidiary and each other Loan Party is in compliance in all material respects with the requirements of all Laws and all orders, writs, injunctions and decrees applicable to it or to its properties, except in such instances in which (a) such requirement of Law or order, writ, injunction or decree is being contested in good faith by appropriate proceedings diligently conducted or (b) the failure to comply therewith, either individually or in the aggregate, could not reasonably be expected to have a Material Adverse Effect.
- 5.17 Intellectual Property; Licenses, Etc. Borrower and its Subsidiaries own, or possess the right to use, all of the trademarks, service marks, trade names, copyrights, patents, patent rights, franchises, licenses and other intellectual property rights (collectively, "IP Rights") that are reasonably necessary for the operation of their respective businesses, without conflict with the rights of any other Person. To the best knowledge of Borrower, no slogan or other advertising device, product, process, method, substance, part or other material now employed, or now contemplated to be employed, by Borrower or any Subsidiary infringes upon any rights held by any other Person. Except as specifically disclosed in Schedule 5.18, no claim or litigation regarding any of the foregoing is pending or, to the best knowledge of Borrower, threatened, which, either individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect.

25

5.18 Rights in Collateral; Priority of Liens. Borrower and each other Loan Party own the property granted by it as Collateral under the Collateral Documents, free and clear of any and all Liens in favor of third parties. Upon the proper filing of UCC financing statements, and the taking of the other actions required by Bank, the Liens granted pursuant to the Collateral Documents will constitute valid and enforceable first, prior and perfected Liens on the Collateral in favor of Bank.

ARTICLE VI AFFIRMATIVE COVENANTS

So long as the Commitment shall be in effect, any Loan or other Obligation hereunder shall remain unpaid or unsatisfied, or any Letter of Credit shall remain outstanding, Borrower shall, and shall (except in the case of the covenants set forth in <u>Sections 6.01</u>, <u>6.02</u>, <u>6.03</u> and <u>6.11</u>) cause each Subsidiary to:

- **6.01 Financial Statements.** Deliver to Bank, in form and detail satisfactory to Bank:
- (a) as soon as available, but in any event within 90 days after the end of each fiscal year of Borrower, a consolidated balance sheet of Borrower and its Subsidiaries as at the end of such fiscal year, and the related consolidated statements of income or operations, shareholders' equity and cash flows for such fiscal year, setting forth in each case in comparative form the figures for the previous fiscal year, all in reasonable detail and prepared in accordance with GAAP, audited and accompanied by a report and opinion of an independent certified public accountant of nationally recognized standing reasonably acceptable to Bank, which report and opinion shall be prepared in accordance with generally accepted auditing standards and shall not be subject to any "going concern" or like qualification or exception or any qualification or exception as to the scope of such audit; and
- (b) as soon as available, but in any event within 45 days after the end of each of the first three fiscal quarters of each fiscal year of Borrower, a consolidated balance sheet of Borrower and its Subsidiaries as at the end of such fiscal quarter, and the related consolidated statements of income or operations, shareholders' equity and cash flows for such fiscal quarter and for the portion of Borrower's fiscal year then ended, setting forth in each case in comparative form the figures for the corresponding fiscal quarter of the previous fiscal year and the corresponding portion of the previous fiscal year, all in reasonable detail and certified by a Responsible Officer of Borrower as fairly presenting the financial condition, results of operations, shareholders equity and cash flows of Borrower and its Subsidiaries in accordance with GAAP, subject only to normal year-end audit adjustments and the absence of footnotes.

As to any information contained in materials furnished pursuant to <u>Section 6.02(f)</u>, Borrower shall not be separately required to furnish such information under clause (a) or (b) above, but the foregoing shall not be in derogation of the obligation of Borrower to furnish the information and materials described in subsections (a) and (b) above at the times specified therein.

- **6.02** Certificates; Other Information. Deliver to Bank, in form and detail satisfactory to Bank:
- (a) concurrently with the delivery of the financial statements referred to in Section 6.01(a), a certificate of its independent certified public accountants certifying such financial statements;
- (b) concurrently with the delivery of the financial statements referred to in Sections 6.01(a) and (b), a duly completed Compliance Certificate signed by a Responsible Officer of Borrower;

- (c) promptly after any request by Bank, copies of any detailed audit reports, management letters or recommendations submitted to the board of directors (or the audit committee of the board of directors) of Borrower by independent accountants in connection with the accounts or books of Borrower or any Subsidiary, or any audit of any of them;
- (d) as soon as available, but in any event within 90 days after the end of each fiscal year of Borrower (commencing with the fiscal year ended January 1, 2005), a financial projection for Borrower

26

and its Subsidiaries, prepared on a quarterly basis, for the next succeeding fiscal year setting forth the projected revenues, expenses, assets, liabilities and equity and the underlying assumptions therefore, all in reasonable detail and certified by a Responsible Officer of Borrower as having been prepared and furnished to Bank in good faith and based on estimates and assumptions that were believed by the management of Borrower to be reasonable in light of the then current and foreseeable business conditions of Borrower and its Subsidiaries:

- (e) as soon as available, but in any event within 90 days after the end of each fiscal year of Borrower (commencing with the fiscal year ended January 1, 2005), an organizational chart for Borrower and its Subsidiaries as of such fiscal year end, setting forth the identity, ownership, location, revenues, assets and equity of each Person legally or beneficially owned, directly, or indirectly through one or more intermediaries, by Borrower, certified by a Responsible Officer of Borrower as being true and correct in all material respects.
- (f) promptly after the same are available, copies of each annual report, proxy or financial statement or other report or communication sent to the stockholders of Borrower, and copies of all annual, regular, periodic and special reports and registration statements which Borrower may file or be required to file with the Securities and Exchange Commission under Section 13 or 15(d) of the Securities Exchange Act of 1934, and not otherwise required to be delivered to Bank pursuant hereto; and
- (g) promptly, such additional information regarding the business, financial or corporate affairs of Borrower or any Subsidiary, or compliance with the terms of the Loan Documents, as Bank may from time to time reasonably request.
 - **6.03 Notices.** Promptly notify Bank:
 - (a) of the occurrence of any Default;
- (b) of any matter that has resulted or could reasonably be expected to result in a Material Adverse Effect, including (i) breach or non-performance of, or any default under, a Contractual Obligation of Borrower or any Subsidiary; (ii) any dispute, litigation, investigation, proceeding or suspension between Borrower or any Subsidiary and any Governmental Authority; (iii) any material loss of, damage to or destruction of the Collateral or any part thereof, or (iv) the commencement of, or any material development in, any litigation or proceeding affecting Borrower or any Subsidiary, including pursuant to any applicable Environmental Laws;
 - (c) of the occurrence of any ERISA Event; and
 - (d) of any material change in accounting policies or financial reporting practices by Borrower or any Subsidiary.

Each notice pursuant to this Section shall be accompanied by a statement of a Responsible Officer of Borrower setting forth details of the occurrence referred to therein and stating what action Borrower has taken and proposes to take with respect thereto. Each notice pursuant to Section 6.03(a) shall describe with particularity any and all provisions of this Agreement and any other Loan Document that have been breached.

6.04 Payment of Obligations. Pay and discharge as the same shall become due and payable, all its obligations and liabilities, including (a) all tax liabilities, assessments and governmental charges or levies upon it or its properties or assets, unless the same are being contested in good faith by appropriate proceedings diligently conducted and adequate reserves in accordance with GAAP are being maintained by Borrower or such Subsidiary; (b) all lawful claims which, if unpaid, would by law become a Lien upon its property; and (c) all Indebtedness, as and when due and payable, but subject to any subordination provisions contained in any instrument or agreement evidencing such Indebtedness.

- 6.05 Preservation of Existence, Etc. (a) Preserve, renew and maintain in full force and effect its legal existence and good standing under the Laws of the jurisdiction of its organization, except in a transaction permitted by Section 7.04 or 7.05; (b) take all reasonable action to maintain all rights, privileges, permits, licenses and franchises necessary or desirable in the normal conduct of its business, except to the extent that failure to do so could not reasonably be expected to have a Material Adverse Effect; and (c) preserve or renew all of its registered patents, trademarks, trade names and service marks, the non-preservation of which could reasonably be expected to have a Material Adverse Effect.
- **6.06 Maintenance of Properties.** (a) Maintain, preserve and protect all of its material properties and equipment necessary in the operation of its business in good working order and condition, ordinary wear and tear excepted; (b) make all necessary repairs thereto and renewals and replacements thereof except where the failure to do so could not reasonably be expected to have a Material Adverse Effect; and (c) use the standard of care typical in the industry in the operation and maintenance of its facilities.

- **6.07 Maintenance of Insurance.** In addition to insurance requirements set forth in the Collateral Documents, maintain with financially sound and reputable insurance companies not Affiliates of Borrower, insurance with respect to its properties and business against loss or damage of the kinds customarily insured against by Persons engaged in the same or similar business, of such types and in such amounts (after giving effect to any self-insurance compatible with the following standards) as are customarily carried under similar circumstances by such other Persons and providing for not less than 30 days' prior notice to Bank of termination, lapse or cancellation of such insurance. All casualty insurance maintained by Borrower in respect of the Collateral shall name Bank as loss payee and all liability insurance shall name Bank as additional insured. Upon request of Bank, furnish Bank, at reasonable intervals (but not more than once per calendar year) a certificate of a Responsible Officer of Borrower (and, if requested by Bank, any insurance broker of Borrower) setting forth the nature and extent of all insurance maintained by Borrower and its Subsidiaries in accordance with this Section 6.07 or any Collateral Documents (and which, in the case of a certificate of a broker, were placed through such broker).
- **6.08** Compliance with Laws. Comply in all material respects with the requirements of all Laws, and all orders, writs, injunctions and decrees applicable to it or to its business or property, except in such instances in which (a) such requirement of Law or order, writ, injunction or decree is being contested in good faith by appropriate proceedings diligently conducted; or (b) the failure to comply therewith could not reasonably be expected to have a Material Adverse Effect.
- **6.09 Books and Records.** (a) Maintain proper books of record and account, in which full, true and correct entries in conformity with GAAP consistently applied shall be made of all financial transactions and matters involving the assets and business of Borrower or such Subsidiary, as the case may be; and (b) maintain such books of record and account in material conformity with all applicable requirements of any Governmental Authority having regulatory jurisdiction over Borrower or such Subsidiary, as the case may be. Borrower shall maintain at all times books and records pertaining to the Collateral in such detail, form and scope as Bank shall reasonably require.
- **6.10 Inspection Rights.** Permit representatives and independent contractors of Bank to visit and inspect any of its properties, to examine its corporate, financial and operating records, and make copies thereof or abstracts therefrom, and to discuss its affairs, finances and accounts with its directors, officers, and independent public accountants, all at the expense of Borrower and at such reasonable times during normal business hours and as often as may be reasonably desired, upon reasonable advance notice to Borrower; provided, however, that when a Default exists Bank (or any of its representatives or independent contractors) may do any of the foregoing at the expense of Borrower at any time during normal business hours and without advance notice.
- **6.11 Use of Proceeds.** Use the proceeds of the Credit Extensions to fund Permitted Acquisitions and for general corporate purposes not in contravention of any Law or of any Loan Document.

28

6.12 Financial Covenants.

- (a) <u>Consolidated EBITDA</u>. Maintain on a consolidated basis, for each fiscal quarter of Borrower, Consolidated EBITDA equal to or greater than Five Million Dollars (\$5,000,000).
- (b) <u>Consolidated Tangible Net Worth</u>. Maintain on a consolidated basis, as of the end of each fiscal quarter of Borrower, Consolidated Tangible Net Worth equal to or greater than the Minimum Tangible Net Worth.
- (c) <u>Consolidated Leverage Ratio.</u> Maintain a Consolidated Leverage Ratio, as of the end of each fiscal quarter of Borrower, equal to or less than 1.0 to 1.0.

6.13 Additional Guarantors; Pledge of Capital Stock.

- (a) Additional Guarantors. Promptly (and in any event within 30 days) following the date an organizational chart is delivered pursuant to Section 6.02(e), cause each Material Subsidiary that is a Domestic Subsidiary and is not then a Guarantor to (a) become a Guarantor by executing and delivering to Bank a counterpart of the Guaranty or such other document as Bank shall deem appropriate for such purpose and (b) deliver to Bank documents of the types referred to in clauses (iv) and (v) of Section 4.01(a), all in form, content and scope reasonably satisfactory to Bank.
- (b) Pledge of Capital Stock. Promptly (and in any event within 30 days) following the date an organizational chart is delivered pursuant to Section 6.02(e), (i) grant a Lien in favor of Bank on (A) 100% of the Capital Stock of each first tier Material Subsidiary that is a Domestic Subsidiary, (B) 100% of the Capital Stock of each first tier Material Subsidiary that is a Disregarded Foreign Subsidiary whose Capital Stock is not owned in any part by a Foreign Subsidiary (other than a Disregarded Foreign Subsidiary), (C) 65% of the voting Capital Stock and 100% of the non-voting Capital Stock of each first tier (after ignoring all Disregarded Foreign Subsidiaries which may be a direct or indirect stockholder) Material Subsidiary that is a Foreign Subsidiary, or (D) the remaining Capital Stock of any first tier Material Subsidiary that was a Foreign Subsidiary (other than a Disregarded Foreign Subsidiary) but that later became a Disregarded Foreign Subsidiary, in each case to the extent Bank does not already have a Lien on such Capital Stock pursuant to the Pledge Agreement, to secure the Obligations by executing and delivering to Bank a supplement to the Pledge Agreement or such other document as Bank shall deem appropriate for such purpose and (ii) deliver to Bank all certificates and instruments representing the Capital Stock described in clause (i) together with stock transfer powers executed in blank with signatures guaranteed as Bank may specify and, if relevant, a completed Federal Reserve Form U-1 for Bank and (iii) take such action as may be necessary or otherwise requested by Bank to ensure that the Lien described in clause (i) is a perfected Lien of first priority.
- **6.14 Security Interests.** Defend, and to cause each other Loan Party to, defend the Collateral against all claims and demands of all Persons at any time claiming the same or any interest therein. Borrower shall, and shall cause each other Loan Party, to do whatever Bank may reasonably request, from time to time, to effect the purposes of this Agreement and the other Loan Documents.

ARTICLE VII NEGATIVE COVENANTS

So long as the Commitment shall be in effect, any Loan or other Obligation hereunder shall remain unpaid or unsatisfied, or any Letter of Credit shall remain outstanding, Borrower shall not, nor shall it permit any Subsidiary to, directly or indirectly:

- **7.01 Liens.** Create, incur, assume or suffer to exist, any Lien upon any of its property, assets or revenues, whether now owned or hereafter acquired, other than the following:
 - (a) Liens pursuant to any Loan Document;

29

- (b) Liens existing on the date hereof and listed on <u>Schedule 7.01</u> and any renewals or extensions thereof, <u>provided</u> that the property covered thereby is not increased and any renewal or extension of the obligations secured or benefited thereby is permitted by <u>Section 7.03(b)</u>;
- (c) Liens for taxes not yet due or which are being contested in good faith and by appropriate proceedings diligently conducted, if (i) adequate reserves with respect thereto are maintained on the books of the applicable Person in accordance with GAAP, and (ii) in the case of a Lien against any of the Collateral, such contest proceedings operate to stay the sale of any portion of the Collateral to satisfy such taxes;
- (d) carriers', warehousemen's, mechanics', materialmen's, repairmen's or other like Liens arising in the ordinary course of business which are not overdue for a period of more than 30 days or which are being contested in good faith and by appropriate proceedings diligently conducted, if (i) adequate reserves with respect thereto are maintained on the books of the applicable Person in accordance with GAAP and (ii) in the case of a Lien against any of the Collateral, such contest proceedings operate to stay the sale of any portion of the Collateral to satisfy such Liens;
- (e) pledges or deposits in the ordinary course of business in connection with workers' compensation, unemployment insurance and other social security legislation, other than any Lien imposed by ERISA;
- (f) deposits to secure the performance of bids, trade contracts and leases (other than Indebtedness), statutory obligations, surety bonds (other than bonds related to judgments or litigation), performance bonds and other obligations of a like nature incurred in the ordinary course of business;
- (g) easements, rights-of-way, restrictions and other similar encumbrances affecting real property which, in the aggregate, are not substantial in amount, and which do not in any case materially detract from the value of the property subject thereto or materially interfere with the ordinary conduct of the business of the applicable Person:
- (h) Liens securing judgments for the payment of money not constituting an Event of Default under <u>Section 8.01(h)</u> or securing appeal or other surety bonds related to such judgments; and
- (i) Liens securing Indebtedness permitted under <u>Section 7.03(f)</u>; <u>provided</u> that (i) such Liens do not at any time encumber any property other than the property financed by such Indebtedness and (ii) the Indebtedness secured thereby does not exceed the cost or fair market value, whichever is lower, of the property being acquired on the date of acquisition.
 - **7.02 Investments.** Make any Investments, except:
- (a) Investments held by Borrower or such Subsidiary in the form of cash equivalents, short-term marketable debt securities or investment grade marketable equity securities;
- (b) advances to officers, directors and employees of Borrower and Subsidiaries in an aggregate amount not to exceed \$250,000 at any time outstanding, for travel, entertainment, relocation and analogous ordinary business purposes;
- (c) Investments of Borrower in any wholly-owned Subsidiary that, prior to making such Investment, was a Guarantor and Investments of any Guarantor in Borrower or in another Guarantor;
- (d) Permitted Acquisitions made by Borrower or any Subsidiary not exceeding \$10,000,000 in aggregate consideration (including assumed liabilities, earnout payments and any other deferred payment) in any fiscal year of Borrower;

- (e) Investments of Borrower in any Subsidiary that is not a Guarantor or a Pledged Subsidiary and Investments of any Guarantor in any Subsidiary that is not a Guarantor not exceeding \$1,000,000 in the aggregate in any fiscal year of Borrower;
- (f) Investments consisting of extensions of credit in the nature of accounts receivable or notes receivable arising from the grant of trade credit in the ordinary course of business, and Investments received in satisfaction or partial satisfaction thereof from financially troubled account debtors to the extent reasonably necessary in order to prevent or limit loss;

- (g) Guarantees permitted by Section 7.03; and
- (h) other Investments made in the ordinary course of business not exceeding \$500,000 in the aggregate in any fiscal year of Borrower.
 - 7.03 Indebtedness. Create, incur, assume or suffer to exist any Indebtedness, except:
 - (a) Indebtedness under the Loan Documents;
- (b) Indebtedness outstanding on the date hereof and listed on Schedule 7.03 and any refinancings, refundings, renewals or extensions thereof; provided that the amount of such Indebtedness is not increased at the time of such refinancing, refunding, renewal or extension except by an amount equal to a reasonable premium or other reasonable amount paid, and fees and expenses reasonably incurred, in connection with such refinancing and by an amount equal to any existing commitments unutilized thereunder; provided, further, that if such Indebtedness constitutes Subordinated Liabilities, such Indebtedness is refinanced on subordination terms at least as favorable to Bank and no more restrictive on Borrower than the Subordinated Liabilities being refinanced, and in an amount not less than the amount outstanding at the time of refinancing;
- (c) obligations (contingent or otherwise) of any Subsidiary existing or arising under bank guaranties issued by Bank in an aggregate principal amount not to exceed \$2,000,000 at any time outstanding.
- (d) Guarantees of Borrower or any Guarantor in respect of Indebtedness otherwise permitted hereunder of Borrower or any other Guarantor;
- (e) obligations (contingent or otherwise) of Borrower or any Subsidiary existing or arising under any Swap Contract, provided that (i) such obligations are (or were) entered into by such Person in the ordinary course of business for the purpose of directly mitigating risks associated with liabilities, commitments, investments, assets, or property held or reasonably anticipated by such Person, or changes in the value of securities issued by such Person and not for purposes of speculation or taking a "market view;" and (ii) such Swap Contract does not contain any provision exonerating the non-defaulting party from its obligation to make payments on outstanding transactions to the defaulting party;
- (f) Indebtedness in respect of capital leases, Synthetic Lease Obligations and purchase money obligations for fixed or capital assets within the limitations set forth in Section 7.01(i); provided, however, that the aggregate amount of all such Indebtedness at any one time outstanding shall not exceed \$500,000; and
 - (g) unsecured Indebtedness in an aggregate principal amount not to exceed \$500,000 at any time outstanding.
- **7.04 Fundamental Changes.** Merge, dissolve, liquidate, consolidate with or into, another Person, or Dispose of (whether in one transaction or in a series of transactions) all or substantially all of its assets (whether now owned or hereafter acquired) to or in favor of any Person, except that, so long as no Default exists or would result therefrom:

- (a) any Subsidiary may merge with (i) Borrower, <u>provided</u> that Borrower shall be the continuing or surviving Person, or (ii) any one or more other Subsidiaries, <u>provided</u> that when any wholly-owned Subsidiary is merging with another Subsidiary, the wholly-owned Subsidiary shall be the continuing or surviving Person, and, <u>provided further</u> that if a Guarantor is merging with another Subsidiary, the Guarantor shall be the continuing or surviving Person; and
- (b) any Subsidiary may Dispose of all or substantially all of its assets (upon voluntary liquidation or otherwise), to Borrower or to another Subsidiary; <u>provided</u> that if the transferor in such a transaction is a wholly-owned Subsidiary, then the transferee must also be a wholly-owned Subsidiary, and, <u>provided further</u> that if the transferor of such assets is a Guarantor, the transferee thereof must either be Borrower or a Guarantor.
 - **7.05 Dispositions.** Make any Disposition or enter into any agreement to make any Disposition, except:
- (a) Dispositions of obsolete or worn out property, whether now owned or hereafter acquired, in the ordinary course of business;
 - (b) Dispositions of inventory in the ordinary course of business;
- (c) Dispositions of equipment or real property to the extent that (i) such property is exchanged for credit against the purchase price of similar replacement property, or (ii) the proceeds of such Disposition are reasonably promptly applied to the purchase price of such replacement property;
- (d) Dispositions of property by any Subsidiary to Borrower or to a wholly-owned Subsidiary, <u>provided</u> that if the transferor of such property is a Guarantor, the transferee thereof must either be Borrower or a Guarantor;
 - (e) Dispositions permitted by <u>Section 7.04</u>;
 - (f) non-exclusive licenses of IP Rights in the ordinary course of business and substantially consistent with past practice for

terms not exceeding five years; and

(g) Dispositions by Borrower and its Subsidiaries not otherwise permitted under this <u>Section 7.05</u>; <u>provided</u> that (i) at the time of such Disposition, no Default shall exist or would result from such Disposition and (ii) the aggregate book value of all property Disposed of in reliance on this clause (g) in any fiscal year shall not exceed \$500,000;

provided, however, that any Disposition pursuant to clauses (a) through (g) shall be for fair market value.

- **7.06 Restricted Payments.** Declare or make, directly or indirectly, any Restricted Payment, or incur any obligation (contingent or otherwise) to do so, except that:
- (a) each Subsidiary may make Restricted Payments to Borrower and to wholly-owned Subsidiaries (and, in the case of a Restricted Payment by a non-wholly-owned Subsidiary, to Borrower and any Subsidiary and to each other owner of capital stock or other equity interests of such Subsidiary on a pro rata basis based on their relative ownership interests);
- (b) Borrower and each Subsidiary may declare and make dividend payments or other distributions payable solely in the common stock or other common equity interests of such Person;
- (c) Borrower and each Subsidiary may purchase, redeem or otherwise acquire shares of its common stock or other common equity interests or warrants or options to acquire any such shares with the proceeds received from the substantially concurrent issue of new shares of its common stock or other common equity interests; and

32

- (d) Borrower may declare or pay cash dividends to its stockholders and purchase, redeem or otherwise acquire shares of its capital stock or warrants, rights or options to acquire any such shares for cash; <u>provided</u> that immediately after giving effect to such proposed action, no Default would exist.
- **7.07 Change in Nature of Business.** Engage in any material line of business substantially different from those lines of business conducted by Borrower and its Subsidiaries on the date hereof or any business substantially related or incidental thereto.
- **7.08 Transactions with Affiliates.** Enter into any transaction of any kind with any Affiliate of Borrower, whether or not in the ordinary course of business, other than on fair and reasonable terms substantially as favorable to Borrower or such Subsidiary as would be obtainable by Borrower or such Subsidiary at the time in a comparable arm's length transaction with a Person other than an Affiliate, <u>provided</u> that the foregoing restriction shall not apply to transactions between or among Borrower and any of its wholly-owned Subsidiaries or between and among any wholly-owned Subsidiaries.
- **7.09 Burdensome Agreements.** Enter into any Contractual Obligation (other than this Agreement or any other Loan Document) that (a) limits the ability (i) of any Subsidiary to make Restricted Payments to Borrower or any Guarantor or to otherwise transfer property to Borrower or any Guarantor, (ii) of any Subsidiary to Guarantee the Indebtedness of Borrower or (iii) of Borrower or any Subsidiary to create, incur, assume or suffer to exist Liens on property of such Person; <u>provided, however</u>, that this clause (iii) shall not prohibit any negative pledge incurred or provided in favor of any holder of Indebtedness permitted under <u>Section 7.03(f)</u> solely to the extent any such negative pledge relates to the property financed by or the subject of such Indebtedness; or (b) requires the grant of a Lien to secure an obligation of such Person if a Lien is granted to secure another obligation of such Person.
- **7.10** Use of Proceeds. Use the proceeds of any Credit Extension, whether directly or indirectly, and whether immediately, incidentally or ultimately, to purchase or carry margin stock (within the meaning of Regulation U of the Board of Governors of the Federal Reserve System of the United States) or to extend credit to others for the purpose of purchasing or carrying margin stock or to refund indebtedness originally incurred for such purpose.

ARTICLE VIII EVENTS OF DEFAULT AND REMEDIES

- **8.01 Events of Default.** Any of the following shall constitute an Event of Default:
- (a) Non-Payment. Borrower or any other Loan Party fails to pay (i) when and as required to be paid herein, any amount of principal of any Loan, or any L/C Obligation, or (ii) within three days after the same becomes due, any interest on any Loan or on any L/C Obligation, or any commitment or other fee due hereunder, or (iii) within five days after the same becomes due, any other amount payable hereunder or under any other Loan Document; or
- (b) Specific Covenants. Borrower fails to perform or observe any term, covenant or agreement contained in any of Sections 6.01, 6.02, 6.03, 6.05, 6.10, 6.11 or 6.12 or Article VII; or
- (c) Other Defaults. Any Loan Party fails to perform or observe any other covenant or agreement (not specified in subsection (a) or (b) above) contained in any Loan Document on its part to be performed or observed and such failure continues for 30 days or any default or event of default occurs under any other Loan Document; or
- (d) <u>Representations and Warranties</u>. Any representation, warranty, certification or statement of fact made or deemed made by or on behalf of Borrower or any other Loan Party herein, in any other Loan Document, or in any document delivered in connection herewith or therewith shall be incorrect or misleading in any material respect when made or deemed made; or

- (e) <u>Cross-Default.</u> (i) Borrower or any Subsidiary (A) fails to make any payment when due (whether by scheduled maturity, required prepayment, acceleration, demand, or otherwise) in respect of any Indebtedness or Guarantee (other than Indebtedness hereunder and Indebtedness under Swap Contracts) having an aggregate principal amount (including undrawn committed or available amounts and including amounts owing to all creditors under any combined or syndicated credit arrangement) of more than the Threshold Amount, or (B) fails to observe or perform any other agreement or condition relating to any such Indebtedness or Guarantee or contained in any instrument or agreement evidencing, securing or relating thereto, or any other event occurs, the effect of which default or other event is to cause, or to permit the holder or holders of such Indebtedness or the beneficiary or beneficiaries of such Guarantee (or a trustee or agent on behalf of such holder or holders or beneficiary or beneficiaries) to cause, with the giving of notice if required, such Indebtedness to be demanded or to become due or to be repurchased, prepaid, defeased or redeemed (automatically or otherwise), or an offer to repurchase, prepay, defease or redeem such Indebtedness to be made prior to its stated maturity, or such Guarantee to become payable or cash collateral in respect thereof to be demanded; or (ii) there occurs under any Swap Contract an Early Termination Date (as defined in such Swap Contract) resulting from (A) any event of default under such Swap Contract as to which Borrower or any Subsidiary is an Affected Party (as so defined) and, in either event, the Swap Termination Value owed by Borrower or such Subsidiary as a result thereof is greater than the Threshold Amount; or
- (f) <u>Insolvency Proceedings, Etc.</u> Any Loan Party or any of its Subsidiaries institutes or consents to the institution of any proceeding under any Debtor Relief Law, or makes an assignment for the benefit of creditors; or applies for or consents to the appointment of any receiver, trustee, custodian, conservator, liquidator, rehabilitator or similar officer for it or for all or any material part of its property; or any receiver, trustee, custodian, conservator, liquidator, rehabilitator or similar officer is appointed without the application or consent of such Person and the appointment continues undischarged or unstayed for 60 calendar days; or any proceeding under any Debtor Relief Law relating to any such Person or to all or any material part of its property is instituted without the consent of such Person and continues undismissed or unstayed for 60 calendar days, or an order for relief is entered in any such proceeding; or
- (g) <u>Inability to Pay Debts; Attachment.</u> (i) Borrower or any Subsidiary becomes unable or admits in writing its inability or fails generally to pay its debts as they become due, or (ii) any writ or warrant of attachment or execution or similar process is issued or levied against all or any material part of the property of any such Person and is not released, vacated or fully bonded within 30 days after its issue or levy; or
- (h) <u>Judgments</u>. There is entered against Borrower or any Subsidiary (i) a final judgment or order for the payment of money in an aggregate amount exceeding the Threshold Amount (to the extent not covered by independent third-party insurance as to which the insurer does not dispute coverage), or (ii) any one or more non-monetary final judgments that have, or could reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect and, in either case, (A) enforcement proceedings are commenced by any creditor upon such judgment or order, or (B) there is a period of 10 consecutive days during which a stay of enforcement of such judgment, by reason of a pending appeal or otherwise, is not in effect; or
- (i) ERISA. (i) An ERISA Event occurs with respect to a Pension Plan or Multiemployer Plan which has resulted or could reasonably be expected to result in liability of Borrower under Title IV of ERISA to the Pension Plan, Multiemployer Plan or the PBGC in an aggregate amount in excess of the Threshold Amount, or (ii) Borrower or any ERISA Affiliate fails to pay when due, after the expiration of any applicable grace period, any installment payment with respect to its withdrawal liability under Section 4201 of ERISA under a Multiemployer Plan in an aggregate amount in excess of the Threshold Amount; or
 - (j) <u>Change of Control</u>. There occurs any Change of Control or Change of Management; or

- (k) <u>Material Adverse Effect</u>. There occurs any event or circumstance that has a Material Adverse Effect; or
- (l) <u>Invalidity of Loan Documents</u>. Any Loan Document, at any time after its execution and delivery and for any reason other than as expressly permitted hereunder or satisfaction in full of all the Obligations, ceases to be in full force and effect; or any Loan Party or any other Person contests in any manner the validity or enforceability of any Loan Document; or any Loan Party denies that it has any or further liability or obligation under any Loan Document, or purports to revoke, terminate or rescind any Loan Document, or any Lien with respect to any material portion of the Collateral intended to be secured thereby ceases to be, or, subject to <u>Section 7.01</u>, is not, valid, perfected and prior to all other Liens or is terminated, revoked or declared void.
- **8.02** Remedies Upon Event of Default. If any Event of Default occurs and is continuing, Bank may take any or all of the following actions:
 - (a) declare the Commitment to be terminated, whereupon the Commitment shall be terminated;
- (b) declare the unpaid principal amount of all outstanding Loans, all interest accrued and unpaid thereon, and all other amounts owing or payable hereunder or under any other Loan Document to be immediately due and payable, without presentment, demand, protest or other notice of any kind, all of which are hereby expressly waived by Borrower;
- (c) require that Borrower Cash Collateralize the L/C Obligations (in an amount equal to the then Outstanding Amount thereof); and

(d) exercise all rights and remedies available to it under the Loan Documents or applicable law;

provided, however, that upon the occurrence of an actual or deemed entry of an order for relief with respect to Borrower under the Bankruptcy Code of the United States, the Commitment shall automatically terminate, the unpaid principal amount of all outstanding Loans and all interest and other amounts as aforesaid shall automatically become due and payable, and the obligation of Borrower to Cash Collateralize the L/C Obligations as aforesaid shall automatically become effective, in each case without further act of Bank.

Application of Funds. After the exercise of remedies provided for in Section 8.02 (or after the Loans have 8.03 automatically become immediately due and payable and the L/C Obligations have automatically been required to be Cash Collateralized as set forth in the proviso to Section 8.02), any amounts received (a) on account of the Obligations, (b) on account of any debt, liability, obligation, covenant or duty of Borrower or any Subsidiary owing to Bank or any Affiliate of Bank arising under any Swap Contract or Financial Transaction Contract (collectively, the "Contract Obligations") and/or (c) in respect of the Collateral, shall be applied by Bank to the Obligations and the Contract Obligations in such order as it elects in its sole discretion.

ARTICLE IX MISCELLANEOUS

9.01 Amendments; Etc. No amendment or waiver of any provision of this Agreement or any other Loan Document, and no consent to any departure by Borrower or any other Loan Party therefrom, shall be effective unless in writing signed by Bank and Borrower or the applicable Loan Party, as the case may be, and each such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given.

35

9.02 Notices and Other Communications; Facsimile Copies.

General. Unless otherwise expressly provided herein, all notices and other communications provided for hereunder shall be in writing (including by facsimile transmission). All such written notices shall be mailed, faxed or delivered, to the applicable address, facsimile number or (subject to subsection (c) below) electronic mail address, and all notices and other communications expressly permitted hereunder to be given by telephone shall be made to the applicable telephone number, specified for such Person below or to such other address, facsimile number, electronic mail address or telephone number as shall be designated by such party in a notice to the other parties.

Bank: Bank of America, N.A.

WA1-501-35-01

800 Fifth Avenue, Floor 35

Seattle, WA 98104

Attention: Mark N. Crawford Telephone: (206) 358-8945 Facsimile: (206) 585-8638

Electronic Mail:

mark.n.crawford@bankofamerica.com

Borrower: USANA Health Sciences, Inc

> 3838 West Parkway Boulevard Salt Lake City, Utah 84120 Attention: Gilbert A. Fuller Telephone: (801) 954-7951 Facsimile: (801) 956-9486

Electronic Mail: gil.fuller@us.usana.com

All such notices and other communications shall be deemed to be given or made upon the earlier to occur of (i) actual receipt by the relevant party hereto and (ii) (A) if delivered by hand or by courier, when signed for by or on behalf of the relevant party hereto; (B) if delivered by mail, four Business Days after deposit in the mails, postage prepaid; (C) if delivered by facsimile, when sent and receipt has been confirmed by telephone; and (D) if delivered by electronic mail (which form of delivery is subject to the provisions of subsection (c) below), when delivered; provided, however, that notices and other communications to Bank pursuant to Article II shall not be effective until actually received by Bank. In no event shall a voicemail message be effective as a notice, communication or confirmation hereunder.

- Effectiveness of Facsimile Documents and Signatures. Loan Documents may be transmitted and/or signed by facsimile. The effectiveness of any such documents and signatures shall, subject to applicable Law, have the same force and effect as manually-signed originals and shall be binding on all Loan Parties and Bank. Bank may also require that any such documents and signatures be confirmed by a manually-signed original thereof; provided, however, that the failure to request or deliver the same shall not limit the effectiveness of any facsimile document or signature.
- Limited Use of Electronic Mail. Electronic mail and internet and intranet websites may be used only to distribute routine communications, such as financial statements, and to distribute Loan Documents for execution by the parties thereto, and may not be used for any other purpose.
 - (d) Reliance by Bank. Bank shall be entitled to rely and act upon any notices (including telephonic Loan Notices)

purportedly given by or on behalf of Borrower even if (i) such notices were not made in a manner specified herein, were incomplete or were not preceded or followed by any other form of notice specified herein, or (ii) the terms thereof, as understood by the recipient, varied from any confirmation thereof. Borrower shall indemnify Bank its Affiliates, and their respective officers, directors, employees, agents and attorneys-in-fact from all losses, costs, expenses and liabilities resulting from the reliance by such Person on each notice purportedly given by or on behalf of Borrower. All telephonic notices to and other communications with Bank may be recorded by Bank, and Borrower hereby consents to such recording.

36

- **9.03 No Waiver; Cumulative Remedies.** No failure by Bank to exercise, and no delay by Bank in exercising, any right, remedy, power or privilege hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. The rights, remedies, powers and privileges herein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law.
- 9.04 Attorney Costs, Expenses and Taxes. Borrower agrees (a) to pay or reimburse Bank for all costs and expenses incurred in connection with the development, preparation, negotiation and execution of this Agreement and the other Loan Documents and any amendment, waiver, consent or other modification of the provisions hereof and thereof (whether or not the transactions contemplated hereby or thereby are consummated), and the consummation and administration of the transactions contemplated hereby and thereby, including all Attorney Costs, and (b) to pay or reimburse Bank for all costs and expenses incurred in connection with the enforcement, attempted enforcement, or preservation of any rights or remedies under this Agreement or the other Loan Documents (including all such costs and expenses incurred during any "workout" or restructuring in respect of the Obligations and during any legal proceeding, including any proceeding under any Debtor Relief Law), including all Attorney Costs. The foregoing costs and expenses shall include all search, filing, recording, title insurance and appraisal charges and fees and taxes related thereto, and other out-of-pocket expenses incurred by Bank and the cost of independent public accountants and other outside experts retained by Bank. All amounts due under this Section 9.04 shall be payable within ten Business Days after demand therefor. The agreements in this Section shall survive the termination of the Commitment and repayment, satisfaction or discharge of all other Obligations.
- Indemnification by Borrower. Whether or not the transactions contemplated hereby are consummated, Borrower shall indemnify and hold harmless Bank, its Affiliates, and their respective directors, officers, employees, counsel, agents and attorneysin-fact (collectively the "Indemnitees") from and against any and all liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses and disbursements (including Attorney Costs) of any kind or nature whatsoever which may at any time be imposed on, incurred by or asserted against any such Indemnitee in any way relating to or arising out of or in connection with (a) the execution, delivery, enforcement, performance or administration of any Loan Document or any other agreement, letter or instrument delivered in connection with the transactions contemplated thereby or the consummation of the transactions contemplated thereby, (b) the Commitment, any Loan or Letter of Credit or the use or proposed use of the proceeds therefrom (including any refusal by Bank to honor a demand for payment under a Letter of Credit if the documents presented in connection with such demand do not strictly comply with the terms of such Letter of Credit), (c) any actual or alleged presence or release of Hazardous Materials on or from any property currently or formerly owned or operated by Borrower, any Subsidiary or any other Loan Party, or any Environmental Liability related in any way to Borrower, any Subsidiary or any other Loan Party, or (d) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory (including any investigation of, preparation for, or defense of any pending or threatened claim, investigation, litigation or proceeding) and regardless of whether Indemnitee is a party thereto (all the foregoing, collectively, the "Indemnified Liabilities"), in all cases, whether or not caused by or arising, in whole or in part, out of the negligence of the Indemnitee; provided that such indemnity shall not, as to any Indemnitee, be available to the extent that such liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses or disbursements are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Indemnitee. No Indemnitee shall have any liability for any indirect or consequential damages relating to this Agreement or any other Loan Document or arising out of its activities in connection herewith or therewith (whether before or after the Closing Date). All amounts due under this Section 9.05 shall be payable within ten Business Days after demand therefor. The agreements in this Section shall survive the termination of the Commitment and the repayment, satisfaction or discharge of all the other Obligations.

- 9.06 Payments Set Aside. To the extent that any payment by or on behalf of Borrower is made to Bank, or Bank exercises its right of set-off, and such payment or the proceeds of such set-off or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by Bank in its discretion) to be repaid to a trustee, receiver or any other party, in connection with any proceeding under any Debtor Relief Law or otherwise, then to the extent of such recovery, the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such set-off had not occurred.
- 9.07 Successors and Assigns. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that Borrower may not assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of Bank. Without limiting the foregoing, Bank may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement (including under the Note) to secure obligations of Bank, including any pledge or assignment to secure obligations to a Federal Reserve Bank. Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person any legal or equitable right, remedy or claim under or by reason of this Agreement.

9.08 Confidentiality. Bank agrees to maintain the confidentiality of the Information (as defined below), except that Information may be disclosed (a) to its and its Affiliates' directors, officers, employees and agents, including accountants, legal counsel and other advisors (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential), (b) to the extent requested by any regulatory authority, (c) to the extent required by applicable laws or regulations or by any subpoena or similar legal process, (d) to any other party hereto, (e) in connection with the exercise of any remedies hereunder or under any other Loan Document or any action or proceeding relating to this Agreement or any other Loan Document or the enforcement of rights hereunder or thereunder, (f) subject to an agreement containing provisions substantially the same as those of this Section, to (i) any assignee of or participant in, or any prospective assignee of or participant in, any of its rights or obligations under this Agreement or (ii) any actual or prospective counterparty (or its advisors) to any swap or derivative transaction relating to Borrower and its obligations, (g) with the consent of Borrower or (h) to the extent such Information (x) becomes publicly available other than as a result of a breach of this Section or (y) becomes available to Bank on a nonconfidential basis from a source other than Borrower. For purposes of this Section, "Information" means all information received from any Loan Party relating to any Loan Party or any of their respective businesses, other than any such information that is available to Bank on a nonconfidential basis prior to disclosure by any Loan Party, provided that, in the case of information received from a Loan Party after the date hereof, such information is clearly identified at the time of delivery as confidential. Any Person required to maintain the confidentiality of Information as provided in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information.

Notwithstanding anything herein to the contrary, "Information" shall not include, and Borrower, the other Loan Parties, Bank and the respective Affiliates of each of the foregoing (and the respective partners, directors, officers, employees, agents, advisors and other representatives of each of the foregoing and their Affiliates) may disclose to any and all Persons, without limitation of any kind (a) any information with respect to the U.S. federal and state income tax treatment of the transactions contemplated hereby and any facts that may be relevant to understanding such tax treatment, which facts shall not include for this purpose the names of the parties or any other Person named herein, or information that would permit identification of the parties or such other Persons, or any pricing terms or other nonpublic business or financial information that is unrelated to such tax treatment or facts, and (b) all materials of any kind (including opinions or other tax analyses) relating to such tax treatment or facts that are provided to any of the Persons referred to above.

9.09 Set-off. In addition to any rights and remedies of Bank provided by law, upon the occurrence and during the continuance of any Event of Default, Bank is authorized at any time and from

38

time to time, without prior notice to Borrower or any other Loan Party, any such notice being waived by Borrower (on its own behalf and on behalf of each Loan Party) to the fullest extent permitted by law, to set off and apply any and all deposits (general or special, time or demand, provisional or final) at any time held by, and other indebtedness at any time owing by, Bank to or for the credit or the account of the respective Loan Parties against any and all Obligations owing to Bank hereunder or under any other Loan Document, now or hereafter existing, irrespective of whether or not Bank shall have made demand under this Agreement or any other Loan Document and although such Obligations may be contingent or unmatured or denominated in a currency different from that of the applicable deposit or indebtedness. Bank agrees promptly to notify Borrower after any such set-off and application; provided, however, that the failure to give such notice shall not affect the validity of such set-off and application.

- 9.10 Interest Rate Limitation. Notwithstanding anything to the contrary contained in any Loan Document, the interest paid or agreed to be paid under the Loan Documents shall not exceed the maximum rate of non-usurious interest permitted by applicable Law (the "Maximum Rate"). If Bank shall receive interest in an amount that exceeds the Maximum Rate, the excess interest shall be applied to the principal of the Loans or, if it exceeds such unpaid principal, refunded to Borrower. In determining whether the interest contracted for, charged, or received by Bank exceeds the Maximum Rate, Bank may, to the extent permitted by applicable Law, (a) characterize any payment that is not principal as an expense, fee, or premium rather than interest, (b) exclude voluntary prepayments and the effects thereof, and (c) amortize, prorate, allocate, and spread in equal or unequal parts the total amount of interest throughout the contemplated term of the Obligations hereunder.
- **9.11 Counterparts.** This Agreement may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- **9.12 Integration.** This Agreement, together with the other Loan Documents, comprises the complete and integrated agreement of the parties on the subject matter hereof and thereof and supersedes all prior agreements, written or oral, on such subject matter. In the event of any conflict between the provisions of this Agreement and those of any other Loan Document, the provisions of this Agreement shall control; <u>provided</u> that the inclusion of supplemental rights or remedies in favor of Bank in any other Loan Document shall not be deemed a conflict with this Agreement. Each Loan Document was drafted with the joint participation of the respective parties thereto and shall be construed neither against nor in favor of any party, but rather in accordance with the fair meaning thereof.
- 9.13 Survival of Representations and Warranties. All representations and warranties made hereunder and in any other Loan Document or other document delivered pursuant hereto or thereto or in connection herewith or therewith shall survive the execution and delivery hereof and thereof. Such representations and warranties have been or will be relied upon by Bank, regardless of any investigation made by Bank or on its behalf and notwithstanding that Bank may have had notice or knowledge of any Default at the time of any Credit Extension, and shall continue in full force and effect as long as any Loan or any other Obligation hereunder shall remain unpaid or unsatisfied or any Letter of Credit shall remain outstanding.
- **9.14 Severability.** If any provision of this Agreement or the other Loan Documents is held to be illegal, invalid or unenforceable, (a) the legality, validity and enforceability of the remaining provisions of this Agreement and the other Loan Documents

shall not be affected or impaired thereby and (b) the parties shall endeavor in good faith negotiations to replace the illegal, invalid or unenforceable provisions with valid provisions the economic effect of which comes as close as possible to that of the illegal, invalid or unenforceable provisions. The invalidity of a provision in a particular jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

9.15 Governing Law; Submission to Jurisdiction.

(a) THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF WASHINGTON APPLICABLE TO AGREEMENTS MADE AND

39

TO BE PERFORMED ENTIRELY WITHIN SUCH STATE; <u>PROVIDED</u> THAT BANK SHALL RETAIN ALL RIGHTS ARISING UNDER FEDERAL LAW.

- (b) ANY LEGAL ACTION OR PROCEEDING WITH RESPECT TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT MAY BE BROUGHT IN THE COURTS OF THE STATE OF WASHINGTON SITTING IN SEATTLE, KING COUNTY, WASHINGTON OR OF THE UNITED STATES FOR THE WESTERN DISTRICT OF SUCH STATE, AND BY EXECUTION AND DELIVERY OF THIS AGREEMENT, BORROWER AND BANK EACH CONSENTS, FOR ITSELF AND IN RESPECT OF ITS PROPERTY, TO THE NON-EXCLUSIVE JURISDICTION OF THOSE COURTS. BORROWER AND BANK EACH IRREVOCABLY WAIVES ANY OBJECTION, INCLUDING ANY OBJECTION TO THE LAYING OF VENUE OR BASED ON THE GROUNDS OF *FORUM NON CONVENIENS*, WHICH IT MAY NOW OR HEREAFTER HAVE TO THE BRINGING OF ANY ACTION OR PROCEEDING IN SUCH JURISDICTION IN RESPECT OF ANY LOAN DOCUMENT OR OTHER DOCUMENT RELATED THERETO. BORROWER AND BANK EACH WAIVES PERSONAL SERVICE OF ANY SUMMONS, COMPLAINT OR OTHER PROCESS, WHICH MAY BE MADE BY ANY OTHER MEANS PERMITTED BY THE LAW OF SUCH STATE.
- 9.16 Waiver of Right to Trial by Jury. EACH PARTY TO THIS AGREEMENT HEREBY EXPRESSLY WAIVES ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION OR CAUSE OF ACTION ARISING UNDER ANY LOAN DOCUMENT OR IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTIES HERETO OR ANY OF THEM WITH RESPECT TO ANY LOAN DOCUMENT, OR THE TRANSACTIONS RELATED THERETO, IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER FOUNDED IN CONTRACT OR TORT OR OTHERWISE; AND EACH PARTY HEREBY AGREES AND CONSENTS THAT, SUBJECT TO SECTION 9.17, ANY SUCH CLAIM, DEMAND, ACTION OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY, AND THAT ANY PARTY TO THIS AGREEMENT MAY FILE AN ORIGINAL COUNTERPART OR A COPY OF THIS SECTION WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE SIGNATORIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.

9.17 Mandatory Arbitration.

- (a) This Section concerns the resolution of any controversies or claims between Borrower and Bank, whether arising in contract, tort or by statute, that arise out of or relate to this Agreement and the other Loan Documents (collectively a "Claim"). At the request of Borrower or Bank, any Claim shall be resolved by arbitration in accordance with the Federal Arbitration Act (Title 9, United States Code) (the "Arbitration Act"). The Arbitration Act will apply even though this Agreement provides that it is governed by the Laws of the state of Washington. Nothing contained in this Section 9.17 shall override any contrary provision contained in any Swap Contract.
- (b) Arbitration proceedings will be determined in accordance with the Arbitration Act, the rules and procedures for the arbitration of financial services disputes of JAMS/ENDISPUTE, LLC, a Delaware limited liability company or any successor thereof ("JAMS"), and the terms of this Section 9.17. In the event of any inconsistency, the terms of this Section 9.17 shall control. The arbitration shall be administered by JAMS and conducted in Seattle, Washington. All Claims shall be determined by one arbitrator; provided, however, that if Claims exceed \$5,000,000, upon the request of any party, the Claims shall be decided by three arbitrators. All arbitration hearings shall commence within 90 days of the demand for arbitration and close within 90 days of commencement and the award of the arbitrator(s) shall be issued within 30 days of the close of the hearing; provided, however, that the arbitrator(s), upon a showing of good cause, may extend the commencement of the hearing for up to an additional 60 days. The arbitrator(s) shall provide a concise written statement of reasons for the award. The arbitration award may be submitted to any court having jurisdiction to be confirmed and enforced. The arbitrator(s) will have the authority to decide whether any Claim is barred by the statute of limitations and, if so, to dismiss the arbitration on that basis. For purposes of the application of the statute of limitations, the service on JAMS under applicable JAMS rules of a notice of Claim is the equivalent of the filing of a lawsuit. Any dispute concerning this arbitration provision or whether a Claim is arbitrable shall be determined by the

40

arbitrator(s). The arbitrator(s) shall have the power to award legal fees pursuant to the terms of this Agreement.

(c) This Section 9.17 does not limit the right of Borrower or Bank to: (i) exercise self-help remedies, such as but not limited to, setoff; (ii) initiate judicial or nonjudicial foreclosure against any real or personal property collateral; (iii) exercise any judicial or power of sale rights; or (iv) act in a court of law to obtain an interim remedy, such as but not limited to, injunctive relief, writ of possession or appointment of a receiver, or additional or supplementary remedies. The filing of a court action is not intended to constitute

a waiver of the right of Borrower or Bank, including the suing party, thereafter to require submittal of the Claim to arbitration.

- **9.18 USA Patriot Act Notice.** Bank hereby notifies Borrower that pursuant to the requirements of the USA Patriot Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)) (the "<u>Patriot Act</u>"), it is required to obtain, verify and record information that identifies Borrower, which information includes the name and address of Borrower and other information that will allow Bank to identify Borrower in accordance with the Patriot Act.
 - **9.19** Time of the Essence. Time is of the essence of the Loan Documents.

9.20 Oral Agreements. <u>ORAL AGREEMENTS OR ORAL COMMITMENTS TO LOAN MONEY, EXTEND CREDIT OR TO FORBEAR FROM ENFORCING REPAYMENT OF A DEBT ARE NOT ENFORCEABLE UNDER WASHINGTON LAW.</u>

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first above written.

USANA HEALTH SCIENCES, INC.

By: /s/ Gilbert A. Fuller

Name: Gilbert A. Fuller

Title: Sr. Vice Pres. and CFO

BANK OF AMERICA, N.A.

By: /s/ Mark N. Crawford
Name: Mark N. Crawford
Title: Senior Vice President

SCHEDULE 5.06

LITIGATION

Creagri Inc., a California corporation ("<u>Creagri</u>"), has initiated a lawsuit claiming that Borrower's use of Borrower's "Olivol" trademark has infringed upon Creagri's use of "Olivenol" which is the name of a stand-alone olive by-product supplement. Borrower was granted a trademark for "Olivol" and the stylized trademark for "Olivol" by the United States Patent and Trademark office.

At this time Borrower believes that there is no merit to this case, that we shall prevail in this matter and that it does not pose any Material Adverse Effect.

SCHEDULE 5.09

ENVIRONMENTAL MATTERS

On November 19, 1997 a traffic accident occurred in the intersection of Parkway Blvd and Bangeter Highway. As a result of the accident a Tractor-Trailer came to rest on the lawn in front of our building. The Tractor-Trailer diesel tank was leaking.

Borrower estimates that 5 to 10 gallons of diesel fuel spilled on the lawn. Borrower hired Glover Nursery to remove the contaminated grass and soil. Glover Nursery removed and replaced approximately 4,000 cubic feet of topsoil.

Borrower believes that it took all of the reasonable necessary to remove the diesel fuel spilled on its property.

SCHEDULE 5.13

SUBSIDIARIES AND OTHER EQUITY INVESTMENTS

Part (a). <u>Subsidiaries</u>.

Name of Owner Subsidiary Formation Delaware 100%

USANA Health Sciences, Inc. USANA Canada Holding, Inc Delaware 100%

USANA Health Sciences, Inc.	USANA Health Sciences New Zealand, Inc.	Delaware	100 %
USANA Health Sciences, Inc.	USANA Australia Pty Ltd	New South Wales	100 %
USANA Health Sciences, Inc.	USANA Hong Kong Limited	Hong Kong	99.9 %
USANA Health Sciences, Inc.	USANA Health Sciences Singapore Pte. Ltd.	Singapore	100 %
USANA Health Sciences, Inc.	USANA Health Sciences Korea	Korea	100 %
USANA Health Sciences, Inc.	USANA México, S.A. de C.V.	Mexico	99.8 %
USANA Health Sciences, Inc.	FMG Productions, Inc.	Utah	100 %
USANA Health Sciences, Inc.	USANA Japan, Inc.	Japan	100 %
USANA Health Sciences, Inc.	International Holdings, Inc.	Delaware	100 %
USANA Health Sciences, Inc.	USANA Acquisition Corp.	Utah	100 %
USANA Canada Holding, Inc	USANA Canada Company	Nova Scotia	100 %
USANA Health Sciences New Zealand, Inc.	USANA Health Sciences (NZ) Corporation	New Zealand	100 %
International Holdings, Inc.	Mercadotecnia Nutricional S de R.L. de C.v.	Mexico	99.9 %
USANA Acquisition Corp.	Wasatch Product Development, Inc.	Utah	100 %
Other Equity Investments.			

Part (b). Other Equity Investments.

Name of Owner	Name of Investment	Jurisdiction of Formation	Percentage
_		_	
-		-	
-		-	

SCHEDULE 5.18

INTELLECTUAL PROPERTY MATTERS

Creagri Inc., a California corporation ("<u>Creagri</u>"), has initiated a lawsuit claiming that Borrower's use of Borrower's "Olivol" trademark has infringed upon Creagri's use of "Olivenol" which is the name of a stand-alone olive by-product supplement. Borrower was granted a trademark for "Olivol" and the stylized trademark for "Olivol" by the United States Patent and Trademark office.

At this time Borrower believes that there is no merit to this case, that Borrower shall prevail in this matter and that it does not pose any Material Adverse Effect.

SCHEDULE 7.01

EXISTING LIENS

Secured Party	UCC Filing Number	Expiration Date	Amount	Collateral
Revco	34991200134	01/08/2006		Sharp Copier, Sharp Paper Desk Sharp Cassette Module, Sharp Large Tray, Sharp Finisher
EMC Corporation	49748200137	06/20/2006		Computer Disk arrays
De Lage Landen Financial Services, Inc	211691200330	03/11/2008		Symmetrix 8530-73 (12) 8030-181M1 (1) and other various Computer Disk arrays

Henrik	ssen/Butler Design Group	243341200434	4/26/2009	\$17,977	Office Furniture	
		E	XISTING INDEBT	EDNESS		SCHEDULE 7.03
None.						
						EXHIBIT A
			FORM OF LOAN N	NOTICE		
Date:	,					
То:	Bank of America, N.A.					
	and Cantlanan					
Ladies	and Gentlemen:					
otherw	Reference is made to that vise modified in writing from en USANA Health Sciences	n time to time, the " <u>(</u>	Credit Agreement;" tl	ne terms defined t	herein being used herein	as therein defined),
otherw	Reference is made to that vise modified in writing from	n time to time, the " <u>Control</u> , Inc., a Utah corpora	Credit Agreement;" tlation ("Borrower"), a	ne terms defined t	herein being used herein	as therein defined),
otherw	Reference is made to that vise modified in writing from the USANA Health Sciences	n time to time, the "g, Inc., a Utah corporarequests (select one):	Credit Agreement;" tlation ("Borrower"), a	ne terms defined to and Bank of Amer	herein being used herein ica, N.A., a national ban	as therein defined),
otherw	Reference is made to that vise modified in writing from en USANA Health Sciences The undersigned hereby n	n time to time, the "g, Inc., a Utah corporarequests (select one):	Credit Agreement;" tl ation (" <u>Borrower</u> "), a :	ne terms defined to and Bank of Amer ntinuation of Loan	herein being used herein ica, N.A., a national ban	as therein defined),
otherw betwee	Reference is made to that vise modified in writing from the USANA Health Sciences The undersigned hereby the A Borrowing of th	n time to time, the "g, Inc., a Utah corporarequests (select one):	Credit Agreement;" thation ("Borrower"), and its action ("Borrower"), and its action of the Aconversion or conversion or convers	ne terms defined to and Bank of Amer ntinuation of Loan	herein being used herein ica, N.A., a national ban	as therein defined),
otherw between	Reference is made to that vise modified in writing from the USANA Health Sciences The undersigned hereby the A Borrowing of the On	n time to time, the "g, Inc., a Utah corporarequests (select one):	Credit Agreement;" thation ("Borrower"), and its action ("Borrower"), and its action of the Aconversion or conversion or convers	ne terms defined to and Bank of Amer ntinuation of Loan	herein being used herein ica, N.A., a national ban	as therein defined),
otherw between	Reference is made to that vise modified in writing from the USANA Health Sciences The undersigned hereby the A Borrowing of On In the amount of \$	n time to time, the "g, Inc., a Utah corporarequests (select one):	Credit Agreement;" thation ("Borrower"), and its action ("Borrower"), and its action of the Aconversion or control (a Business Day).	ne terms defined to and Bank of Amer ntinuation of Loan	herein being used herein ica, N.A., a national ban	as therein defined),
otherw between	Reference is made to that vise modified in writing from the USANA Health Sciences The undersigned hereby the A Borrowing of On In the amount of \$	n time to time, the "g, Inc., a Utah corporatequests (select one): f Loans • [Type of Loan requests time to time, the "general time to time, the "general time".	Credit Agreement;" thation ("Borrower"), and its ation ("Borrower"), and its ation (a Business Day of the conversion or conversi	ne terms defined to and Bank of Amer ntinuation of Loan	herein being used herein ica, N.A., a national ban	as therein defined),
otherwed 1. 2.	Reference is made to that vise modified in writing from the USANA Health Sciences The undersigned hereby to the A Borrowing of	n time to time, the "g, Inc., a Utah corporate requests (select one): f Loans [Type of Loan requests: with an Interest	Credit Agreement;" thation ("Borrower"), and its ation ("Borrower"), and its ation (and its ation or continuous continuou	ne terms defined the terms defined the terms defined the terms of Amer notinuation of Loan).	herein being used herein ica, N.A., a national ban ns	as therein defined), king association.
otherwed 1. 2.	Reference is made to that vise modified in writing from the USANA Health Sciences The undersigned hereby in the A Borrowing of the On the A Borrowing of the Comprised of the Eurodollar Rate Leave to the A Borrowing of the	n time to time, the "g, Inc., a Utah corporate requests (select one): f Loans [Type of Loan requests: with an Interest	Credit Agreement;" thation ("Borrower"), and its ation ("Borrower"), and its ation (and its ation or continuous continuou	ne terms defined the terms def	herein being used herein ica, N.A., a national ban ns	as therein defined), king association.

EXHIBIT B

NOTE

\$10,000,000 June 16, 2004

FOR VALUE RECEIVED, the undersigned (the "Borrower") hereby promises to pay to the order of BANK OF AMERICA, N.A. (the "Bank"), on the Maturity Date (as defined in the Credit Agreement referred to below) the principal amount of Ten Million Dollars (\$10,000,000), or such lesser principal amount of Loans (as defined in such Credit Agreement) due and payable by Borrower to Bank on the Maturity Date under that certain Credit Agreement, dated as of June 16, 2004 (as amended, restated, extended, supplemented or otherwise modified in writing from time to time, the "Credit Agreement," the terms defined therein being used herein as therein defined), between Borrower and Bank.

Borrower promises to pay interest on the unpaid principal amount of each Loan from the date of such Loan until such principal amount is paid in full, at such interest rates and at such times as are specified in the Credit Agreement. All payments of principal and interest shall be made to Bank in Dollars in immediately available funds at Bank's Lending Office. If any amount is not paid in full when

due hereunder, such unpaid amount shall bear interest, to be paid upon demand, from the due date thereof until the date of actual payment (and before as well as after judgment) computed at the per annum rate set forth in the Credit Agreement.

This Note is the Note referred to in the Credit Agreement, is entitled to the benefits thereof and is subject to optional prepayment in whole or in part as provided therein. This Note is also entitled to the benefits of the Guaranty and is secured by the Collateral. Upon the occurrence and continuation of one or more of the Events of Default specified in the Credit Agreement, all amounts then remaining unpaid on this Note shall become, or may be declared to be, immediately due and payable all as provided in the Credit Agreement. Loans made by Bank shall be evidenced by one or more loan accounts or records maintained by Bank in the ordinary course of business. Bank may also attach schedules to this Note and endorse thereon the date, amount and maturity of the Loans and payments with respect thereto.

Borrower, for itself, its successors and assigns, hereby waives diligence, presentment, protest and demand and notice of protest, demand, dishonor and non-payment of this Note.

THIS NOTE SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF WASHINGTON.

USANA HEALTH SCIENCES, INC.

A mount of

By: /s/ Gilbert A. Fuller

Name: Gilbert A. Fuller

Title: Sr. Vice Pres. and CFO

REVOLVING LOANS AND PAYMENTS WITH RESPECT THERETO

				Amount or		
				Principal or	Outstanding	
			End of	Interest	Principal	
	Type of	Amount of	Interest	Paid This	Balance	Notation
Date	Loan Made	Loan Made	Period	Date	This Date	Made By
Date	• •		Interest	Paid This	Balance	

EXHIBIT C

CONTINUING GUARANTY

This CONTINUING GUARANTY ("<u>Guaranty</u>") is entered into as of June 16, 2004, by USANA ACQUISITION CORP., a Utah corporation ("<u>USANA Acquisition</u>"), WASATCH PRODUCT DEVELOPMENT, INC., a Utah corporation ("<u>Wasatch</u>"), USANA HEALTH SCIENCES NEW ZEALAND, INC., a Delaware corporation ("<u>USANA New Zealand</u>"), USANA CANADA HOLDING, INC., a Delaware corporation ("<u>USANA Canada</u>"), FMG PRODUCTIONS, INC., a Utah corporation ("<u>FMG</u>"), INTERNATIONAL HOLDINGS, INC., a Delaware corporation ("<u>Holdings</u>" and together with USANA Acquisition, Wasatch, USANA New Zealand, USANA Canada, FMG and any additional guarantors becoming a party hereto as provided in <u>Section 18</u> hereof and their respective successors, collectively, the "<u>Guarantors</u>" and individually, a "<u>Guarantor</u>"), in favor of BANK OF AMERICA, N.A., a national banking association (together with its successors and assigns, "<u>Bank</u>"), and any Affiliate of Bank a party to a Swap Contract (as defined in the Credit Agreement referred to below).

RECITALS

- A. USANA Health Sciences, Inc., a Utah corporation ("<u>Borrower</u>") is a party to that certain Credit Agreement dated as of June 16, 2004 by and between Borrower and Bank (as amended, restated, modified, renewed, supplemented or extended from time to time, the "<u>Credit Agreement</u>").
- B. It is a condition precedent to Bank's obligation to make its initial Credit Extension under the Credit Agreement that the Guarantors enter into this Guaranty.
 - C. Each Guarantor, as a direct or indirect wholly-owned Subsidiary of Borrower, will derive substantial and direct benefits

(which benefits are hereby acknowledged by each Guarantor) from the Loans and the Letters of Credit and other benefits to be provided to Borrower under the Credit Agreement.

NOW, THEREFORE, in consideration of the foregoing and in order to induce Bank to make Credit Extensions under the Credit Agreement, each Guarantor hereby agrees as follows:

- 1. **Definitions; Interpretation.** All capitalized terms used in this Guaranty and not otherwise defined herein have the meanings specified in the Credit Agreement. The rules of construction and interpretation specified in <u>Sections 1.02</u> and <u>1.05</u> of the Credit Agreement also apply to this Guaranty and are incorporated herein by this reference.
- **2. Guaranty.** Each Guarantor hereby irrevocably, absolutely and unconditionally guarantees, jointly with the other Guarantors and severally, as a primary obligor and not merely as a surety, the full and punctual payment or performance when due, whether at stated maturity, by required prepayment, declaration, acceleration, demand or otherwise, and at all times thereafter, all of the following debts, liabilities and obligations (collectively, the "<u>Guaranteed Obligations</u>"): (a) all advances to, and debts, liabilities, obligations, covenants and duties of, Borrower owing to Bank arising under the Credit Agreement and each other Loan Document or otherwise with respect to any Loan or Letter of Credit; (b) all debts, liabilities, obligations, covenants and duties of, Borrower or any of its Subsidiaries owing to Bank or any Affiliate of Bank and arising under any Swap Contract made or entered into at any time, or in effect at any time, whether directly or indirectly, and whether as a result of assignment or transfer or otherwise, between Bank or any Affiliate of Bank and Borrower or any Subsidiary of Borrower, including liabilities and obligations arising in connection with or as a result of early termination of any such Swap Contract; and (c) any and all fees, costs or out-of-pocket expenses (including Attorney Costs) incurred by Bank in enforcing any rights under this Guaranty and the other Loan Documents, in each case whether direct or indirect (including those acquired by assumption), absolute or contingent, due or to become due, now existing or hereafter arising and including interest and fees that accrue after the commencement by or against any Loan Party or any Affiliate thereof of any proceeding under any Debtor Relief Laws naming such Person as the debtor in such proceeding, regardless of whether such interest

and fees are allowed claims in such proceeding. The Bank's books and records showing the amount of the Guaranteed Obligations shall be admissible in evidence in any action or proceeding, and shall be binding upon the Guarantor and conclusive for the purpose of establishing the amount of the Guaranteed Obligations in the absence of manifest error. This Guaranty shall not be affected by the genuineness, validity, regularity or enforceability of the Guaranteed Obligations or any instrument or agreement evidencing any Guaranteed Obligations, or by the existence, validity, enforceability, perfection, or extent of any collateral therefor, or by any fact or circumstance relating to the Guaranteed Obligations which might otherwise constitute a defense to the obligations of the Guarantor under this Guaranty.

- 3. Limitation of Liability. Notwithstanding anything in this Guaranty to the contrary, the obligations of each Guarantor hereunder shall be limited to an aggregate amount equal to the largest amount that would not render its obligations hereunder subject to avoidance under Section 548 of the Bankruptcy Code (Title 11, United States Code) or any comparable provisions of any applicable state law.
- 4. No Setoff or Deductions; Taxes. Each Guarantor represents and warrants that it is incorporated and resident in the United States of America. All payments by each Guarantor hereunder shall be paid in full, without setoff or counterclaim or any deduction or withholding whatsoever, including for any and all present and future taxes. If one or more Guarantors must make a payment under this Guaranty (each, a "Paying Guarantor"), each such Paying Guarantor represents and warrants that it will make the payment from one of its U.S. resident offices to Bank so that no withholding tax is imposed on the payment. If notwithstanding the foregoing, a Paying Guarantor makes a payment under this Guaranty to which withholding tax applies, or any taxes (other than taxes on net income (a) imposed by the country or any subdivision of the country in which Bank's Lending Office is located and (b) measured by the United States taxable income Bank would have received if all payments under or in respect of this Guaranty were exempt from taxes levied by such Paying Guarantor's country) are at any time imposed on any payments under or in respect of this Guaranty including payments made pursuant to this Section 4, such Paying Guarantor shall pay all such taxes to the relevant authority in accordance with applicable Law such that Bank receives the sum it would have received had no such deduction or withholding been made and shall also pay to Bank, on demand, all additional amounts which Bank specifies as necessary to preserve the after-tax yield Bank would have received if such taxes had not been imposed. Each Paying Guarantor that makes a payment under this Guaranty to which withholding tax applies or to which any taxes are imposed on such payment, such Paying Guarantor shall promptly provide Bank with an original receipt or certified copy issued by the relevant authority evidencing the payment of any such amount required to be deducted or withheld.
- 5. No Termination. This Guaranty is a continuing and irrevocable guaranty of all Guaranteed Obligations now or hereafter existing and shall remain in full force and effect until all Guaranteed Obligations and any other amounts payable under this Guaranty are indefeasibly paid and performed in full and any commitments of Bank (including the Commitment) or facilities provided by Bank with respect to the Guaranteed Obligations are terminated. At Bank's option, all payments under this Guaranty shall be made to an office of Bank located in the United States and in U.S. Dollars.
- **6. Waiver of Notices.** Each Guarantor waives notice of the acceptance of this Guaranty and of the extension or continuation of the Guaranteed Obligations or any part thereof. Each Guarantor further waives presentment, protest, notice, dishonor or default, demand for payment and any other notices to which such Guarantor might otherwise be entitled.
- 7. **Subrogation.** No Guarantor shall exercise any right of subrogation, contribution or similar rights with respect to any payments it makes under this Guaranty until all of the Guaranteed Obligations and any amounts payable under this Guaranty are indefeasibly paid and performed in full and any commitments of the Bank (including the Commitment) or facilities provided by Bank with respect to the Guaranteed Obligations are terminated. If any amounts are paid to any Guarantor in violation of the foregoing limitation, then such amounts shall be held in trust for the benefit of Bank and shall forthwith be paid to Bank to reduce the amount of the

- **8.** Waiver of Suretyship Defenses. Each Guarantor agrees that Bank may, at any time and from time to time, and without notice to any Guarantor, make any agreement with Borrower or with any other Person liable on any of the Guaranteed Obligations or providing collateral as security for the Guaranteed Obligations, for the extension, renewal, payment, compromise, discharge or release of the Guaranteed Obligations or any collateral (in whole or in part), or for any modification or amendment of the terms thereof or of any instrument or agreement evidencing the Guaranteed Obligations or the provision of collateral, all without in any way impairing, releasing, discharging or otherwise affecting the obligations of any Guarantor under this Guaranty. Each Guarantor waives any defense arising by reason of any disability or other defense of Borrower or any other guarantor (including any Guarantor), or the cessation from any cause whatsoever of the liability of Borrower, or any claim that such Guarantor's obligations exceed or are more burdensome than those of Borrower and waives the benefit of any statute of limitations affecting the liability of such Guarantor hereunder. Each Guarantor waives any right to enforce any remedy which Bank now has or may hereafter have against Borrower and waives any benefit of and any right to participate in any security now or hereafter held by Bank. Further, each Guarantor consents to the taking of, or failure to take, any action which might in any manner or to any extent vary the risks of such Guarantor under this Guaranty or which, but for this provision, might operate as a discharge of such Guarantor.
- **9. Exhaustion of Other Remedies Not Required.** The obligations of each Guarantor hereunder are those of primary obligor, and not merely as surety, and are independent of the Guaranteed Obligations. Each Guarantor waives diligence by Bank and action on delinquency in respect of the Guaranteed Obligations or any part thereof, including any provisions of Law requiring Bank to exhaust any right or remedy or to take any action against Borrower, any other guarantor (including any Guarantor) or any other Person or property before enforcing this Guaranty against such Guarantor.
- **10. Reinstatement.** Notwithstanding anything in this Guaranty to the contrary, this Guaranty shall continue to be effective or be reinstated, as the case may be, if at any time any payment of any portion of the Guaranteed Obligations is revoked, terminated, rescinded or reduced or must otherwise be restored or returned upon the insolvency, bankruptcy or reorganization of Borrower, any guarantor (including any Guarantor) or any other Person, or otherwise, as if such payment had not been made and whether or not Bank is in possession of or has released this Guaranty and regardless of any prior revocation, rescission, termination or reduction.
- 11. Subordination. Each Guarantor hereby subordinates the payment of all obligations and indebtedness of Borrower owing to such Guarantor, whether now existing or hereafter arising, including any obligation of Borrower to such Guarantor as subrogee of Bank or resulting from such Guarantor's performance under this Guaranty, to the indefeasible payment in full of all Guaranteed Obligations. If Bank so requests, any such obligation or indebtedness of Borrower to such Guarantor shall be enforced and performance received by such Guarantor as trustee for Bank and the proceeds thereof shall be paid over to Bank on account of the Guaranteed Obligations, but without reducing or affecting in any manner the liability of such Guarantor under this Guaranty.
- 12. Information. Each Guarantor agrees to furnish promptly to Bank any and all financial or other information regarding such Guarantor or its property as Bank may reasonably request in writing.
- 13. Stay of Acceleration. In the event that acceleration of the time for payment of any of the Guaranteed Obligations is stayed, upon the insolvency, bankruptcy or reorganization of Borrower or any other Person, or otherwise, all such amounts shall nonetheless be payable by Guarantors immediately upon demand by Bank.
- 14. Expenses. Each Guarantor shall pay on demand any and all fees, costs or out-of-pocket expenses (including Attorney Costs) incurred by Bank in enforcing or protecting any of Bank's rights under this Guaranty, including any incurred in the preservation, protection or enforcement of any rights of Bank in any case commenced by or against any Guarantor under the Bankruptcy Code (Title 11, United States Code) or any similar or successor statute. The obligations of each Guarantor under the preceding sentence shall survive termination of this Guaranty.

- 15. Counterparts. This Guaranty may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 16. Integration. This Guaranty, together with the other Loan Documents to which any Guarantor is a party, comprises the complete and integrated agreement of the parties on the subject matter hereof and thereof and supersedes all prior agreements, written or oral, on such subject matter. Each Guarantor acknowledges that this Guaranty and the other Loan Documents may contain covenants and other terms and provisions variously stated regarding the same or similar matters, and agrees that all such covenants, terms and provisions are cumulative and all shall be performed and satisfied in accordance with their respective terms. The are no unwritten oral agreements between any Guarantor and Bank or any of them.
- 17. Amendments. No provision of this Guaranty may be waived, amended, supplemented or modified, except by a written instrument executed by Bank and each Guarantor.
- **18.** Additional Guarantors. Pursuant to Section 6.13(a) of the Credit Agreement, each Material Subsidiary that is a Domestic Subsidiary that was not in existence or was not a Guarantor on the date of the Credit Agreement is required to enter into this Guaranty as a Guarantor upon becoming a Material Subsidiary. Upon the execution and delivery by such Domestic Subsidiary of an

instrument in the form of Annex 1 hereto and acceptance thereof by Bank, such Domestic Subsidiary shall become a Guarantor hereunder with the same force and effect as if originally named as a Guarantor herein. The execution and delivery of any such instrument shall not require the consent of any other Guarantor hereunder. The rights and obligations of each Guarantor hereunder shall remain in full force and effect notwithstanding the addition of any new Guarantor as a party to this Guaranty.

19. Notices. Unless otherwise expressly provided herein, all notices and other communications provided for hereunder shall be in writing (including by facsimile transmission). All such written notices shall be mailed, faxed or delivered, to the address or facsimile number specified for such Person below or to such other address, facsimile number, electronic mail address or telephone number as shall be designated by such party in a notice to the other parties.

Bank: Bank of America, N.A.

WA1-501-35-01

800 Fifth Avenue, Floor 35

Seattle, WA 98104

Attention: Mark N. Crawford Facsimile: (206) 585-8638

Guarantors: c/o USANA Health Sciences, Inc

3838 West Parkway Boulevard Salt Lake City, Utah 84120 Attention: Gilbert A. Fuller Facsimile: (801) 956-9486

All such notices and other communications shall be deemed to be given or made upon the earlier to occur of (a) actual receipt by the relevant party hereto and (b) (i) if delivered by hand or by courier, when signed for by or on behalf of the relevant party hereto; (ii) if delivered by mail, four Business Days after deposit in the mails, postage prepaid; and (iii) if delivered by facsimile, when sent and receipt has been confirmed by telephone. In no event shall a voicemail message be effective as a notice, communication or confirmation hereunder.

20. No Waiver; Enforceability. No failure by Bank to exercise, and no delay in exercising, any right, remedy or power hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy or power hereunder preclude any other or further exercise thereof or the exercise of any other right. The remedies herein provided are cumulative and not exclusive of any remedies provided by law or in equity. The unenforceability or invalidity of any provision of this Guaranty

4

shall not affect the enforceability or validity of any other provision herein. The obligations hereunder shall not be affected, limited or impaired by any acts of any legislative body or governmental authority affecting Borrower, including any restrictions on or regarding the conversion of currency or repatriation or control of funds or any total or partial expropriation of Borrower's property, or by any economic, political, regulatory or other events in the countries where Borrower is located.

21. Assignment. This Guaranty shall (a) bind each Guarantor and their respective successors and assigns, <u>provided</u> that no Guarantor may assign its rights or obligations under this Guaranty without the prior written consent of Bank (and any attempted assignment without such consent shall be void) and (b) inure to the benefit of Bank and any Affiliate of Bank a party to a Swap Contract and their respective successors and assigns and Bank and any Affiliate of Bank a party to a Swap Contract may, without notice to any Guarantor and without affecting any Guarantor's obligations hereunder, assign or sell participations in the Guaranteed Obligations and this Guaranty, in whole or in part. Each Guarantor agrees that Bank and any Affiliate of Bank a party to a Swap Contract may disclose to any prospective purchaser and any purchaser of all or part of the Guaranteed Obligations any and all information in Bank's possession concerning any Guarantor, this Guaranty and any security for this Guaranty.

22. Governing Law; Submission to Jurisdiction.

- (a) THIS GUARANTY SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF WASHINGTON APPLICABLE TO AGREEMENTS MADE AND TO BE PERFORMED ENTIRELY WITHIN SUCH STATE; <u>PROVIDED</u> THAT BANK SHALL RETAIN ALL RIGHTS ARISING UNDER FEDERAL LAW.
- (b) ANY LEGAL ACTION OR PROCEEDING WITH RESPECT TO THIS GUARANTY OR ANY OTHER LOAN DOCUMENT MAY BE BROUGHT IN THE COURTS OF THE STATE OF WASHINGTON SITTING IN SEATTLE, KING COUNTY, WASHINGTON OR OF THE UNITED STATES FOR THE WESTERN DISTRICT OF SUCH STATE, AND BY EXECUTION AND DELIVERY OF THIS GUARANTY, EACH GUARANTOR CONSENTS, FOR ITSELF AND IN RESPECT OF ITS PROPERTY, TO THE NON-EXCLUSIVE JURISDICTION OF THOSE COURTS. EACH GUARANTOR IRREVOCABLY WAIVES ANY OBJECTION, INCLUDING ANY OBJECTION TO THE LAYING OF VENUE OR BASED ON THE GROUNDS OF FORUM NON CONVENIENS, WHICH IT MAY NOW OR HEREAFTER HAVE TO THE BRINGING OF ANY ACTION OR PROCEEDING IN SUCH JURISDICTION IN RESPECT OF THIS GUARANTY, ANY OTHER LOAN DOCUMENT OR OTHER DOCUMENT RELATED THERETO. EACH GUARANTOR WAIVES PERSONAL SERVICE OF ANY SUMMONS, COMPLAINT OR OTHER PROCESS, WHICH MAY BE MADE BY ANY OTHER MEANS PERMITTED BY THE LAW OF SUCH STATE.
- 23. Waiver of Right to Trial by Jury. EACH GUARANTOR HEREBY EXPRESSLY WAIVES ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION OR CAUSE OF ACTION ARISING UNDER THIS GUARANTY, ANY OTHER LOAN DOCUMENT OR IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF

ANY GUARANTOR AND BANK OR ANY OF THEM WITH RESPECT TO THIS GUARANTY, ANY OTHER LOAN DOCUMENT OR THE TRANSACTIONS RELATED THERETO, IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER FOUNDED IN CONTRACT OR TORT OR OTHERWISE; AND EACH GUARANTOR HEREBY AGREES AND CONSENTS THAT, SUBJECT TO SECTION 24, ANY SUCH CLAIM, DEMAND, ACTION OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY, AND THAT ANY PARTY TO THIS GUARANTY MAY FILE AN ORIGINAL COUNTERPART OR A COPY OF THIS SECTION WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE SIGNATORIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.

5

24. Mandatory Arbitration.

- (a) This Section concerns the resolution of any controversies or claims between any Guarantor and Bank or any of them, whether arising in contract, tort or by statute, that arise out of or relate to this Guaranty and the other Loan Documents (collectively a "Claim"). At the request of any Guarantor or Bank, any Claim shall be resolved by arbitration in accordance with the Federal Arbitration Act (Title 9, United States Code) (the "Arbitration Act"). The Arbitration Act will apply even though this Guaranty provides that it is governed by the Laws of the state of Washington. Nothing contained in this Section 24 shall override any contrary provision contained in any Swap Contract.
- (b) Arbitration proceedings will be determined in accordance with the Arbitration Act, the rules and procedures for the arbitration of financial services disputes of JAMS/ENDISPUTE, LLC, a Delaware limited liability company or any successor thereof ("JAMS"), and the terms of this Section 24. In the event of any inconsistency, the terms of this Section 24 shall control. The arbitration shall be administered by JAMS and conducted in Seattle, Washington. All Claims shall be determined by one arbitrator; provided, however, that if Claims exceed \$5,000,000, upon the request of any party, the Claims shall be decided by three arbitrators. All arbitration hearings shall commence within 90 days of the demand for arbitration and close within 90 days of commencement and the award of the arbitrator(s) shall be issued within 30 days of the close of the hearing; provided, however, that the arbitrator(s), upon a showing of good cause, may extend the commencement of the hearing for up to an additional 60 days. The arbitrator(s) shall provide a concise written statement of reasons for the award. The arbitration award may be submitted to any court having jurisdiction to be confirmed and enforced. The arbitrator(s) will have the authority to decide whether any Claim is barred by the statute of limitations and, if so, to dismiss the arbitration on that basis. For purposes of the application of the statute of limitations, the service on JAMS under applicable JAMS rules of a notice of Claim is the equivalent of the filing of a lawsuit. Any dispute concerning this arbitration provision or whether a Claim is arbitrable shall be determined by the arbitrator(s). The arbitrator(s) shall have the power to award legal fees pursuant to the terms of this Guaranty.
- (c) This Section 24 does not limit the right of any Guarantor or Bank to: (i) exercise self-help remedies, such as but not limited to, setoff; (ii) initiate judicial or nonjudicial foreclosure against any real or personal property collateral; (iii) exercise any judicial or power of sale rights; or (iv) act in a court of law to obtain an interim remedy, such as but not limited to, injunctive relief, writ of possession or appointment of a receiver, or additional or supplementary remedies. The filing of a court action is not intended to constitute a waiver of the right of any Guarantor or Bank, including the suing party, thereafter to require submittal of the Claim to arbitration.
- 25. Condition of the Borrower. Each Guarantor acknowledges and agrees that it has the sole responsibility for, and has adequate means of, obtaining from Borrower such information concerning the financial condition, business and operations of Borrower as such Guarantor requires, and that Bank has no duty, and such Guarantor is not relying on Bank at any time, to disclose to Guarantor any information relating to the business, operations or financial condition of Borrower.
- **26. Setoff.** If and to the extent any payment is not made when due hereunder, Bank may setoff and charge from time to time any amount so due against any or all of any Guarantor's accounts or deposits with Bank.
- **27. Other Guarantees.** Unless otherwise agreed by Bank and the applicable Guarantor in writing, this Guaranty is not intended to supersede or otherwise affect any other guaranty now or hereafter given by such Guarantor for the benefit of Bank or any term or provision thereof.
- **28.** Representations and Warranties. Each Guarantor represents and warrants that (a) such Guarantor is duly organized and in good standing under the laws of the jurisdiction of its organization and has full capacity and right to make and perform this Guaranty, and all necessary authority has been obtained; (b) this Guaranty constitutes such Guarantor's legal, valid and binding obligation enforceable in accordance with its terms; (c) the making and performance of this Guaranty

6

does not and will not violate the provisions of any applicable Laws, and does not and will not result in the breach of, or constitute a default or require any consent under, any material agreement, instrument, or document to which such Guarantor a party or by which it or any of its property may be bound or affected; (d) all consents, approvals, licenses and authorizations of, and filings and registrations with, any Governmental Authority required under applicable Laws for the making and performance of this Guaranty have been obtained or made and are in full force and effect; (e) by virtue of such Guarantor's relationship with Borrower, the execution, delivery and performance of this Guaranty is for the direct benefit of such Guarantor and it has received adequate consideration for this Guaranty; and (f) the financial information, that has been delivered to Bank by or on behalf of such Guarantor, is complete and correct in all respects and accurately presents the financial condition and the operational results of such Guarantor and since the date of the most recent financial statements delivered to Bank, there has been no material adverse change in the financial condition or operational results of such Guarantor.

Oral Agreements. ORAL AGREEMENTS OR ORAL COMMITMENTS TO LOAN MONEY, EXTEND CREDIT OR TO FORBEAR FROM ENFORCING REPAYMENT OF A DEBT ARE NOT ENFORCEABLE UNDER **WASHINGTON LAW.**

IN WITNESS WHEREOF, each Guarantor has executed this Guaranty by its duly authorized officer as of the day and year first above written.

USANA ACQUISITION CORP.

By: /s/ Gilbert A. Fuller Gilbert A. Fuller Name:

Title: Treasurer

WASATCH PRODUCT DEVELOPMENT, INC.

By: /s/ Gilbert A. Fuller

Name: Gilbert A. Fuller

Title: <u>Treasurer</u>

USANA HEALTH SCIENCES NEW ZEALAND, INC.

By: /s/ Gilbert A. Fuller

Name:

Gilbert A. Fuller Title: Treasurer

7

USANA CANADA HOLDING, INC.

By: /s/ Gilbert A. Fuller

Name: Gilbert A. Fuller Title: Secretary

FMG PRODUCTIONS, INC.

By: /s/ Gilbert A. Fuller

Title: V.P. and Treasurer

Gilbert A. Fuller Name:

INTERNATIONAL HOLDINGS, INC.

By: /s/ Mitchell Walkington

Name: Mitchell Walkington

Title: Secretary and Treasurer

8

ANNEX 1

SUPPLEMENT

, to the Continuing Guaranty dated as of June 16, 2004 (the "Guaranty") SUPPLEMENT NO. dated as of made by USANA ACQUISITION CORP., a Utah corporation ("USANA Acquisition"), WASATCH PRODUCT DEVELOPMENT, INC., a Utah corporation ("Wasatch"), USANA HEALTH SCIENCES NEW ZEALAND, INC., a Delaware corporation ("USANA New Zealand"), USANA CANADA HOLDING, INC., a Delaware corporation ("USANA Canada"), FMG PRODUCTIONS, INC., a Utah corporation ("FMG"), INTERNATIONAL HOLDINGS, INC., a Delaware corporation ("Holdings" and together with USANA Acquisition, Wasatch, USANA New Zealand, USANA Canada, FMG and any additional guarantors becoming a party thereto as provided in Section 18 thereof and their respective successors, collectively, the "Guarantors" and individually, a "Guarantor"), in favor of BANK OF AMERICA, N.A., a national banking association (together with its successors and assigns, "Bank"), and any Affiliate of Bank a party

RECITALS

- A. USANA Health Sciences, Inc., a Utah corporation ("<u>Borrower</u>") is a party to that certain Credit Agreement dated as of June 16, 2004 by and between Borrower and Bank (as amended, restated, modified, renewed, supplemented or extended from time to time, the "<u>Credit Agreement</u>").
- B. Capitalized terms used herein and not otherwise defined herein shall have the meanings assigned to such terms in the Guaranty.
- C. Guarantors have entered into the Guaranty in order to induce Bank to make Credit Extensions under the Credit Agreement, and pursuant to Section 6.13(a) of the Credit Agreement, each Material Subsidiary that is a Domestic Subsidiary that was not in existence or was not a Guarantor on the date of the Credit Agreement is required to enter into the Guaranty as a Guarantor upon becoming a Material Subsidiary.
- D. The undersigned Subsidiary ("New Guarantor") is executing this Supplement in accordance with the requirements of the Credit Agreement to become a Guarantor under the Guaranty in consideration for Bank to make Credit Extensions under the Credit Agreement.

Accordingly, New Guarantor agrees as follows:

- 1. In accordance with <u>Section 18</u> of the Guaranty, New Guarantor by its signature below becomes a Guarantor under the Guaranty with the same force and effect as if originally named therein as a Guarantor and New Guarantor hereby (a) agrees to all the terms and provisions of the Guaranty applicable to it as a Guarantor thereunder and (b) represents and warrants that the representations and warranties made by it as a Guarantor thereunder are true and correct on and as of the date hereof. Each reference to a "Guarantor" in the Guaranty shall be deemed to include New Guarantor. The Guaranty is hereby incorporated herein by reference.
- 2. New Guarantor represents and warrants to Bank that this Supplement has been duly authorized, executed and delivered by it and constitutes its legal, valid and binding obligation, enforceable against it in accordance with its terms.
- 3. This Supplement may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. This Supplement shall become effective when Bank shall have received counterparts of this Supplement that, when taken together, bear the signatures of New Guarantor and Bank. Delivery of an executed signature page to this Supplement by facsimile transmission shall, subject to applicable Law, be as effective as delivery of a manually-signed original thereof.
 - 4. Except as expressly supplemented hereby, the Guaranty shall remain in full force and effect.
- 5. THIS SUPPLEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF WASHINGTON APPLICABLE TO AGREEMENTS MADE AND TO BE PERFORMED ENTIRELY WITHIN SUCH STATE.
- 6. The unenforceability or invalidity of any provision of this Supplement shall not affect the enforceability or validity of any other provision herein.
- 7. All communications and notices hereunder shall be in writing and given as provided in <u>Section 19</u> of the Guaranty. All communications and notices hereunder to New Guarantor shall be given to it at the address set forth under its signature below.
- 8. New Guarantor agrees to reimburse Bank for its out-of-pocket expenses (including Attorney Costs) incurred in connection with this Supplement.
- 9. ORAL AGREEMENTS OR ORAL COMMITMENTS TO LOAN MONEY, EXTEND CREDIT OR TO FORBEAR FROM ENFORCING REPAYMENT OF A DEBT ARE NOT ENFORCEABLE UNDER WASHINGTON LAW.

IN WITNESS WHEREOF, New Guarantor has executed this Supplement by its duly authorized officer as of the day and year first above written.

[NEW GUARANTOR]

Rv.		
By: Name:		
Title:		
·		

PLEDGE AGREEMENT

This PLEDGE AGREEMENT ("Agreement") is entered into as of June 16, 2004, by USANA HEALTH SCIENCES, INC., a Utah corporation (together with its successors, "Pledgor") in favor of BANK OF AMERICA, N.A., a national banking association (together with its successors and assigns, "Pledgoe").

RECITALS

- A. Pledgor is a party to that certain Credit Agreement dated as of June 16, 2004 by and between Pledgor and Pledgee (as amended, restated, modified, renewed, supplemented or extended from time to time, the "<u>Credit Agreement</u>").
- B. It is a condition precedent to Pledgee's obligation to make its initial Credit Extension under the Credit Agreement that Pledgor enter into this Agreement and grant to Pledgee the security interests hereinafter provided to secure the obligations of Pledgor described below.

NOW, THEREFORE, in consideration of the foregoing, the mutual covenants and agreements herein contained, and for other good and valuable consideration receipt of which is hereby acknowledged, Pledgor hereby agrees as follows:

1. Definitions; Interpretation.

- (a) <u>Terms Defined in Credit Agreement</u>. All capitalized terms used in this Agreement and not otherwise defined herein have the meanings specified in the Credit Agreement.
 - (b) <u>Certain Defined Terms.</u> As used in this Agreement, the following terms have the following meanings:

"Capital Stock" means (i) in the case of a corporation, capital stock, (ii) in the case of an association or business entity, any and all shares, interests, participations, rights or other equivalents (however designated) of capital stock, (iii) in the case of a partnership, partnership interests (whether general or limited), (iv) in the case of a limited liability company, membership interests and (v) any other interest or participation that confers on a Person the right to receive a share of the profits and losses of, or distributions of assets of, the issuing Person.

"Credit Agreement" has the meaning set forth in the preamble hereto.

"Default Rate" has the meaning specified in Section 9.

"<u>Domestic Subsidiary</u>" means any Subsidiary that is organized and existing under the laws of the United States or any state or commonwealth thereof or under the laws of the District of Columbia.

"Foreign Subsidiary" means any Subsidiary that is not a Domestic Subsidiary.

"Pledged Collateral" has the meaning specified in Section 2.

"Pledged Capital Stock" means all of the Capital Stock referred to in Sections 2(a) and (b).

"Pledged Domestic Subsidiary" has the meaning specified in Section 2(a).

"Pledged Foreign Subsidiary" has the meaning specified in Section 2(a).

"Pledged Subsidiary" means any Pledged Domestic Subsidiary or any Pledged Foreign Subsidiary.

"Secured Obligations" means, collectively (i) all advances to, and debts, liabilities, obligations, covenants and duties of Pledgor arising under the Credit Agreement and each other Loan Document, including with respect to any Loan or Letter of Credit; (ii) all debts, liabilities, obligations, covenants and duties of Pledgor owing to Pledgee or any Affiliate of Pledgee and arising under any Swap Contract, including liabilities and obligations arising in connection with or as a result of early termination of any such Swap Contract; (iii) all debts, liabilities, obligations, covenants and duties of Pledgor or any of its Subsidiaries owing to Pledgee or any Affiliate of Pledgee and arising under any Financial Transaction Contract; and (iv) all debts, liabilities, obligations, covenants and duties of Pledgor arising under this Agreement, in each case whether direct or indirect (including those acquired by assumption), absolute or contingent, due or to become due, now existing or hereafter arising and including interest and fees that accrue after the commencement by or against any Loan Party or any Affiliate thereof of any proceeding under Debtor Relief Laws naming such Person as the debtor in such proceeding, regardless of whether such interest and fees are allowed claims in such proceeding.

"SEC" means the Securities and Exchange Commission, or any Governmental Authority succeeding to any of its principal functions.

"Securities Act" has the meaning specified in Section 11(d).

"<u>UCC</u>" means the Uniform Commercial Code as the same may, from time to time, be in effect in the State of Washington;

<u>provided</u>, <u>however</u>, in the event that, by reason of mandatory provisions of law, any or all of the attachment, perfection or priority of the security interest in any Pledged Collateral is governed by the Uniform Commercial Code as in effect in a jurisdiction other than the State of Washington, the term "UCC" shall mean the Uniform Commercial Code as in effect in such other jurisdiction for purposes of the provisions hereof relating to such attachment, perfection or priority and for purposes of definitions related to such provisions.

- (c) <u>Terms Defined in UCC</u>. Terms used in this Agreement that are defined in the UCC have the meanings given to them in the UCC.
- (d) <u>Interpretation</u>. The rules of construction and interpretation specified in <u>Sections 1.02</u> and <u>1.05</u> of the Credit Agreement also apply to this Agreement and are incorporated herein by this reference.
- 2. Pledge. As security for the payment or performance, as the case may be, in full of the Secured Obligations, Pledgor hereby pledges, assigns, transfers, hypothecates and sets over to Pledgee, its successors and assigns, and grants to Pledgee, its successors and assigns, a security interest in all of Pledgor's right, title and interest in, to and under the following, whether now existing or owned or hereafter acquired or arising (collectively, the "Pledged Collateral"):
- (a) <u>Pledged Capital Stock</u>. (i) All of the Capital Stock of each Domestic Subsidiary forth in Part (a) of <u>Schedule 1</u> hereto (each, a "<u>Pledged Domestic Subsidiary</u>") owned by Pledgor and (ii) all of the Capital Stock of each Foreign Subsidiary forth in Part (b) of <u>Schedule 1</u> hereto (each, a "<u>Pledged Foreign Subsidiary</u>") owned by Pledgor; provided that such Capital Stock shall not include more than 65% of all of the voting Capital Stock (within the meaning of Treas. Reg. Section 1.956 2(c)(2)) of any Foreign Subsidiary;
- (b) Additional Capital Stock. (i) All of the Capital Stock of any Pledged Domestic Subsidiary and, subject to the percentage restrictions described in subsection (a) above, all of the Capital Stock of any Pledged Foreign Subsidiary hereafter acquired, received or owned by Pledgor (whether in connection with any recapitalization, reclassification or reorganization of the capital of a Pledged Subsidiary or otherwise); and (ii) subject to the percentage restrictions described in subsection (a) above, all of the Capital Stock hereafter acquired, received or owned by Pledgor of any Person who, after the date hereof, becomes, as a result of any occurrence, a Pledged Subsidiary;
- (c) <u>Additional Interests, Etc.</u> (i) All certificates, instruments or other writings representing or evidencing the Pledged Capital Stock; (ii) all warrants, options and other rights entitling Pledgor to acquire

2

any interest in any Pledged Capital Stock, subject to the percentage restrictions described in subsection (a) above; and (iii) all dividends, cash, instruments and other property from time to time received, receivable or otherwise distributed or distributable in respect of or in exchange for any or all of the Pledged Capital Stock; and

- (d) <u>Proceeds.</u> All cash and non cash Proceeds of the foregoing, however and whenever acquired and in whatever form.
- 3. **Delivery of the Pledged Collateral; Continuing Security Interest.** Pledgor hereby agrees that:
- (a) <u>Delivery of Certificates</u>. Pledgor shall deliver to Pledgee (i) simultaneously with or prior to the execution and delivery of this Agreement, all certificates, instruments or other writings representing or evidencing Pledged Capital Stock and (ii) promptly upon the receipt thereof by or on behalf of Pledgor, all other certificates, instruments or other writings representing or evidencing Pledged Capital Stock together with a duly executed instrument in the form of <u>Annex 1</u> hereto (a "<u>Supplement</u>") identifying such additional Pledged Capital Stock; <u>provided</u> that the failure to deliver any such Supplement shall not affect the validity of such pledge of such Pledged Capital Stock. Prior to delivery to Pledgee, all such certificates and instruments constituting Pledged Collateral shall be held in trust by Pledgor for the benefit of Pledgee pursuant hereto. All such certificates shall be delivered in suitable form for transfer by delivery or shall be accompanied by duly executed instruments of transfer or assignment satisfactory to Pledgee. Pledgor hereby authorizes Pledgee to attach each Supplement to this Agreement and agrees that all Capital Stock listed thereon shall for all purposes hereunder constitute Pledged Collateral.
- (b) Additional Pledged Collateral. Subject to the percentage restrictions described in Section 2(a) above, if Pledgor shall receive by virtue of its being or having been the owner of any Pledged Collateral, any (i) certificate, including any certificate representing a dividend or distribution in connection with any increase or reduction of capital, reclassification, merger, consolidation, sale of assets, combination of shares or membership or equity interests, stock splits, spin-off or split-off, promissory notes or other instrument; (ii) warrant, option or other right, whether as an addition to, substitution for, or an exchange for, any Pledged Collateral or otherwise; (iii) dividends payable in securities; or (iv) distributions of securities or other equity interests in connection with a partial or total liquidation, dissolution or reduction of capital, capital surplus or paid in surplus, then Pledgor shall forthwith deliver all of the foregoing to Pledgee to hold as Pledged Collateral and shall, if received by Pledgor, be received in trust for the benefit of Pledgee, be segregated from the other property or funds of Pledgor, and be forthwith delivered to Pledgee as Pledged Collateral in the same form as so received, together with duly executed instruments of transfer or assignment satisfactory to Pledgee, as further collateral security for the Secured Obligations.
- (c) <u>Financing Statements</u>. Pledgor hereby irrevocably authorizes Pledgee at any time and from time to time to file in any relevant jurisdiction any initial financing statements and amendments thereto that contain the information required by Article 9 of the UCC of each applicable jurisdiction for the filing of any financing statement or amendment in order to perfect and protect the security interest of Pledgee in the Pledged Collateral.
- (d) <u>Continuing Security Interest.</u> Pledgor acknowledges and agrees that the security interest of Pledgee in the Pledged Collateral constitutes continuing collateral security for all of the Secured Obligations.

- **4. Representations and Warranties.** In addition to the representations and warranties of Pledgor set forth in the Credit Agreement, which are incorporated herein by this reference, Pledgor represents and warrants to Pledgee that:
- (a) <u>Ownership and Authority</u>. Pledgor is the sole legal and beneficial owner of the Pledged Collateral and has the right, power and authority to pledge, assign, transfer, hypothecate and set over to

3

Pledgee and grant to Pledgee a security interest in such Pledged Collateral pursuant hereto and to execute, deliver and perform its obligations in accordance with the terms of this Agreement.

- (b) <u>Authorization of Pledged Capital Stock</u>. All of the Pledged Capital Stock is duly authorized and validly issued, is fully paid and nonassessable and is not subject to the preemptive first-refusal or other similar rights of any Person. All Capital Stock hereafter constituting Pledged Collateral will be duly authorized and validly issued, fully paid and nonassessable and not subject to the preemptive first-refusal or other similar rights of any Person.
- (c) <u>Validity of Security Interest</u>. This Agreement creates a valid security interest in favor of Pledgee, its successors and assigns, in all of the Pledged Collateral. Upon the taking possession by Pledgee of the certificates (if any) representing the Pledged Capital Stock and all other certificates and instruments constituting Pledged Collateral, Pledgee will have a first priority perfected security interest in all certificated Pledged Capital Stock and such certificates and instruments. Upon the filing of UCC financing statements in the appropriate filing office in the jurisdiction of the incorporation or organization of Pledger, Pledgee shall have a first priority perfected security interest in all uncertificated Pledged Capital Stock consisting of partnership or limited liability company interests that do not constitute a "security" pursuant to Section 8-103(c) of the UCC. Except as set forth in this subsection (c), no action is necessary to perfect or otherwise protect such security interest.
- (d) <u>Absence of Liens and Claims</u>. Except for the security interest of Pledgee created hereby, the Pledged Collateral is free and clear of any Liens. There exists no "adverse claim" within the meaning of Section 8-102 of the UCC with respect to any of the Pledged Capital Stock.
- (e) <u>No Transfer Restrictions</u>. Except for restrictions imposed by the Loan Documents, the Pledged Collateral is free of contractual restrictions that might prohibit, impair, delay or otherwise affect the pledge of any Pledged Collateral hereunder or the sale or disposition thereof pursuant hereto.
- (f) <u>No Other Capital Stock</u>. Except as set forth in <u>Schedule 1</u> hereto, Pledgor owns no Capital Stock of any Pledged Subsidiary. The Pledged Capital Stock represents that percentage as set forth on <u>Schedule 1</u> hereto of all of the Capital Stock of each Pledged Subsidiary and sets forth all Capital Stock required to be pledged under this Agreement.
- (g) <u>Delivery of Capital Stock.</u> Pledgor has delivered or otherwise caused the transfer to Pledgee, pursuant to <u>Sections 3(a)</u> or <u>(b)</u>, as applicable, of all certificates, instruments or other writings representing, evidencing or constituting Pledged Collateral. The Pledged Collateral is not and shall not be represented or evidenced by any certificates, instruments or other writings other than those delivered hereunder.
- (h) <u>Margin Regulations</u>. The pledge of the Pledged Capital Stock pursuant to this Agreement does not violate the provisions of Regulation T, U or X of the Board of Governors of the Federal Reserve System as now and from time to time hereafter in effect. None of the Pledged Capital Stock constitutes "margin stock" within the meaning of such term under Regulation U.
- (i) <u>Partnership and Limited Liability Company Interests</u>. Except as previously disclosed in writing to Pledgee, none of the Pledged Capital Stock consisting of partnership or limited liability company interests (i) is dealt in or traded on a securities exchange or in a securities market, (ii) by its terms expressly provides that it is a security governed by Article 8 of the UCC, (iii) is an investment company security, (iv) is held in a securities account or (v) constitutes a security or a financial asset.
- (j) <u>Consents.</u> No consent or authorization of, filing with, or other act by or in respect of, any arbitrator or Governmental Authority and no consent of any other Person (including any stockholder, member or creditor of Pledgor), is required (i) for the pledge made by Pledgor or for the granting of the security interest by Pledgor pursuant to this Agreement or for the execution, delivery or performance of this Agreement by Pledgor or (ii) or the exercise by Pledgee of the rights and remedies provided for in this Agreement, except as may be required by Laws affecting the offering and sale of securities.

/

The foregoing representations and warranties shall survive the execution and delivery of this Agreement and shall be deemed restated automatically at each such time as any additional Pledged Collateral is delivered hereunder to Pledgee.

- **5. Covenants.** In addition to the covenants of Pledgor set forth in the Credit Agreement, which are incorporated herein by this reference, so long as the Commitment or any Swap Contract between any Loan Party and Pledgee or any Affiliate of Pledgee shall be in effect or any Secured Obligation shall remain unpaid or unsatisfied, Pledgor shall:
 - (a) <u>Defense of Pledged Collateral</u>. At its own cost and expense, take any and all actions necessary to defend title to the

Pledged Collateral against all Persons and to defend the security interest of Pledgee in the Pledged Collateral and the priority thereof against any Lien.

- (b) <u>Disposition of Pledged Collateral</u>. Not make or permit to be made any sale, transfer or other disposition of any of the Pledged Collateral or grant any option, warrant or other right or interest with respect to, any of the Pledged Collateral.
- (c) <u>Change of Name, Identity or Structure</u>. Promptly notify Pledgee in writing of any change: (i) in its corporate or organization name; (ii) in the location of its chief executive office, its principal place of business, any office in which it maintains books or records relating to the Pledged Collateral; (iii) in its identity, type of organization, corporate structure or jurisdiction of incorporation or organization; or (iv) in its Federal Taxpayer Identification Number or other identification number given by its jurisdiction of incorporation or organization, and not to effect or permit any change referred to in clauses (i) through (iv) unless all filings have been made under the UCC or otherwise that are required in order for Pledgee to continue at all times following such change to have a valid, legal and perfected first priority security interest in all the Pledged Collateral.
- (d) <u>No Liens</u>. Not make or permit to be made an assignment, pledge or hypothecation of any of the Pledged Collateral or create or permit to exist any Lien upon or with respect to any of the Pledged Collateral other than the security interest of Pledgee created hereby.
- (e) <u>Shareholder Agreements</u>. Not enter into any shareholder agreement, voting agreement, voting trust, irrevocable proxies or any other similar agreement or instrument with respect to any Pledged Collateral.
- (f) <u>Additional Capital Stock.</u> Immediately upon its acquisition, receipt or ownership, directly or indirectly, of any Pledged Capital Stock hereafter, pledge such Capital Stock pursuant to <u>Section 3(b)</u> and deliver or otherwise cause the transfer of all certificates, instruments or other writings representing, or evidencing such Pledged Capital Stock to Pledgee pursuant to <u>Sections 3(a)</u> or <u>(b)</u>, as applicable.
- (g) <u>Compliance with Securities Laws</u>. File all reports and other information now or hereafter required to be filed by Pledgor with the SEC and any other state, federal or foreign agency in connection with Pledgor's ownership of the Pledged Collateral.
- (h) <u>Partnership and Limited Liability Company Interests</u>. Not take any action or permit any action to be taken that would cause any membership interest in a limited liability company or partnership interest constituting Pledged Collateral to become a "security" as defined in Article 8 of the UCC, unless such membership interest or partnership interest has been certificated and pledged to Pledgee pursuant to this Agreement.
- 6. Further Assurances. Pledgor agrees, at its own expense, to execute, acknowledge, deliver and cause to be duly filed all such further instruments and documents and take all such actions as Pledgee may from time to time request to better assure, preserve, protect and perfect the Lien of Pledgee in the Pledged Collateral and the rights and remedies of Pledgee hereunder, including the payment of any fees and taxes required in connection with the execution and delivery of this Agreement, the granting of the security interest hereunder and the filing of any financing statements or other documents in

5

connection herewith or therewith. Without limiting the generality of the foregoing, Pledgor further agrees that it shall, concurrently with the execution of this Agreement and at any time and from time to time thereafter (a) procure, execute and deliver to Pledgee all stock powers, endorsements, financing statements, assignments and other instruments of transfer requested by Pledgee and (b) deliver to Pledgee immediately upon receipt the originals of all Pledged Capital Stock and all certificates, instruments or other writings representing, evidencing or constituting Pledged Collateral. Pledgor hereby authorizes Pledgee, with prompt notice thereof to Pledgor, to supplement this Agreement by supplementing Schedule1 hereto or adding additional schedules hereto to specifically identify any certificate, instrument or other writing that may represent or evidence Pledged Capital Stock; provided, however, that Pledgor shall have the right, exercisable within ten days after it has been notified by Pledgee of the specific identification of such Pledged Capital Stock, to advise Pledgee in writing of any inaccuracy of the representations and warranties made by Pledgor hereunder with respect to such Pledged Capital Stock. Pledgor agrees that it will use its reasonable best efforts to take such action as shall be necessary in order that all representations and warranties hereunder shall be true and correct with respect to such Pledged Capital Stock within 30 days after the date it has been notified by Pledgee of the specific identification of such Pledged Capital Stock.

7. Registration in Nominee Name; Denominations. Pledgee shall have the right (in its sole and absolute discretion) to hold the Pledged Capital Stock in its own name as pledgee, the name of its nominee (as pledgee or as sub agent) or the name of Pledgor, endorsed or assigned in blank or in favor of Pledgee. Pledgor will promptly give to Pledgee copies of any material notices or other communications received by it with respect to Pledged Capital Stock registered in the name of Pledgor. Pledgee shall at all times have the right to exchange the certificates representing Pledged Capital Stock for certificates of smaller or larger denominations for any purpose consistent with this Agreement.

8. Voting Rights; Dividends.

- (a) <u>Prior to the Occurrence of a Default</u>. So long as no Default or Event of Default shall exist or result therefrom (and, in the case of subparagraph (i) below, so long as written notice has not been given by Pledgee to Pledgor):
 - (i) <u>Voting Rights.</u> Pledgor shall be entitled to exercise any and all voting and other consensual rights pertaining to the Pledged Capital Stock or any part thereof for any purpose not inconsistent with the terms of this Agreement or the Credit Agreement; <u>provided</u>, <u>however</u>, that Pledgor shall not exercise or shall refrain from exercising any such right if, in the judgment of

Pledgee, such action would have a material adverse effect on the value of the Pledged Collateral or any part thereof or the interest of Pledgee therein, and, <u>provided</u>, <u>further</u>, that Pledgor shall give Pledgee at least five Business Days' prior written notice of the manner in which it intends to exercise, or the reasons for refraining from exercising, any such right.

(ii) <u>Dividends</u>. Pledgor shall be entitled to receive and retain any and all dividends or distributions paid in respect of the Pledged Capital Stock, in compliance with the terms of the Credit Agreement, except the following: (A) dividends paid or payable other than in cash in respect of, and instruments and other property received, receivable or otherwise distributed in respect of, or in exchange for, any Pledged Capital Stock; (B) dividends and other distributions paid or payable in cash in respect of any Pledged Capital Stock in connection with a partial or total liquidation or dissolution or in connection with a reduction of capital, capital surplus or paid-in-surplus; and (C) cash paid, payable or otherwise distributed in redemption of, or in exchange for, any Pledged Capital Stock; all of which shall be forthwith delivered to Pledgee to hold as, Pledged Collateral and shall, if received by Pledgor, be received in trust for the benefit of Pledgee, be segregated from the other property or funds of Pledgor, and be forthwith paid over or otherwise delivered to Pledgee as Pledged Collateral in the same form as so received, together with duly executed instruments of transfer or assignment satisfactory to Pledgee, as further collateral security for the Secured Obligations.

- (iii) <u>Proxies, Etc.</u> Pledgee shall execute and deliver (or cause to be executed and delivered) to Pledgor all such proxies and other instruments as Pledgor may request for the purpose of enabling Pledgor to exercise the voting and other rights which it is entitled to exercise pursuant to subparagraph (i) above and to receive the dividends or distributions which it is authorized to receive and retain pursuant to subparagraph (ii) above.
- (b) <u>Upon the Occurrence of a Default</u>. Upon the occurrence and during the continuance of a Default or an Event of Default:
 - (i) <u>Voting Rights.</u> All rights of Pledgor to exercise the voting and other consensual rights which it would otherwise be entitled to exercise pursuant to <u>Section 8(a)(i)</u> above shall cease upon written notice thereof from Pledgee, and all such rights shall thereupon become vested in Pledgee who shall thereupon have the sole right to exercise such voting and other consensual rights.
 - (ii) <u>Dividends</u>. All rights of Pledgor to receive the dividends or distributions which it would otherwise be authorized to receive and retain pursuant to <u>Section 8(a)(ii)</u> above shall cease, and all such rights shall thereupon become vested in Pledgee who shall thereupon have the sole right to receive and hold as Pledged Collateral such dividends. All dividends or distributions which are received by Pledgor contrary to the provisions of this subparagraph (ii) shall be received in trust for the benefit of Pledgee, shall be segregated from other funds of Pledgor and shall be forthwith paid over or otherwise delivered to Pledgee as Pledged Collateral in the same form as so received, together with duly executed instruments of transfer or assignment satisfactory to Pledgee, as further collateral security for the Secured Obligations.
 - (iii) <u>Proxies, Etc.</u> In order to permit Pledgee to exercise the voting and other rights which it may be entitled to exercise pursuant to subparagraph (i) above, and to receive all dividends and distributions which it may be entitled to receive under subparagraph (ii) above, Pledgor shall, if necessary, upon written notice of Pledgee, from time to time execute and deliver to Pledgee appropriate proxies, dividend payment orders and other instruments as Pledgee may request.
- Appointment of Pledgee. So long as the Commitment or any Swap Contract between Pledgor and Pledgee or any Affiliate of Pledgee shall be in effect or any Secured Obligation shall remain unpaid or unsatisfied, Pledgor does hereby designate and appoint Pledgee its true and lawful attorney coupled with an interest and with power irrevocable, upon the occurrence and during the continuation of any Event of Default, for the purpose of carrying out the provisions of this Agreement and taking any action and executing any instrument that Pledgee may deem necessary or advisable to accomplish the purposes hereof, including all of the following: (i) receive, endorse and collect all instruments made payable to Pledgor representing any dividend, interest payment or other distribution in respect of the Pledged Collateral or any part thereof and to give full discharge for the same; (ii) perfect or continue perfected, maintain the priority of or provide notice of Pledgee's security interest in the Pledged Collateral; (iii) exercise dominion and control over, and refuse to permit further withdrawals from any securities account constituting part of the Pledged Collateral; (iv) execute any and all endorsements, assignments or other documents and instruments necessary to sell, lease, assign, convey or otherwise transfer title in or dispose of the Pledged Collateral; and (v) execute any and all such other documents and instruments, and do any and all acts and things for and on behalf of Pledgor, which Pledgee may deem necessary or advisable to maintain, protect, realize upon and preserve the Pledged Collateral and Pledgee's security interest therein and to accomplish the purposes of this Agreement. The acceptance of this appointment by Pledgee shall not obligate it to perform any duty, covenant or obligation required to be performed by the Pledgor under or by virtue of the Pledged Collateral or to take any action in connection therewith. All expenses incurred by Pledgee in connection with exercising any of its rights under this Section shall bear interest at a per annum rate equal to the Base Rate (changing as the Base Rate changes) plus 4% per annum (the "Default Rate") from the date incurred until repaid by Pledgor; provided, however, that interest shall not accrue at a rate in excess of the maximum rate permitted by applicable Law. All amounts described in this Section shall be repayable by Pledgor on demand and Pledgor's obligation to make such

- 10. Events of Default. The occurrence of any of the following events shall constitute an "Event of Default:"
- (a) an Event of Default as defined in the Credit Agreement shall have occurred and be continuing;
- (b) any representation or warranty made or deemed made by Pledgor under or in connection with this Agreement shall prove to have been incorrect in any material respect when made or deemed made; or
- (c) Pledgor shall fail to perform or observe any other covenant, obligation or term of this Agreement and such failure shall continue unremedied for a period of 30 days after the earlier of (i) the date upon which written notice thereof shall have been given to Pledgor by Pledge or (ii) the date upon which a Responsible Officer of Pledgor knew or reasonably should have known of such failure.

11. Remedies.

- (a) <u>General Remedies</u>. If an Event of Default shall occur, Pledgee shall have, in addition to all other rights and remedies granted to it in this Agreement, the Credit Agreement, the other Loan Documents and the Environmental Agreement, all rights and remedies of a secured party under the UCC and other applicable Laws. Without limiting the generality of the foregoing, Pledgor agrees that Pledgee may:
 - (i) require Pledgor to assemble all or any part of the Pledged Collateral and make it available to Pledgee at any place and time designated by Pledgee;
 - (ii) exercise dominion and control over, and refuse to permit further withdrawals (whether of money, securities, instruments, investment property or other property) from any securities account constituting part of the Pledged Collateral; and
 - (iii) subject to the requirements of Laws affecting the offering and sale of securities, sell, resell, assign, transfer or otherwise dispose of any or all of the Pledged Collateral at public or private sale or at any broker's board or on any securities exchange, by one or more contracts, in one or more parcels, at the same or different times, for cash or credit, or for future delivery without assumption of any credit risk, all as Pledgee deems advisable.
- (b) Sale of Pledged Collateral. Each purchaser at any sale pursuant to this Agreement shall hold the property sold absolutely, free from any claim or right on the part of Pledgor, and Pledgor hereby waives, to the fullest extent permitted by applicable Laws, all rights of redemption, stay and appraisal which Pledgor now has or may at any time in the future have under any applicable Law now existing or hereafter enacted. Neither Pledgee's compliance with the UCC or any other applicable Law, in the conduct of any sale made pursuant to this Agreement, nor its disclaimer of any warranties relating to the Pledged Collateral, shall be considered to adversely affect the commercial reasonableness of such sale. Pledgee shall give Pledgor 10 days' written notice (which Pledgor agrees is reasonable notice within the meaning of Section 9-612 of the UCC) of Pledgee's intention to make any sale of Pledged Collateral. Pledgee shall not be obligated to make any sale of any Pledged Collateral if it shall determine not to do so, regardless of the fact that notice of sale of such Pledged Collateral shall have been given. Pledgee may, without notice or publication, adjourn any public or private sale or cause the same to be adjourned from time to time by announcement at the time and place fixed for sale, and such sale may, without further notice, be made at the time and place to which the same was so adjourned. To the fullest extent permitted by applicable Laws, Pledgee may bid for or purchase the Pledged Collateral or any part thereof offered for sale and may make payment on account thereof by using any claim then due and payable to

8

Pledgee from Pledgor as a credit against the purchase price and Pledgee may, upon compliance with the terms of sale, hold, retain and dispose of such property without further accountability to Pledgor therefor. Pledgor agrees to pay any deficiency remaining after collection or realization by Pledgee on the Pledged Collateral.

- (c) <u>Waiver of Rights to Purchase</u>. Pledgor, for itself and its successors and assigns, does hereby irrevocably waive and release all preemptive, first-refusal and other similar rights of Pledgor to purchase any or all of the Pledged Capital Stock upon any sale thereof by Pledgee under this Agreement, whether such right to purchase arises under any of the Pledged Subsidiary's Organization Documents, by agreement, by operation of law or otherwise.
- (d) Exempt Sales Transactions. Pledgor recognizes that, by reason of certain prohibitions contained in the Securities Act of 1933, as from time to time amended (the "Securities Act") and applicable state securities laws, Pledgee may be compelled, with respect to any sale of all or any part of the Pledged Capital Stock, to limit purchasers to those who will agree, among other things, to acquire such securities for their own account, for investment, and not with a view to the distribution or resale thereof. Pledgor acknowledges and agrees that any such sale may result in prices and other terms less favorable to the seller than if such sale were a public sale without such restrictions and notwithstanding such circumstances, agrees that any such sale shall be deemed to have been made in a commercially reasonable manner. Pledgee shall be under no obligation to delay the sale of any of the Pledged Capital Stock for the period of time necessary to permit Pledgor to register such securities for public sale under the Securities Act, or under applicable state securities laws, even if Pledgor would agree to do so. If Pledgee determines to exercise its right to sell any or all of the Pledged Capital Stock, upon written request, Pledgor shall and shall cause each of the Pledged Subsidiaries to, from time to time, furnish to Pledgee all such information as Pledgee may request in order to determine the number of shares and other instruments included in the Pledged Capital Stock which may be sold by Pledgee as exempt transactions under the Securities Act and rules of the SEC thereunder, as the same are from time to time in effect.
 - (e) Retention of Pledged Collateral. Pledgee may, after providing the notices required by Section 9-620(a) of the UCC or

otherwise complying with any requirement of applicable Law, accept or retain the Pledged Collateral or any part thereof in satisfaction of the Secured Obligations. Unless and until Pledgee shall have provided such notices, however, Pledgee shall not be deemed to have retained any Pledged Collateral in satisfaction of any Secured Obligations for any reason.

- (f) <u>Duty of Care</u>. Except for the exercise of reasonable care in the custody of any Pledged Collateral in its possession and the accounting for moneys actually received by it hereunder, Pledgee shall have no duty as to any Pledged Collateral or as to the taking of any necessary steps to preserve rights against prior parties or any other rights pertaining to any Pledged Collateral. Pledgee shall be deemed to have exercised reasonable care in the custody and preservation of Pledged Collateral in its possession if such Pledged Collateral is accorded treatment substantially equal to that which Pledgee accords its own property. Neither Pledgee nor any of its Affiliates, directors, officers, employees, counsel, agents and attorneys-in-fact shall be liable for failure to demand, collect or realize upon all or any part of the Pledged Collateral or for any delay in doing so or shall be under any obligation to sell or otherwise dispose of any Pledged Collateral upon the request of a Pledgor or otherwise.
- (g) <u>Application of Proceeds</u>. Except as otherwise required by the UCC or other applicable Laws, the cash proceeds actually received from the sale or other disposition or collection of the Pledged Collateral, and any other amounts received in respect of the Pledged Collateral the application of which is not otherwise provided for herein, shall be applied by Pledgee in accordance with the terms of the Credit Agreement. Pledgor shall remain liable to Pledgee and the other Secured Parties for any deficiency which exists after any sale or other disposition or collection of the Pledged Collateral.
- 12. Certain Waivers. Pledgor waives, to the fullest extent permitted by applicable Laws, (a) any right of redemption with respect to the Pledged Collateral, whether before or after sale hereunder, and all rights, if any, of marshalling of the Pledged Collateral or other collateral or security for the Secured

9

Obligations; (b) any right to require Pledgee (i) to proceed against any Person, (ii) to exhaust any other collateral or security for any of the Secured Obligations, (iii) to pursue any remedy in Pledgee's power, or (iv) to make or give any presentments, demands for performance, notices of nonperformance, protests, notices of protests or notices of dishonor in connection with any of the Pledged Collateral; and (c) all claims, damages, and demands against Pledgee arising out of the repossession, retention, sale or application of the proceeds of any sale of the Pledged Collateral.

- 13. No Waiver; Cumulative Remedies. No failure by Pledgee to exercise, and no delay in exercising, any right, power or remedy under this Agreement shall operate as a waiver thereof, nor shall any single or partial exercise of any right, power or remedy hereunder preclude any other or further exercise thereof or the exercise of any other right, power, or remedy. The exercise of any right, power, or remedy shall in no event constitute a cure or waiver of any Event of Default or prejudice the rights of Pledgee in the exercise of any right hereunder. The rights and remedies provided herein are cumulative and not exclusive of any right or remedy provided by applicable Law.
- 14. Expenses Incurred. Pledgee is not required to, but may, at its option, pay any tax, insurance premium, filing or recording fees, or other charges payable by Pledgor hereunder and any such amount shall bear interest at the Default Rate from the date of payment until repaid. Pledgor agrees to pay or reimburse Pledgee on demand for all fees, costs or out-of-pocket expenses (including Attorney Costs) incurred by Pledgee in connection with (i) the custody or preservation of, or the sale of, collection from or other realization upon any of the Pledged Collateral (including all expenses of sales and collections of the Pledged Collateral), (ii) the preservation, protection or enforcement of any of the rights of Pledgee under this Agreement (including any fees, costs and out-of-pocket expenses incurred in the preservation, protection or enforcement of any rights of Pledgee in any case commenced by or against Pledgor under the Bankruptcy Code (Title 11, United States Code) or any similar or successor statute) or (iii) the failure of Pledgor to perform or observe any its obligations under this Agreement, and any such amount shall bear interest at the Default Rate from the date such expenditures are made by Pledgee until repaid. All amounts described in this Section shall be repayable by Pledgor on demand and Pledgor's obligation to make such repayment shall constitute an additional Secured Obligation.
- 15. Integration. This Agreement, together with the other Loan Documents to which Pledgor is a party and the Environmental Indemnity Agreement, comprises the complete and integrated agreement of the parties on the subject matter hereof and thereof and supersedes all prior agreements, written or oral, on such subject matter. Pledgor acknowledges that this Agreement, the other Loan Documents and the Environmental Indemnity Agreement may contain covenants and other terms and provisions variously stated regarding the same or similar matters, and agrees that all such covenants, terms and provisions are cumulative and all shall be performed and satisfied in accordance with their respective terms.
- **16. Amendments.** No provision of this Agreement may be waived, amended, supplemented or modified, except by a written instrument executed by Pledgee and Pledgor.
- 17. Notices. All communications and notices hereunder shall (except as otherwise expressly permitted herein) be in writing and given as provided in Section 9.02 of the Credit Agreement.
- 18. Governing Law. THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF WASHINGTON APPLICABLE TO AGREEMENTS MADE AND TO BE PERFORMED ENTIRELY WITHIN SUCH STATE (EXCEPT TO THE EXTENT THE VALIDITY OR PERFECTION OF THE SECURITY INTERESTS HEREUNDER, OR THE REMEDIES HEREUNDER, IN RESPECT OF ANY PLEDGED COLLATERAL ARE GOVERNED BY THE LAW OF A JURISDICTION OTHER THAN THE STATE OF WASHINGTON); PROVIDED THAT SECURED PARTY SHALL RETAIN ALL RIGHTS ARISING UNDER FEDERAL LAW.

(b) inure to the benefit of Pledgee and any Affiliate of Pledgee and their respective successors and assigns a party to a Swap Contract and/or Financial Transaction Contract with Pledgor and Pledgee and any Affiliate of Pledgee a party to a Swap Contract and/or Financial Transaction Contract with Pledgor may, without notice to Pledgor and without affecting Pledgor's obligations hereunder, assign or sell participations in the Secured Obligations and this Agreement, in whole or in part. Pledgor agrees that Pledgee and any Affiliate of Pledgee a party to a Swap Contract and/or Financial Transaction Contract with Pledgor may disclose to any prospective purchaser and any purchaser of all or part of the Secured Obligations any and all information in Pledgee's possession concerning Pledgor, this Agreement and/or the Pledged Collateral.

20. No Waiver; Enforceability. No failure by Pledgee to exercise, and no delay in exercising, any right, remedy or power hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy or power hereunder preclude any other or further exercise thereof or the exercise of any other right. The remedies herein provided are cumulative and not exclusive of any remedies provided by applicable Law or in equity. The unenforceability or invalidity of any provision of this Agreement shall not affect the enforceability or validity of any other provision herein.

IN WITNESS WHEREOF, Pledgor has caused this Agreement to be duly executed as of the date first above written.

USANA HEALTH SCIENCES, INC.

By: /s/ Gilbert A. Fuller

Name: Gilbert A. Fuller

Title: Sr. Vice Pres. and CFO

11

SCHEDULE 1

PLEDGED CAPITAL STOCK

Part (a). <u>Pledged Domestic Subsidiaries</u>.

Name of Subsidiary	Number of Shares	Certificate Number	Percentage Ownership
USANA Canada Holding, Inc.	1,000	01	100%
USANA Health Sciences			
New Zealand, Inc.	1,000	1	100%
FMG Productions, Inc.	2,300	2	76.67%
FMG Productions, Inc.	700	3	23.33%
International Holdings, Inc.	1,000	000001	100%
USANA Acquisition Corp.	1,000	1	100%

Part (b). <u>Pledged Foreign Subsidiaries</u>.

Name of Subsidiary	Number of Shares	Certificate Number	Percentage Ownership
USANA Australia Pty Ltd	3,025,085	4	100%
USANA Hong Kong Limited	6,990,499	3	99.9%
USANA Health Sciences			
Singapore Pte. Ltd.	2	3	100%

USANA México, S.A. de C.V.	499	1	99.8%
USANA Japan, Inc.	170	1	100%

ANNEX 1

SUPPLEMENT

SUPPLEMENT NO. dated as of , to the Pledge Agreement dated as of June 16, 2004 (the "<u>Pledge Agreement</u>") made by USANA HEALTH SCIENCES, INC., a Utah corporation (together with its successors, "<u>Pledgor</u>") in favor of BANK OF AMERICA, N.A., a national banking association (together with its successors and assigns, "<u>Pledgee</u>").

RECITALS

- A. Pledgor is a party to that certain Credit Agreement dated as of June 16, 2004 by and between Pledgor and Pledgee (as amended, restated, modified, renewed, supplemented or extended from time to time, the "<u>Credit Agreement</u>").
- B. Capitalized terms used herein and not otherwise defined herein shall have the meanings assigned to such terms in the Pledge Agreement.
- C. Pledgor has entered into the Pledge Agreement in order to induce Pledgee to make Credit Extensions under the Credit Agreement, and pursuant to Section 3(a) of the Pledge Agreement Pledgor is required to deliver to Pledgee all certificates, instruments or other writings representing or evidencing Pledged Capital Stock received by Pledgor after the date of the Pledge Agreement together with a duly executed Supplement substantially the form hereof.
- D. Pledgor is executing this Supplement in accordance with the requirements of the Pledge Agreement in consideration for Pledgee to make Credit Extensions under the Credit Agreement.

Accordingly, Pledgor agrees as follows:

- 1. Pledgor agrees that this Supplement may be attached to the Pledge Agreement and that the shares of capital stock or other equity securities listed on this Supplement shall be and become part of the Pledged Collateral referred to in the Pledge Agreement and shall secure all Secured Obligations.
- 2. Pledgor agrees that the shares of capital stock and other equity securities listed below shall for all purposes constitute Pledged Collateral and shall be subject to the security interest created by the Pledge Agreement.

	Number of	Certificate	Percentage
Name of Subsidiary	Shares	Number	Ownership

- 3. Pledgor certifies that the representations and warranties set forth in <u>Section 4</u> of the Pledge Agreement are true and correct with respect to the shares of capital stock or other equity securities listed on this Supplement on and as of the date hereof.
 - 4. Except as expressly supplemented hereby, the Pledge Agreement shall remain in full force and effect.
- 5. THIS SUPPLEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAW OF THE STATE OF WASHINGTON APPLICABLE TO AGREEMENTS MADE AND TO BE PERFORMED ENTIRELY WITHIN SUCH STATE.
- 6. Pledgor agrees to reimburse Pledgee for its out-of-pocket expenses (including Attorney Costs) incurred in connection with this Supplement.

IN WITNESS WHEREOF, Pledgor has executed this Supplement by its duly authorized officer as of the day and year first above written.

USANA HEALTH SCIENCES, INC.

By: /	s/ Gilbert A. Fuller	
Name:		Gilbert A. Fuller
Title:	Sr. Vice Pres. and CFO	<u> </u>

Accepted:

BANK OF AMERICA, N.A.

By: /s/ Mark N. Crawford	
Name: Mark N. Crawford	
Title: Senior Vice President	
	2

EXHIBIT E

COMPLIANCE CERTIFICATE

Financial Statement Date:

To: Bank of America, N.A.

Ladies and Gentlemen:

Reference is made to that certain Credit Agreement, dated as of June 16, 2004 (as amended, restated, extended, supplemented or otherwise modified in writing from time to time, the "<u>Credit Agreement</u>;" the terms defined therein being used herein as therein defined), between USANA Health Sciences, Inc., a Utah corporation ("<u>Borrower</u>"), and Bank of America, N.A., a national banking association.

The undersigned Responsible Officer hereby certifies as of the date hereof that he/she is the of Borrower, and that, as such, he/she is authorized to execute and deliver this Certificate to Bank on behalf of Borrower, and that:

[Use following for fiscal year-end financial statements]

1. Attached hereto as <u>Schedule 1</u> are the year-end audited financial statements required by <u>Section 6.01(a)</u> of the Credit Agreement for the fiscal year of Borrower ended as of the above date, together with the report and opinion of an independent certified public accountant required by such section.

[Use following for fiscal quarter-end financial statements]

- 1. Attached hereto as <u>Schedule 1</u> are the unaudited financial statements required by <u>Section 6.01(b)</u> of the Credit Agreement for the fiscal quarter of Borrower ended as of the above date. Such financial statements fairly present the financial condition, results of operations and cash flows of Borrower and its Subsidiaries in accordance with GAAP as at such date and for such period, subject only to normal year-end audit adjustments and the absence of footnotes.
- 2. The undersigned has reviewed and is familiar with the terms of the Credit Agreement and has made, or has caused to be made under his/her supervision, a detailed review of the transactions and condition (financial or otherwise) of Borrower during the accounting period covered by the attached financial statements.
- 3. A review of the activities of Borrower during such fiscal period has been made under the supervision of the undersigned with a view to determining whether during such fiscal period Borrower performed and observed all its Obligations under the Loan Documents, and

[select one:]

[to the best knowledge of the undersigned during such fiscal period, Borrower performed and observed each covenant and condition of the Loan Documents applicable to it.]

-or-

[the following covenants or conditions have not been performed or observed and the following is a list of each such Default or Event of Default and its nature and status:]

4. The representations and warranties of Borrower contained in <u>Article V</u> of the Credit Agreement, or which are contained in any document furnished at any time under or in connection with the Loan Documents, are true and correct on and as of the date hereof, except to the extent that such

representations and warranties specifically refer to an earlier date, in which case they are true and correct as of such earlier date, and except that for purposes of this Compliance Certificate, the representations and warranties contained in subsections (a) and (b) of Section 5.05 of the Credit Agreement shall be deemed to refer to the most recent statements furnished pursuant to clauses (a) and (b), respectively, of Section 6.01 of the Credit Agreement, including the statements in connection with which this Compliance Certificate is delivered.

IN WITNESS WHEREOF, the undersigned has exc	ecuted this Certificate as of ,		
	USANA HEALTH SCIENCES,	USANA HEALTH SCIENCES, INC.	
	By: /s/ Gilbert A. Fuller Name: Title: Sr. Vice Pres. and CFO	Gilbert A. Fuller	
	2		
	For the Quarter/Year ended	("Statement Date")	
to th	SCHEDULE 2 the Compliance Certificate (\$ in 000's)		

CHIEF EXECUTIVE OFFICER CERTIFICATION

- I, Myron W. Wentz, Chief Executive Officer of USANA Health Sciences, Inc., certify that:
 - 1. I have reviewed this Quarterly Report on Form 10-Q of USANA Health Sciences, Inc. (the "Registrant");
 - Based on my knowledge, this Quarterly Report does not contain any untrue statement of a material fact or omit to state a
 material fact necessary to make the statements made, in light of the circumstances under which such statements were made,
 not misleading with respect to the period covered by this Quarterly Report;
 - 3. Based on my knowledge, the financial statements, and other financial information included in this Quarterly Report, fairly present in all material respects the financial condition, results of operations and cash flows of the Registrant as of, and for, the periods presented in this Quarterly Report;
 - 4. The Registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the Registrant and have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the Registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this Quarterly Report is being prepared;
 - evaluated the effectiveness of the Registrant's disclosure controls and procedures and presented in this Quarterly Report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this Quarterly Report based on such evaluation; and
 - c) disclosed in this Quarterly Report any change in the Registrant's internal control over financial reporting that occurred during the Registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the Registrant's internal control over financial reporting; and
 - 5. The Registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the Registrant's board of directors (or persons performing the equivalent functions):
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the Registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the the Registrant's internal control over financial reporting.

Date: August 5, 2004 /s/ Myron W. Wentz
Myron W. Wentz, PhD

Chief Executive Officer (Principal Executive Officer)

CHIEF FINANCIAL OFFICER CERTIFICATION

- I, Gilbert A. Fuller, Chief Financial Officer of USANA Health Sciences, Inc., certify that:
 - 1. I have reviewed this Quarterly Report on Form 10-Q of USANA Health Sciences, Inc. (the "Registrant");
 - Based on my knowledge, this Quarterly Report does not contain any untrue statement of a material fact or omit to state a
 material fact necessary to make the statements made, in light of the circumstances under which such statements were made,
 not misleading with respect to the period covered by this Quarterly Report;
 - 3. Based on my knowledge, the financial statements, and other financial information included in this Quarterly Report, fairly present in all material respects the financial condition, results of operations and cash flows of the Registrant as of, and for, the periods presented in this Quarterly Report;
 - 4. The Registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the Registrant and have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the Registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this Quarterly Report is being prepared;
 - evaluated the effectiveness of the Registrant's disclosure controls and procedures and presented in this Quarterly Report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this Quarterly Report based on such evaluation; and
 - c) disclosed in this Quarterly Report any change in the Registrant's internal control over financial reporting that occurred during the Registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the Registrant's internal control over financial reporting; and
 - 5. The Registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the Registrant's board of directors (or persons performing the equivalent functions):
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the Registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the the Registrant's internal control over financial reporting.

Date: August 5, 2004 /s/ Gilbert A. Fuller

Gilbert A. Fuller Chief Financial Officer (Principal Financial and Accounting Officer)

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

The undersigned hereby certifies that the Quarterly Report on Form 10-Q of USANA Health Sciences, Inc. for the quarter ended July 3, 2004 as filed August 5, 2004 with the Securities and Exchange Commission, fully complies with the requirements of Section 13(a) or 15(d) of The Securities Exchange Act of 1934 (15 U.S.C. 78m) and that the information contained in the Quarterly Report fairly presents, in all material respects, the financial condition and results of operations of USANA Health Sciences, Inc.

Date: August 5, 2004 /s/ Myron W. Wentz

Chairman and Chief Executive Officer (Principal Executive Officer)

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

The undersigned hereby certifies that the Quarterly Report on Form 10-Q of USANA Health Sciences, Inc. for the quarter ended July 3, 2004 as filed August 5, 2004 with the Securities and Exchange Commission, fully complies with the requirements of Section 13(a) or 15(d) of The Securities Exchange Act of 1934 (15 U.S.C. 78m) and that the information contained in the Quarterly Report fairly presents, in all material respects, the financial condition and results of operations of USANA Health Sciences, Inc.

Date: August 5, 2004 /s/ Gilbert A. Fuller

Gilbert A. Fuller

Senior Vice President and Chief Financial Officer (Principal Financial and Accounting Officer)